### AGENDA | APRIL 2, 2025

### April 2, 2025 | 3:15PM -5:00PM | ZOOM

- Call to Order
- 2. Approval of the March 6, 2025 Minutes (Action)
- 3. Report of the Chair (Information)
- 4. Special Order

  Jennifer King Rice

  Senior Vice President and Provost

  Strategic Plan Update
- 5. ERG Report: Proposal for Review of the University Senate Bylaws, Committee Presiding Officer Selection (Senate Document #24-25-11) (Information)
- 6. ERG Report: Proposal for Review of The Campus Transportation Advisory Committee (CTAC) Senate Representation (Senate Document #19-20-15) (Information)
- 7. PCC Report: Proposal to Revise the Policy and Guidelines for Combined Bachelor's/Master's Programs (Senate Document #24-25-20) (Action)
- 8. New Business
- 9. Adjournment

### MINUTES | MARCH 6, 2025

3:15PM - 5:00PM | ZOOM | MEMBERS PRESENT: 109

### CALL TO ORDER

Senate Chair Sly called the meeting to order at 3:27p.m.

### APPROVAL OF THE MINUTES, FEBRUARY 5, 2025, MEETING

Chair Sly asked if there were any corrections to the minutes of the February 5, 2025, meeting; hearing none, Chair Sly declared the minutes approved as distributed.

### REPORT OF THE CHAIR

### Thanking Chair-Elect Dammeyer

Chair Sly thanked Chair-Elect Dammeyer for leading the February 5th Senate Meeting.

### Senate Conduct of Members

Chair Sly reviewed concerns regarding Senator conduct from the February 5, 2025, Senate Meeting, as reminded members that the University Senate cannot allow inappropriate language of any kind. Chair Sly shared that while there was, and remains, some frustration with the voting system, upon review the issues Senators encountered were primarily user error.

Chair Sly addressed inappropriate behavior with the chat feature and reminded members that the chat is not to be used as a side bar during meetings. All senator communications during a meeting should be vocalized and read into the record and should not be placed in the chat. Chair Sly encouraged participation during University Senate meetings and asked that members please use the Raise Hand feature to comment on business items or use the chat to simply state "point of order" and wait to be called on to speak to the assembly.

### Concerns on the Antisemitism and Islamophobia Task Force Report

Chair Sly shared that Senate Leadership has been in communication about the Antisemitism and Islamophobia Task Force Report, which was presented at the February 5, 2025, University Senate Meeting. Chair Sly said, there is no further action on this item at this time, however, there is at least one note of dissent on The Report stemming from definitional elements within the report, disagreement regarding action items from the report and a question about the formulation of the Joint-Task Force with the Office of the President and the University Senate. There are also statements of support for the Task Force Report which refute the elements the dissenting voice has raised and argue for the recommendations.

Chair Sly explained that this item is closed, but that there will be space to discuss the recommendations of the Task Force Report as they come to the Senate at a later date.

Chair Sly said that Senators have asked how to circulate information amongst Senators to aid in voting, and explained that while this is not common practice, information can be provided and included alongside Resolutions and voting items.

Holly Mercer, Dean of Libraries Starting July 1, 2025

Chair Sly announced that Holly Mercer has been named the new Dean of University Libraries, effective July 1, 2025. Chair Sly shared the hope that Mercer will provide visionary leadership, guiding our libraries' growth through innovative, user-centered programs, services, and spaces.

Chair Sly thanked Daniel Mack on behalf of the University Senate for leadership on campus as Interim Dean of University Libraries.

### Announcing Michele Eastman's Retirement in Summer 2025

Chair Sly announced the Summer 2025 retirement of Michele Eastman, Assistant President and Chief of Staff. Chair Sly shared that Earman has been with the University of Maryland since 1981, and in the role of Assistant President and Chief of Staff since 2011.

Chair Sly highlighted Eatman's transformative impacts at the University, including:

- The creation of the School of Languages, Literatures and Cultures with a merger of four departments
- Enhanced UMD's reputation as a leading center for the arts with the opening of The Clarice
- Guiding the transition of key research centers
- Played a significant role in the formation of the Graduate School and the Division of Research

Chair Sly also shared that Eastman has been a central figure in the MPowering the State initiative and contributed to countless campus committees. Chair Sly said that Eatman has helped lead UMD through significant challenges and milestones, leaving a lasting legacy.

Chair Sly thanked Eastman on behalf of the University Senate for the guidance and wisdom provided to the University of Maryland.

### Senate Meeting Schedule

Chair Sly informed the Senate that there are three meetings left for this academic year – all on Zoom. There will be two meetings in April where much of the work that is in the Standing Committees will be coming forward to the Senate for a vote. Chair Sly encouraged Senators to make every attempt to attend those important meetings.

Chair Jarzynski also reminded all Senators that the May 7, 2025 Senate meeting is the transition meeting where all continuing Senators and newly elected Senators will elect the next Chair-Elect of the Senate and vote on the elected committees and councils of the Senate such as the Senate Executive Committee, the Committee on Committees, and others that the Nominations Committee is developing slates for candidates.

### Nominations for Elected Committees and Councils

Chair Sly informed the Senate that the Senate Nominations Committee is in the process of generating slates of candidates to run for open positions on Senate-elected committees and councils, including next year's Senate Executive Committee, Senate Committee on Committees, University Athletic Council, Council of University System Faculty, and Campus Transportation Advisory Committee.

Chair Sly also informed the Senate that the Nominations Committee is still looking for candidates to run for these important positions. Current Senators have already been contacted regarding these positions, and newly elected Senators will be contacted shortly to solicit self-nominations.

### Volunteering for Senate Standing Committees

Chair Sly informed the Senate that in addition to the Elected Committees and Councils, the Senate Committee on Committees will soon begin filling the University Senate's ten standing committees. These are volunteer positions, and most do not require one to be a Senator to serve.

The process will begin in April, and Senators are encouraged to share this as an opportunity to become involved in our campus shared governance. The link to learn more about joining a committee is available on the Senate website.

### Senate Elections

Chair Sly informed the Senate that the candidacy period for the staff, student, and single-member constituency elections for the 2025-2026 Senate ended on February 7, 2025. In addition, elections for these constituencies began on February 17, 2025, and will end on March 7, 2025.

### SPECIAL ORDER OF THE DAY

# Darryll J. Pines President of the University of Maryland State of the Campus Address

Chair Sly invited Dr. Darryll Pines, President of the University of Maryland, to present on the State of the Campus.

President Pines thanked Senators for service and support to shared governance at the University of Maryland, and acknowledged the importance of the partnership between University Administration and the University Senate.

President Pines recognized the service of Daniel Mack, Interim Dean of Libraries and the hiring of Holly Mercer as the new Dean of University Libraries.

President Pines recognized the service of Assistant President and Chief of Staff, Michele Eastman, who is retiring this summer. President Pines shared that Eastman has been at the University for over forty years, serving three Presidents.

President Pines recognized Gregory Ball, Vice President of Research who will step down this summer after four years of service.

President Pines introduced Anne Buckley, Vice President for Marketing and Communications and Charles "Bob" Reuning, Interim Vice President and Chief Administrative Officer, and acknowledged their service.

### Major Initiatives

President Pines shared that the University of Maryland (UMD) remains focused on its mission to provide excellent teaching, research, and service, while advancing knowledge in areas of importance to the State, Nation, and the World.

President Pines shared following Major Initiatives:

- A \$500 Billion investment towards the Artificial Intelligence (AI) Infrastructure initiative by President Trump as a response to the artificial intelligence movement will have a global impact. The Artificial Intelligence Interdisciplinary Institute at Maryland (AIM), led by Dr. Hal Daumé, was designed to be a collaborative space to advance responsible and ethical AI for the public good. AIM has 25 affiliate faculty members across all disciplines within the University and all 12 colleges. It is creating new degrees and general education courses. It will also provide seed funding for AI related projects that positively impact society. Pines shared that UMD was ahead of the curve with regards to AI. There will be new degree programs offered this fall, including an Artificial Intelligence degree in Philosophy followed by an Artificial Intelligence degree in Computer Science.
- An investment of up to \$1 Billion towards the Capital Quantum announced by Governor Moore will have a national impact. This is a direct result of work by UMD faculty, students, and postdocs in Physics and Engineering. As part of Governor Moore's Economic Development agenda, the Capital of Quantum Initiative is a strategic partnership that will combine state, federal, private sector, and philanthropic resources over the next five years. President Pines shared that Governor Moore made this announcement in College Park at IONQ, a quantum computing company located in the Discovery District.

For the University, the initiative will recruit top quantum scientists and engineers, expand access to the National Quantum Laboratory (QLab), and construct a new building for the Applied Research Laboratory for Intelligence and Security (ARLIS). It will also create additional facilities to house an expanded Quantum Startup Foundry, which helps ideas from faculty staff and students in the quantum area get translated to startups. The initiative will develop a high school quantum curriculum, which is the first of its kind.

The Capital of Quantum Initiative will also make a difference for IONQ, allowing it to grow its corporate headquarters in College Park, and develop a larger skilled workforce for this discipline and industry.

• The new Edward and Jennifer St. John Center for Translational Engineering and Medicine is being launched with a \$10 Million gift from the St. John's, both their foundation and their personal commitment as well as a five-year, \$12.7 Million award from MPower. The Center will drive medical innovation through UMD and the University of Maryland, Baltimore and will foster face-to-face partnerships between clinicians and engineers at UMSOM and Clark School.

The Center will be located on the fourth floor of 4MLK, a new state-of-the-art facility in University of Maryland's BioPark in Baltimore. It will ensure that real world patient concerns are translated into devices, diagnostics and treatment. The focus is to create "Physicianeers", visionary people who can bridge the medical and engineering worlds to build a healthier future for all.

### Indicators of Excellence

President Pines shared Indicators of Excellence at the University, including US News & World Report rankings. With regards to online programs, President Pines noted that the Master of Business Administration (MBA) Program was ranked #14, and has spent nine years in the Top 15, with the MBAs in Marketing (#6) and General Management (#7) earning Top 10 rankings. The Master of Engineering Program ranked at #9 and the Master's in Electrical Engineering at #6.

President Pines proudly shared that the University has 71 programs ranked in the Top 25, compared five years ago, where there were 60.

President Pines shared that the University is once again a top Fulbright producing institution with 17 recent graduates, graduate students, and alums awarded the Fulbright in 2024-25. President Pines also shared that the University has been recognized as a top institution 11 times by *The Chronicle of Higher Education* and more than 500 UMD students, alums, staff, and faculty have earned Fulbright awards since 1946.

### **Faculty Success**

President Pines shared that the University has 105 total memberships in the National Academies. There were 29 new memberships since 2020; the most recent being Dr. Yang Toa and Dr. Uzi Vishkin, both named to the National Academy of Inventors.

President Pines recognized Dr. Rita Colwell, Dr. Alan Leshner, and former President of the University of Maryland, Dr. C.D. (Dan) Mote, Jr., for holding three memberships in three different National Academies.

President Pines also recognized the four winners of the prestigious Presidential Early Career Award for Scientists and Engineers (PECASE): Dr. John Beieler, Dr. Damena Agonafer, Dr. Soheil Feizi, and Dr. Zohreh Davoudi. Pines noted this is the first time the University has had four winners in any cycle. There have been 17 faculty recognized since 1997.

### **Student Success**

President Pines said that faculty success is mirrored by student success, and shared that the University placed in the Top 10 in the prestigious Putnam Mathematical Competition. President Pines asked for a round of applause for the Math Department and its students.

President Pines acknowledged the work of Malcolm Maas, who will graduate this May with a degree in Atmospheric and OCeanic Sciences and Physics. Maas was awarded a Churchill Scholarship and previously won The Barry Goldwater Scholarship and Excellence in Education. Maas is 1 of 15 students nationwide to win a Churchill this year, and the 9th Terp in 7 years. Mass will pursue a master's degree in Math at Cambridge University in England.

President Pines also acknowledged the work of Ian Smith, a Bioengineering doctoral student and Clark Doctoral Fellow, who recently patented the "tumor on a chip" device to help clinicians observe how cancer cells detach and invade blood vessels. Pines said this is a great example of Terp ingenuity and will further the University's mission to personalize medical treatment and make the lives of every Marylander healthier.

President Pines shared that the Department of Animal and Avian Sciences welcomed two new dairy goats at the farm, Sadie and Ariana. The goats are the beginning of the first goat herd and will be the focus of a new goat management course. The goats were donated by the family of Cheyenne Van Echo, a 2024 Animal Science alum.

President Pines also reminded members about Maryland Day, which will be the 100th anniversary of the original "Ag Day".

Finally, President Pines extended a special thanks to Terp Thon, who's annual dance marathon fundraiser earned over \$240,000 this year, totaling in over \$6.5M since its founding in 2010. The

fundraiser benefits Children's National Hospital and supports pediatric patients and their families.

### Terp Pride

President Pines provided results from this year's Giving Day, which occurred on March 5, 2025. President Pines shared this was the largest giving day on record, with \$5.3M raised. There were 11,794 gifts from 8,949 donors. President Pines acknowledged the Maryland Hillel as the top fundraiser with \$1.1M.

President Pines also shared several Admissions highlights. There were more than 66,000 applications total, which is an 11% increase over last year. More than half of these students applied as test optional. There were 21,000 Fall 2025 early action admits who averaged SAT scores from 1440-1540, and 4.47 GPAs.

President Pines acknowledged the work of the Enrollment Management staff. A <u>video</u> of the #NowATerp Tour for the Class of 2029 was shared.

### **Current Events**

President Pines addressed the last six weeks with the new administration, and how it may impact the University. President Pines explained that the situation in Washington, D.C. represents an evolving and significant threat to higher education.

Dear College Letters Executive Orders may impact the community and academic programming. The University of Maryland's Government Affairs website has created a <u>site</u> for tracking what is happening and affecting the University, along with guidance and strategic response approaches.

President Pines shared that the University's strategic response is to engage, challenge, adapt, plan, and listen to the community.

The University will engage by continuing federal advocacy with MD Delegation, Agency Heads, and in partnership with AAU, ACE and APLU.

The University will challenge what is not legal by following and joining course cases and litigation in partnership with OAG.

The University will adapt by analyzing federal impact and make plans for moving forward by assessing state and federal budget implications and range of actions to lower expenses and increase revenues.

President Pines reminded members of a working group that was previously created out of guidance from the Senate, to work through the fiscal 2021 budget shortfall. President Pins shared that another working group will be formed with Senate leadership and develop a financial plan for the institution.

The University is also building marketing and communication plans to aid improve the public's opinion of higher education. The University will partner with AAU and APLU to explain the importance of the University's research to the public, by telling the stories more plainly.

President Pines concluded the state of the campus and Chair Sly opened the floor to questions.

Senator Wien, TTK, ARHU asked President Pines a question about the vision for the role of the

humanities at the University and how the humanities can be strengthened.

President Pines shared a forthcoming humanities initiative being developed with Provost Rice and the Arts & Humanities department. Despite his STEM background, Pines emphasized the critical importance of humanities in addressing societal challenges and providing ethical frameworks for technological advancement, citing the university's campus-wide approach to AI ethics as an example. He concluded by reaffirming the university's strong commitment to arts and humanities.

Senator Moradi, PTK, CMNS asked President Pines for an update regarding research faculty who are funded through cooperative institutes such as NASA or NOAA. Additionally, Senator Moradi asked President Pines a follow-up question about the university policy of providing 90 day or six months' notice before terminating research faculty positions and if the policy would be upheld in the event of government decisions or executive orders.

President Pines thanked Senator Moradi for raising the issue, and shared that leadership was in conversation with NASA and NOAA earlier in the day. President Pines explained that the University is aware of the partnerships that the Earth System Science Interdisciplinary Center (ESIC) has with NASA and NOAA.

President Pines suggested that faculty work through the Director of ESIC and the Dean of CMNS because as the working group moves forward, these issues will be reviewed as it relates to research faculty, and tenure track faculty. President Pines said that the University hopes to develop a strategy that will help support people going forward. They are also in conversation with the agencies to find out what is happening with their funding stream.

Senator Cole, TTK, ARHU, said that the English department graduate students were informed that half of their summer fellowships have disappeared. The English department could not supply matching funds. Cole asked if there is a mechanism to privilege graduate students in the times that we will be going into, and where they are the most vulnerable members of the UMD community?

President Pines thanks Senator Coles for representing the graduate student and for their leadership in the English department. President Pines said the University is focused on investing in its people and communities and will take the issue up through the Graduate School, as well as the proposed working group. President Pines reiterated that the University is committed to the excellence of its graduate students and supporting them going forward.

Senator Ivy Lyons, Graduate Student, JOUR, introduced Harriet Goers, a doctoral student in GVPT. Goers shared that they, and many of their colleagues, have not been paid in two months and detailed the financial burdens this has imposed. Goers said the problem has been cited as an issue with Workday implementation and asked why graduate students are the ones to bear this financial burden. Goers also question the University's priorities with regards to paying its employees.

President Pines thanked Senator Lyons and Goers for representing their fellow graduate students. President Pines was not aware of the graduate student payment issue addressed by Goers and offered to look into the problem. President Pines apologized on behalf of the University and acknowledged that this should not be happening.

Chair Sly thanked President Pines.

# PCC PROPOSAL TO RENAME THE BACHELOR OF SCIENCE IN "FAMILY SCIENCE" TO "FAMILY HEALTH" (SENATE DOCUMENT #24-25-28) (ACTION)

Chair Sly invited Wendy Stickle, Chair of the Program, Curricula and Courses Committee (PCC) to present on the proposal.

Stickle provided background and information on the proposal.

Chair Sly thanked Stickle and opened the floor to discussion of the proposal.

Hearing none, Chair Sly called for a vote on this proposal. The result was 99 in favor, 2 opposed, and 6 abstentions. **The proposal passed.** 

# IT COUNCIL REPORT: PROPOSAL FOR THE CHANGE IN UMD POLICY X-3.03(A) ON POLICY AND PROCEDURES CONCERNING THE ACQUISITION OF TELECOMMUNICATION SERVICES AND EQUIPMENT (SENATE DOCUMENT #23-24-28) (ACTION)

Chair Sly invited Jeff Klauda, Chair of the IT Council to present the report.

Klauda shared the IT Council's report proposed the following:

- Not merge the policy with broader UMD policies for a concern in loss of specific telecom details.
- Reorganize the policy to specify the purpose, then present definitions and end with references citing the recently updated cellular devices policy and a link to the IT service desk.
- Replace outdated terminology with generic terms like "voice, video and data transmission services".
- Including wording "including but not limited to" to avoid a policy to become quickly obsolete.

Klauda shared that the IT Council recommends the Senate Approve the updated policy as included in the Senate Meeting Materials, noting this will put the policy up to date with current DIT procedures, modernize the terminology and provide flexibility for future changes in technology.

Chair Sly thanked Klauda and opened the floor to discussion of the proposal.

Naomi Feldman, TTK, ARHU, noted that there are units who run their own IT systems, and asked for clarification on the requirement for obtaining approval for all communications equipment. Feldman asked if the policy recommendation is changing the current procedures from the last semester or formalizing what was already happening?

Klauda confirmed the latter.

Chair Sly asked for further discussion. Seeing no further discussion, Chair Sly called for a vote on the proposal. The result was 88 in favor, 3 opposed, and 10 abstentions. The proposal passed.

### SPECIAL ORDER OF THE DAY

### Jack Blanchard Associate Provost for Enterprise Resource Planning *Elevate Update*

Chair Sly introduced Associate Provost for Enterprise Resource Planning, Jack Blanchard, who provided an update on the Elevate program.

Blanchard provided an update on Workday, sharing that the system is in a period of stabilization, which includes learning, identifying issues, making improvements, and fixing problems as they are identified. Blanchard acknowledged the challenges with using a new system, such as issues with workflows, reports, business processes, security roles, etc. and assured members that this is a normal part of such a massive transformation.

Blanchard noted that the Elevate team is reviewing all service tickets and data, in addition to talking with leadership in Divisions and central units. The Elevate team is also reviewing the responses collected from the user survey, which produced over 1,100 responses.

Blanchard shared Workday performance data, noting that over 1,000 hires were processed through workday in January. The data indicates that the system is doing the work it is supposed to, and that there has been a decline in the submission of new support tickets in February.

Blanchard further identified pain points from the implementation of Workday as providing support in a timely fashion, reporting, security training, and processes such as hiring and approvals. The ERP team is reviewing processes to see how they can be improved. Blanchard explained that implementing solutions for these concerns is an ongoing, iterative process.

Blanchard also shared that the Elevate Enterprise Team has completed 49 enhancements which can be found using the tracker on <u>elevate.umd.edu</u> as well as new issues and solutions. These solutions and improvements have included increasing the skills in the staff which will allow them to take on more tickets and reduce wait time. Training materials and resources, such as quick start guides, have been developed, and virtual sessions and office hours are being offered. Blanchard reminded members that <u>elevate.umd.edu</u> is the best source of information and to access these resources.

Next, Blanchard provided updates on the implementation of Workday Student. The current student information system has components which are over 30 years old and no longer meet the needs of our students; it was the driver for modernizing the systems. Workday will serve across the life cycle of students when they come to campus and will benefit staff and faculty as well. The new system will have 24/7 mobile access, and students will have greater control about their personal identity information. They will have dynamic academic planning which will show them what they need to do next and provide clear financial information, all of which will be accessible from their phone. Blanchard emphasized that this will be another heavy lift. Currently the program is in the design phase, with over 80 staff and subject matter experts working with the Elevate team in the design sessions.

Blanchard reported that in early 2027, the first "go live" of functionality of Workday Student will occur in the summer of 2027, with a total of five "go lives", delivering functionality through the fall of 2028.

Chair Sly thanked Blanchard and opened the floor to discussion.

Senator Espindola, TTK, CMNS, shared concerns that Workday did not utilize input from the Senate

or the public, and asked if the Student Workday was receiving input for its implementation and the willingness of students to accept it?

Blanchard responded, sharing that there are project implementation teams involving staff members from every college and every unit, as well as customer confirmation sessions. From these sessions information was gained about engaging subject matter, experts, local perspectives, and the students. Blanchard explained that Student Workday is more complicated than HCM and Finance Workday because of its complexities and nuances. The Elevate team is engaging with subject matter experts and had a meeting with over 150 attendees representing every college, division, and school. The design sessions are ongoing. Blanchard also shared that the Provost has asked the Elevate team to focus on Academic Affairs and engaging at a local level. Their goal is to ensure those at the local level recognize and understand what's being delivered.

Senator Yaich, Non-Exempt Staff, CMNS, asked about prioritization of tickets. Senator Yaich shared that non-exempt employees are also facing issues to Workday with timely payments, citing concerns with the Fair Labor Standards Act (FLSA).

Blanchard confirmed that the highest priority is payment, and tickets requesting assistance with payment issues are at the highest priority level. Blanchard continued that Workday should not be an explanation for why someone is not being paid for a period of time. There have been circumstances where issues have been identified, and emergency payments are made. Blanchard said that payment issues are escalated quickly by the team to be resolved, and that Workday should not be the explanation or reason why someone is not being paid, given what they are monitoring within the system.

Chair Sly thanked Blanchard.

### CONTINUED BUSINESS

Chair Sly opened the floor for continued business and provided a summary of the Adoption of Special Rule of Procedure Regarding Resolutions (Senate Document #24-25-26). Chair Sly explained that this adoption means any new business that's brought substance and needs voting would be pushed automatically into continuing business.

Chair Sly invited the proposer, Senator Miller, PTK, CMNS to address any questions.

Senator Miller explained that this is a fairly commonly adopted Special Rule, as it allows Senators to review what's proposed at a meeting, before the meeting. Senator Miller reminded members that there have been Resolutions that were brought to the Senate for a vote, however, documents were not available beforehand for review. Senator Miller noted that they believe the chances of adoption will go up if Senators can read about an item before voting.

Chair Sly called for a motion to adopt the rule as stated. A motion was made and seconded. Chair Sly asked if there was any further discussion.

Senator Stillwell, PTK, PLCY, asked why this wasn't already the procedure, and shared support for the proposal.

Chair Sly explained that this change means if new business is of substance and requires additional thought, it will automatically move to continuing business to allow more time for documentation to be

provided and reviewed. Chair Sly clarified that this is a change to current procedure because it is not the standard practice in Robert's Rules of Order.

Chair Sly reminded the Senate that this would be a change to the Bylaws and therefore requires a two-thirds majority vote to pass.

Chair Sly restated that there was a motion and a second and moved to a vote. The result of the vote was 85 in favor, 3 opposed, and 3 abstentions. **The Adoption of the Special Rule passed.** 

### **NEW BUSINESS**

There was no new business

### **ADJOURNMENT**

The meeting was adjourned at 5:00 p.m.

### REPORT | #24-25-11

Senate Elections, Representation, & Governance (ERG) Committee

### Review of the University Senate Bylaws, Committee Chair Selection

### 2024-2025 Committee Members

Kim Gonzalez (Chair)
Janet Adesina (Graduate Student)
Paul Brown (Faculty)
Sophia Conrad (Undergraduate Student)
Jon Crocker (Faculty)
Tom Hatcher (Faculty)
Lian Kish (Exempt Staff)
Suraj Krishnamurti (Graduate Student)
Jessica O'Hara (Faculty)

Polly O'Rourke (Faculty)
Pamela Phillips (Ex-Officio- Associate VP IRPA Rep)
Madeline Reynoso (Undergraduate Student)
Sajeela Yaqub (Faculty)

Date of Submission February 28, 2025

### **BACKGROUND**

At the September 5, 2024 University Senate Meeting, Senator Hajiaghayi, TTK, CMNS inquired about amending the Bylaws to allow committees to nominate candidates for committee chair<sup>1</sup>, with Senate Chair selecting final candidate. The Senate voted in favor of charging the Elections, Representation, & Governance (ERG) Committee to review the University Senate Bylaws, to the purpose of reviewing committee chair selection.

On September 5, 2024, the Senate Executive Committee (SEC) provided a charge document (Appendix 1) to the ERG Committee that included reviewing section 5.5i of the University Senate Bylaws (Appendix 2) and similar policies and procedures at Big 10 peer institutions, consulting with stakeholders including the original proposer and Senate Leadership, and considering the impact on committee membership, procedure, and replacements.

### COMMITTEE WORK

The ERG Committee began its consideration of the charge in October of 2024. The committee reviewed the Bylaws and Plan of Organization, discussing current committee chair selection process and limitations, including individual interest, annual transition timing, and requirements as stipulated in the Bylaw (specifically that committee chairs must be among current Senators for the academic year of planned service).

The ERG Committee completed Big 10 peer research, which revealed diverse policies and practices in committee selection process. This diversity often stemmed from a difference in overall Senate structure, as well as variations in committee and leadership types, between University of Maryland and other peer institutions.

However, the committee discussed these other practices for committee chair selection, and found that where equitable comparisons were equitable, only five schools used a type of "elected"

<sup>&</sup>lt;sup>1</sup> Committee chair is the familiar verbiage for this role, but the official language used in the Bylaws is *presiding officer of the committee*. When referencing committee discussion this report will use the term familiar to members, *committee chair*, and when referencing any official language or Bylaw changes will use the term *presiding officer*.

process, and ten schools used a type of "appointment process". It was decided that the University of Maryland falls within the majority by maintaining an appointment process for committee leadership.

The ERG Committee did take note of exemplary policies within Big 10 shared governance, which included prioritizing past committee members or chairs for leadership role or allowing nominations to take place before selecting committee chairs.

After familiarizing themselves with the proposal and research of peer institutions, the committee then invested time in consulting with stakeholders about the current practice of chair selection, as well as potential for changes.

This included consultations with the Senator who originally proposed the motion to the Senate floor, the University Senate's Parliamentarian, Chair and Chair-Elect, as well as Faculty Affairs Committee, Staff Affairs Committee, Student Affairs Committee, and Committee on Committees.

The general feedback from all committees was that the current process as outlined in the University Bylaws had benefits and potential for improvement, but that any amendments to that would also have concerns as well as potential merits.

The Committee on Committees shared consensus that time constraints is the main concern with any procedural change, and members would feel burdened with any timeline adjustment to the committee selection process.

Senate Leadership shared concerns that timeline, as well as individual interest, were barriors to any change in selection/election process. Senate Leadership shared an ideal procedure would include many interested candidates being elected for the role, but the reality is that there are limited number of interested parties who are both viable (as current Senators) and interested in the additional workload. Senate Leadership shared that efforts for a full committee chair election are further constrained by the limited timeline between of Senate operations. Between the May transition meeting, where new Senators begin their term, the summer semester where many faculty are off contractual hours, and the beginning of the academic year when committees begin work, leave an open time frame that is not feasible for a full election. Senate Leadership was supportive of a process where individuals can nominate, and help garner interest, thus removing one of the current challenges of finding committee chairs.

However, the consultations also revealed interest in a more transparent process, where individuals are given the opportunity to express interest in the role. The original proposer shared hope that in the interest of the democratic process, nominations for committee chair could be considered when selecting the role. The proposer also shared knowledge of the limiting factors, and remained comfortable with Senate Leadership (Senate Chair-Elect, as per the Bylaws) making the final choice in committee chairs.

The other consultations concurred with interest in transparency. Faculty, staff, and student groups were consulted via asynchronous feedback form, and shared general views that it would be beneficial to do the following: allow Senators to indicate interest in committee leadership, to include more information and communication about committee chair selection and move toward a practice where the Senate Chair-Elect selects committee chairs from amongst a prescribed, interested group.

Considering the peer groups, consultations, and logistical implications the ERG Committee voted to make no amendments to the current University Senate Bylaws.

The ERG Committee did elect to include recommendations to the Senate Office on operational elements that reflect the constituency recommendations and the ERG Committee's beliefs on shared governance.

The recommendations were discussed and finalized at the January 28, 2025 ERG Committee Meeting. These recommendations are listed below in the Recommendations section.

The University Senate Bylaws were not reviewed by the Office of General Counsel, as the ERG Committee is proposing no change to the current Bylaws.

### RECOMMENDATIONS

The ERG Committee recommends that no change be made to the <u>University of Maryland Senate Bylaws</u>, and that the following recommendations be considered for the next academic year in the selection of committee presiding officers.

The ERG Committee recommends that, in the selection process, Senate Leadership and the University Senate Office prioritize the following by implementing a "Call for Service" to all incoming and continuing Senators, ideally in the Spring of the academic year that precedes the committee chairs' terms of service.

During the process of soliciting nominations the committee recommends the Chair-Elect and Senate Leadership should consider, in the following order of priority:

- Those with previous experience on the committee that individual would be presiding over.
- Experience on the Senate (continuing or previously served Senators)
- Experience with the subject matter of that committee that they will be presiding over, for incoming Senators who have not had the aforementioned experience
- The inclusion of both self-nomination as well as peer-nominations (meaning Senator to Senator) to encourage the inclusion of diverse personalities
- The Senate Chair-Elect will thoroughly consider those individuals self-nominated before proposing external candidates
- The final selection of committee chair, or presiding officer, will ultimately lay with the Senate Chair-Elect

### **APPENDICES**

Appendix 1 - Charge from the Senate Executive Committee

Appendix 2 - University Senate Bylaws



### **CHARGE**

Charged: September 5, 2024 | Deadline: February 28. 2025

### Review the University Senate Bylaws, Committee Presiding Officer Selection (Senate Document #24-25-11) Elections, Representation & Governance | Chair: Kim Gonzalez

The University Senate and Senate Chair Sly request that the Elections, Representation & Governance (ERG) Committee review the proposal entitled *Review the University Senate Bylaws, Committee Presiding Officer Selection*.

Specifically, the ERG Committee should:

- 1. Review the following:
  - a. <u>University Senate Bylaws</u>, specifically Section 5.5.i, concerned with the appointment of presiding officers of the University Senate.
  - b. The <u>Plan of Organization for Shared Governance</u>, specifically Article 7 concerned with the University Senate Bylaws.
  - c. Similar policies and procedures on presiding officer selection at Big 10 and other peer institutions.

### 2. Consult:

- a. University Senate Leadership on the current procedures, timelines, and logistics for the presiding officer selection.
- b. The University Senate's Faculty Affairs, Staff Affairs, and Student Affairs Committees to solicit feedback on presiding officer selection.
- c. The original proposer of the Senate Bylaw Review on Committee Presiding Officer Selection
- 3. Consider the impact on the University Senate Bylaws, specifically Section 5.1 through 5.6, concerned with membership, procedure, and replacements of University Senate Committees.
- 4. If appropriate, recommend whether the University Senate Bylaws should be revised and if so, provide suggested revisions.

We ask that you submit a report to the University Senate Office no later than **February 28, 2025**. If you have questions or need assistance, please contact the University Senate Office, senate-admin@umd.edu.

Appendix 2- University Senate Bylaws

**BYLAWS** 

**OF THE** 

**UNIVERSITY SENATE** 

**AT THE** 

**UNIVERSITY OF MARYLAND** 



# BYLAWS OF THE UNIVERSITY SENATE University of Maryland, College Park TABLE OF CONTENTS

ARTICLE 1 - AUTHO	ORIZATION	3
ARTICLE 2 - MEMB	ERSHIP	3
<b>ARTICLE 3 - MEETI</b>	NGS	5
ARTICLE 4 - EXECU	JTIVE COMMITTEE	6
ARTICLE 5 - COMM	IITTEES OF THE SENATE	8
ARTICLE 6 - STAND	DING COMMITTEE SPECIFICATIONS	11
6.1 Academic P	Procedures and Standards Committee	11
	airs Committee	
	on Committees	
	Affairs Committee	
	Representation, and Governance Committee	
	ersity, & Inclusion Committee	
	irs Committee	
	Curricula, and Courses Committee	
	airs Committee	
	nduct Committee	
	AL COMMITTEE ON UNIVERSITY FINANCE	
<b>ARTICLE 8 - UNIVE</b>	RSITY COUNCILS	20
<b>ARTICLE 9 - UNIVE</b>	RSITY COUNCIL SPECIFICATIONS	22
9.1 University L	ibrary Council	22
	tesearch Council	
9.3 University IT	Γ Council	23
ARTICLE 10 - THE	ATHLETIC COUNCIL	24
ARTICLE 11 - DUTII	ES OF THE EXECUTIVE SECRETARY AND DIRECTOR	25
ARTICLE 12 - ANNU	JAL TRANSITION OF THE SENATE	26
APPENDIX 1 - BYLA	AWS OF THE UNIVERSITY LIBRARY COUNCIL	28
	AWS OF THE UNIVERSITY RESEARCH COUNCIL	
APPENDIX 3 - BYLA	AWS OF THE UNIVERSITY IT COUNCIL	31
APPENDIX 4 - CHAI	RTER OF THE UNIVERSITY ATHLETIC COUNCIL	35
APPENDIX 5 - PRO	CEDURES FOR ELECTIONS OF UMCP REPRESENTATIVES TO THE COUNCIL	ı
	JNIVERSITY SYSTEM FACULTY (CUSF)	
APPENDIX 6 - PRO	CEDURES FOR ELECTIONS OF UMCP REPRESENTATIVES TO THE COUNC	IL OF
	VERSITY SYSTEM STAFF (CUSS)	42
APPENDIX 7 - PRO	CEDURES FOR REVIEW OF COLLEGE AND SCHOOL PLANS OF	4.5
	GANIZATION	
Dates of Approval,	Updates and Amendments to the Senate Bylaws	45

### BYLAWS OF THE UNIVERSITY SENATE The University of Maryland, College Park

### ARTICLE 1 AUTHORIZATION

1.1 These *Bylaws of the University Senate* (hereafter referred to as the *Bylaws*) are adopted according to Article 7 of the *University of Maryland Plan of Organization for Shared Governance* (hereafter referred to as the *Plan*), and are subject to amendment as provided for in the *Plan*.

# ARTICLE 2 MEMBERSHIP

2.1 The members of the Senate are as designated in Article 3 of the *Plan a*nd further specified in 2.1 and 2.2 below. All elected members are subject to the conditions stated in the *Plan,* including its provisions for expulsion, recall, and impeachment (Articles 4.10, 4.11, and 5.8 of the *Plan* and Article 2.3, 2.4, and 2.5 below).

#### 2.1.a Staff Senators

For the purpose of Senate representation, the Staff Constituency is divided into the following categories. Each category shall elect one Senator from among its ranks for each 200 staff members or major fraction thereof.

- 1. Exempt Staff with appointment in Colleges, Schools, and Academic Affairs
- 2. Exempt Staff with appointment in Divisions
- 3. Non-Exempt Staff with appointment in Colleges, Schools, and Academic Affairs
- 4. Non-Exempt Staff with appointment in Divisions
- 2.1.b Staff member job categories will not include the category designated for the President, vice presidents, provosts, and deans if they hold faculty rank.
- 2.1.c Any individual within the faculty member voting constituency cannot be included in the staff member voting constituency or nominated for election as a staff Senator. Staff candidates for the Senate must have been employed at the University of Maryland College Park for 12 months prior to standing as candidates for the Senate. Staff members may not stand for Senate elections while in the probationary period of employment.
- 2.1.d An ex officio member denoted in the *Plan* (Article 3.6.a.) who is not precluded from staff member categories as noted in Articles 2.1.b and 2.1.c may be elected as a voting member of the Senate by an appropriate constituency. Such ex officio members should also have been employed by the University of Maryland College Park for 12 months prior to standing as candidates for the Senate.
- 2.1.e As noted in the *Plan* (Article 3.3.c), the term of each staff Senator shall be three (3) years. Terms of staff members will be staggered in such a way that for each term, one-third of the total members from a job category are serving the first year of their term. Not every member of a specific staff job category shall be elected in the same year. However, if the University or these *Bylaws* redefine the staff job categories outside of a normal reapportionment, the staff Senate seats will be vacated. A subsequent election will be held to populate all staff Senate seats within the new categories with staggered terms as follows:
  - (1) One-third of the members in a job category who received the lowest number of votes will serve a one-year term,
  - (2) One-third of the members in a job category who received the second lowest number of votes will serve two-year terms,
  - (3) One-third of the members in a job category who received the highest number of votes will serve three year-terms.

A person serving less than a three-year term is defined as not to have served a full term and is eligible for re-election to a full term the following year.

### 2.2 Single Member Constituencies

The Senators defined in (a)-(g) below shall be voting members of the Senate. All elections held pursuant to this section shall be organized by the Office of the University Senate.

- (a) Part-Time Research, Part-Time Teaching, Adjunct, and both Full-Time and Part-Time Visiting Faculty who are not members of the Faculty Constituency as defined in Section 3.2 of the Plan shall together elect one (1) Senator, for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as a Faculty Senator.
  - (b) Emeriti Faculty who are not members of the Faculty Constituency as defined in Section 3.2 of the *Plan* shall elect one (1) Senator from among their ranks for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as a Faculty Senator.
- (c) Head Coaches who are not members of the Faculty Constituency as defined in Section 3.2 of the University Plan of Organization together shall elect one Senator from among their ranks to serve for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as a faculty Senator.
- (d) Post-Doctoral Scholars, Post-Doctoral Associates (formerly Research Associates), Junior Lecturers, and Faculty Assistants (formerly Faculty Research Assistants) who are not members of any Senate constituency as defined in Article 3 of the *Plan* together shall elect one (1) Senator, for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as a Faculty Senator.
- (e) The Contingent II staff shall elect one (1) Senator from among their ranks for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as all other staff Senators. The Contingent II staff Senator shall have been employed by the University for twelve months prior to their election.
- (f) The part-time undergraduate students shall elect one (1) Senator from among their ranks for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as all other student Senators.
- (g) The part-time graduate students shall elect one (1) Senator from among their ranks for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as all other student Senators.
- 2.3 Elected Senators shall not be absent from two (2) consecutive regularly scheduled meetings of the Senate without notifying the Office of the University Senate that they will require an excused absence (Article 4.10.a of the *Plan*). The Senator shall be counted in the total membership when a quorum is defined for a meeting unless that Senator is expelled.
- 2.4 If an elected Senator is no longer a member of the constituency by which they were elected, the seat may be vacated and the Senator may be replaced according to the following guidelines:
  - 2.4.a If there was a runner-up in the election in which the Senator was elected, the runner-up shall replace that Senator immediately, provided they are still eligible.
  - 2.4.b If there was no runner-up in the election in which the Senator was elected and the vacancy occurs in the spring semester, that Senator shall serve for the remainder of the Senate year and shall be replaced in the next election cycle for the remainder of the term.
  - 2.4.c If there was no runner-up in the election in which the Senator was elected and the vacancy occurs prior to the spring semester, or if the Senator is unable to serve the remainder of the Senate year, the

Senate Executive Committee, in consultation with the appropriate constituency, shall appoint a replacement for that Senator.

- 2.5 If an elected Senator is no longer in satisfactory standing at the University, they shall be replaced immediately upon notification to and verification by the Office of the University Senate in accordance with 2.4.a or 2.4.c above.
- 2.6 All elections shall be completed by the Transition Meeting of the Senate.

# ARTICLE 3 MEETINGS

### 3.1 Regular Meetings:

The Senate shall schedule at least four (4) regular meetings each semester. The notice, agenda, and supporting documents will normally be provided by the Office of the University Senate to the membership one week prior to each regular meeting unless otherwise approved by the Executive Committee.

### 3.2 Special Meetings:

- 3.2.a Special meetings of the Senate may be called in any of the following ways, with the matter(s) to be considered to be specified in the call:
  - (1) By the presiding officer of the Senate;
  - (2) By a majority vote of the Executive Committee of the Senate;
  - (3) By written petition of a majority of the elected members of the Senate. The petition shall be delivered to the Chair or the Executive Secretary and Director of the Senate. The Chair shall give notice of arrangements for the meeting within seventy-two (72) hours of receipt of a valid petition; or
  - (4) By resolution of the Senate.
- 3.2.b The notice of a special meeting shall include the agenda and shall be sent to the members of the Senate as far in advance of the meeting as possible. The agenda of a special meeting may specify a scheduled time of adjournment.
- 3.2.c The scheduling of a special meeting shall reflect the urgency of the matter(s) specified in the call, the requirement of reasonable notice, and the availability of the membership.

### 3.3 Openness of Meetings and Floor Privileges:

- 3.3.a Meetings of the Senate shall be open to all members of the campus community except when the meetings are being conducted in closed session.
- 3.3.b Representatives of the news media shall be admitted to all meetings of the Senate except when the meetings are conducted in closed session. The use of television, video, or recording equipment shall not be permitted except by express consent of the Senate.
- 3.3.c When a report of a committee of the Senate is being considered, members of that committee who are not members of the Senate may have a voice but not a vote in the deliberations of the Senate on that report.
- 3.3.d Any Senator may request the privilege of the floor for any member of the campus community to speak on the subject before the Senate. The Chair shall rule on such requests.
- 3.3.e By vote of the Senate, by ruling of the Chair, or by order of the Executive Committee included in the agenda of the meeting, the Senate shall go into closed session. The ruling of the Chair and the order

- of the Executive Committee shall be subject to appeal, but the Chair shall determine whether such appeal shall be considered in open or closed session.
- 3.3.f While in closed session, the meeting shall be restricted to voting members of the Senate (Article 3 in the *Plan*), members granted a voice but not a vote (Articles 3.6, 5.2.c, and 5.5.c. of the *Plan*), the Executive Secretary and Director, the Parliamentarian, the immediate Past Chair of the Senate, any staff required for meeting operations, and other persons expressly invited by the Senate Chair.

### 3.4 Rules for Procedure:

- 3.4.a The version of *Robert's Rules of Order* that shall govern the conduct of Senate meetings shall be *Robert's Rules of Order*, *Newly Revised*.
- 3.4.b A quorum for meetings shall be defined as a majority of elected Senators who have not submitted an excused absence to the Office of the University Senate, or sixty (60) Senators, whichever number is higher. For the purpose of determining a quorum, ex officio members without vote shall not be considered.
- 3.4.c Voting shall be restricted to eligible members of the Senate (Article 3 in the *Plan*) who are participating in the Senate meeting at the time of the vote.

## ARTICLE 4 EXECUTIVE COMMITTEE

### 4.1 Membership and Election:

- 4.1.a As set forth in the *Plan* (Article 8.2), the members of the Executive Committee shall include the Chair and Chair-Elect of the Senate, thirteen (13) members elected from the voting membership of the Senate, and five (5) non-voting ex officio members.
- 4.1.b The election of the Executive Committee should follow the election of the Chair-Elect as provided for in the *Plan* (Article 5.3 and 5.7.a). In the event of a tie vote in the election for members of the Executive Committee, a ballot will be made available to each Senator in the appropriate constituency as soon as the votes are counted and the tie discovered. The election to break the tie should end one (1) week from the start date.
- 4.1.c In the event of a vacancy on the Executive Committee, the available candidate who had received the next highest number of votes in the annual election for the Executive Committee shall fill the remainder of the unexpired term.

### 4.2 Charge: The Executive Committee shall exercise the following functions:

- 4.2.a Assist in carrying into effect the actions of the Senate;
- 4.2.b Act for the Senate as provided for by and subject to the limitations stated in Article 4.3;
- 4.2.c Act as an initiating body suggesting possible action by the Senate;
- 4.2.d Assist in the administrative implementation of policies adopted by the Senate, as needed;
- 4.2.e Prepare the agenda for each Senate meeting as provided for by and subject to limitations stated in Article 4.4;
- 4.2.f Serve as a channel through which any member of the campus community may introduce matters for consideration by the Senate or its committees;
- 4.2.g Ensure that information on the Senate's work each year is available to the campus community through the Senate website and reported to the President;

- 4.2.h Provide feedback on the operations of the Office of the University Senate, and make recommendations to the President or his or her designee for improvements in those operations and for suitable candidates for the role of the Executive Secretary and Director;
- 4.2.i Serve as the channel through which the Senate and the campus community may participate in the selection of administrators at the University;
  - 4.2.j Perform such other functions as maybe given it in other provisions of these *Bylaws* and the *Plan*; and
- 4.2.k Make recommendations on nominees for campus-wide and system-wide committees and councils requiring representatives, when necessary.

### 4.3 Rules Governing Executive Committee Action for the Senate:

- 4.3.a Where time or the availability of the membership precludes a meeting of the Senate, as, for example, during the summer or between semesters, the Executive Committee may act on behalf of the Senate.
- 4.3.b A report of all actions taken by the Executive Committee when acting on behalf of the Senate, with supporting material, shall be included with the agenda of the next regular meeting of the Senate. By written request of ten (10) Senators, received by the Chair of the Senate prior to the call to order of that meeting, any Executive Committee action on behalf of the Senate shall be vacated and the item in question placed on the agenda for that meeting. If any such item is not petitioned to the floor, it shall stand as an approved action of the Senate.

### 4.4 Rules Governing Preparation of the Senate Agenda:

- 4.4.a The order of business for regular meetings of the Senate shall be set by the Executive Committee. Typically, the order of the agenda will follow *Robert's Rules of Order*, but the Executive Committee maintains authority to reorder the agenda based on the needs of the Senate.
- 4.4.b For regular meetings the Executive Committee shall consider all submissions for inclusion on the Senate agenda. The Executive Committee may not alter a submission, but may delay its inclusion, may include it on the agenda of a special meeting, may submit the material directly to a committee of the Senate, or may refuse to place it on the agenda if the material is inappropriate, incomplete, or unclear. The party making a submission shall be notified of the action taken in this regard by the Executive Committee.
- 4.4.c The order of business for a special meeting of the Senate shall be set by the Executive Committee.
- 4.4.d For a special meeting the agenda shall include the matter(s) specified in the call of that meeting as the Special Order. Other items may be included on the agenda as the Executive Committee deems appropriate.
- 4.5 **Meetings of the Executive Committee**: A quorum of the Executive Committee shall be eight (8) voting members. Minutes of the meetings shall be kept. The agenda shall be made publicly available prior to each meeting. The Executive Committee shall meet at the call of the Chair or by petition of eight (8) voting members of the Executive Committee, or by petition of twenty-five (25) voting members of the Senate.
- 4.6 **The Senate Budget**: The Executive Secretary and Director shall be responsible for the Senate budget, shall consult with the Senate Chair on the preparation of the budget request, and shall report to the Executive Committee on the status of the budget.
  - 4.6.a The Executive Secretary and Director shall make an annual report to the Associate Vice President for Personnel and Budget on expenditure of the Senate budget.
  - 4.6.b Consent of the Associate Vice President for Personnel and Budget shall be required before any change in the budgeted use of Senate funds involving more than ten percent (10%) of the total may be undertaken.

- 4.7 Referral of Items to Standing Committees: The Executive Committee shall refer items to the standing committees.
  - 4.7.a The Executive Committee shall refer an item to an appropriate committee when instructed by the Senate or when requested by the President, or when petitioned by 150 members of the Senate electorate.
  - 4.7.b The Executive Committee may also refer any item it deems appropriate, and the standing committee shall give due consideration to such requests from the Executive Committee.
  - 4.7.c The Chair of the Senate may, as need requires, act for the Executive Committee and refer items to standing committees. All such actions shall be reported at the next meeting of the Executive Committee.
- 4.8 To the extent permitted by law and University policy, the records of the Senate shall be open.

# ARTICLE 5 COMMITTEES OF THE SENATE

- 5.1 **Standing Committees Specifications:** The specifications of each standing committee of the Senate shall state its name, its specific charge, and any exceptions or additions to the basic charge to standing committees stated in Article 5.2. The specifications shall list all voting ex officio members and shall define committee composition.
  - 5.1.a Standing Committees: In an appropriate section of Article 6 there shall be specifications for each committee.
- 5.2 **Standing Committees Basic Charge:** In its area of responsibility, as defined in its specifications, each committee shall be an arm of the Senate with the following powers:
  - (1) To formulate and review policies to be established by the Senate according to the *Plan* (Article 1);
  - (2) To review established policies and their administration and to recommend any changes in policies or their administration that may be desirable;
  - (3) To serve in an advisory capacity, upon request, regarding the administration of policies;
  - (4) To function on request of the President or of the Executive Committee as a board of appeal with reference to actions and/or decisions made in the application of policies; and
  - (5) To recommend the creation of special subcommittees (Article 5.7-5.9) when deemed necessary.

### 5.3 Standing Committees - Committee Operation:

- 5.3.a Agenda Determination:
  - (1) Issues within the committee's purview shall be referred and charged to the committee by the Executive Committee (Article 4.7). Such matters should take priority on the committee's agenda.
  - (2) A committee shall have principal responsibility for identifying matters of present and potential concern to the campus community within its area of responsibility. Such matters should be placed on the agenda of the committee.
  - (3) Committee agendas shall be made publicly available prior to each meeting.
- 5.3.b Minutes of the proceedings of each committee meeting shall be kept.

- 5.3.c Rules for Procedure of Standing Committees: Standing Committees are typically governed by *Robert's Rules of Order for Small Committees, Newly Revised.* Standing Committees shall determine how technology, such as video conferencing and other electronic methods of participation, can be used for their purposes. Standing Committees may choose to conduct votes online or via email, and shall agree on any other mechanisms for conducting business outside of meetings, when necessary.
- 5.3.d Quorum Requirements of Standing Committees: Unless a quorum number is specified in the membership description of a committee, the quorum shall be a majority of voting members of the committee.
- 5.4 **Standing Committees Reporting Responsibilities:** Each committee shall be responsible through its presiding officer for the timely delivery of the following reports.
  - 5.4.a Information on the schedule of committee meetings stating the date, time, and location should be made available on the Senate. The agenda for each meeting shall be made available on the Senate website as far in advance of the meeting as possible.
  - 5.4.b The committee shall report its progress on agenda items as required by the Executive Secretary and Director or the Chair of the Senate.
  - 5.4.c Reports providing information and/or recommendations to the Senate shall be submitted to the Executive Committee for inclusion on the Senate agenda. Reports resulting from the committee's advisory or board of appeals function shall be submitted to the appropriate Senate or campus officer, and the Executive Committee notified of the submission.
  - 5.4.d Upon written request of at least four (4) members of a committee, the presiding officer of that committee shall include a minority statement with any committee report. Those requesting inclusion need not support the substance of the minority statement.
  - 5.4.e An annual report shall be presented to the Chair of the Senate at the end of the academic year, or, if approved by the Chair, no later than August 16, for submission to the Executive Committee. The report shall include a list of all items placed on the committee's agenda, noting the disposition of each and a summary of the committee's deliberations. An overview of the committee's past work shall be made available on the Senate website. In the case of committees with little activity, the committee may recommend inactive status the ensuing year until charged by the Executive Committee to address a specific matter:
    - (1) A committee maybe placed on inactive status with approval of the Executive Committee. No presiding officer or members shall be appointed to the committee while on inactive status.
    - (2) A committee on inactive status may be reactivated by the Executive Committee when matters within its purview, as stipulated in Article 6, are brought to the Executive Committee for review. Following reactivation, the Office of the University Senate shall solicit volunteers for the committee in its annual volunteer period, and the Committee on Committees shall select members for the committee, in accordance with the provisions of 5.5 below.
    - (3) A Special Committee (Article 5.9) may not be appointed to consider matters within standing committee specifications in lieu of reactivating an inactive committee.
- 5.5 **Standing Committees Selecting Members:** Persons shall be named to standing committees in accordance with the procedures listed below.
  - 5.5.a The Office of the University Senate shall provide information on the charge and membership specifications of each committee.
  - 5.5.b The Office of the University Senate shall solicit volunteers for the Senate's standing committees on an annual basis through an online process. During this volunteer period, all faculty, staff, and students shall be eligible to indicate their top three preferences for any committees with vacancies in their

- constituency and include a candidacy statement for consideration by the Committee on Committees. The Office of the University Senate will maintain these records for potential future use.
- 5.5.c The Committee on Committees shall develop slates of nominees to fill vacancies on the standing committees and University Councils. No person shall be nominated for a committee position without consenting to serve on that committee, either through indicated preference or explicit agreement. In making nominations, the Committee on Committees shall keep in view the continuing membership of the committee to ensure that the full membership complies with specifications of the *Plan* and these *Bylaws*. Committee members shall be nominated consistent with requirements for diversity specified in Section 8.1 of the *Plan*.
- 5.5.d Ex officio members named in a committee's specifications shall be voting members unless otherwise specified in the *Bylaws*. Upon recommendation of the Committee on Committees, the Executive Committee may appoint ex officio members with particular expertise or benefit to the committee. Such members shall serve with voice, but without vote. The Executive Committee is empowered to make such changes in non-voting ex officio membership as appropriate.
- 5.5.e The Committee on Committees shall forward a slate of nominees for committee service to the Executive Committee to place on the Senate agenda for approval. Each nominee shall be identified by name and constituency. The notice of nomination shall also include the name and constituency of continuing members of the committee, and the name and office of the ex officio members, listed for information only. The nominations shall be subject to action by the Senate consistent with the *Plan* and the specifications of these *Bylaws*.
- 5.5.f Terms on standing committees shall be two (2) years for faculty and staff, and one (1) year for students. Appointments to two-year terms shall be staggered: that is, as far as practical, half of the terms from each faculty or staff constituency shall expire each year. Terms shall begin on July 1 of the appropriate year.
- 5.5.g A member of a standing committee whose term is expiring may be appointed to another term, subject to restrictions (1) and (2) below. The Committee on Committees is particularly charged to consider the reappointment of active student members.
  - (1) No reappointment shall be made that would cause the appointee to serve longer thanfour consecutive years on the same committee.
  - (2) At most, half of the non-student members of a committee whose terms are expiring in any given year may be reappointed.
- 5.5.h Terms as presiding officer of a committee shall be one year. A presiding officer may be reappointed if his/her tenure as a Senator is continuing; however, no one shall serve as presiding officer of a committee for longer than two (2) consecutive years.
- 5.5.i Appointments of the presiding officers of committees shall be made by the Chair of the Senate, designated on the annual committee slate, and shall be approved by the Senate.
- 5.6 Standing Committees Replacing Presiding Officers and Members: The presiding officer and members of any active standing committee may be replaced for cause after inquiry by the Office of the University Senate with approval of the Executive Committee.
  - 5.6.a Cause, for presiding officers, is defined as the following:
    - (1) Failure to activate the committee during the first semester after appointment in order to organize its business and determine an agenda; or
    - (2) Failure to activate the committee in order to respond to communications referred from the Executive Committee: or

- (3) Failure to activate the committee in order to carry out specific charges required in Article 6 or other Senate documents; or
- (4) Continual absence from scheduled committee meetings.
- 5.6.b Cause, for members, is defined as the following:
  - Continual absence from committee meetings and/or lack of participation in committee activities;
  - (2) Lack of registration on campus for students or termination of employment on campus for faculty and staff.
- 5.6.c Procedure for replacing presiding officers and members:
  - (1) The decision to replace a presiding officer rests with the Senate Chair; and
  - (2) The presiding officer of a committee shall submit the request to replace a committee member to the Chair of the Committee on Committees.
- 5.6.d The Senate Chair and the Chair of the Committee on Committees shall consult with the Office of the University Senate to identify a replacement when a decision is made to replace a presiding officer or a committee member.
- 5.7 **Standing Committees Appointing Special Subcommittees:** A standing committee of the Senate may appoint special subcommittees to assist in the effective performance of its responsibilities. Persons appointed to special subcommittees who are not members of standing committees must be approved by the Executive Committee. The Chair of any special subcommittee must be a member of the standing committee making the appointment.
- Standing Committees Appointing Special Joint Subcommittees: Two or more standing committees of the Senate may appoint special joint subcommittees to assist in the effective review of issues that pertain to the charge of multiple committees. Persons appointed to serve who are not members of associated standing committees must be approved by the Executive Committee. The Chair of any such subcommittee must be a member of one of the associated standing committees making the appointment. Special Joint Subcommittees will report directly to the full associated standing committees for final action.
- 5.9 **Special Committees:** A special committee of the Senate may be established by resolution of the Senate to carry out a specified task. The empowering resolution shall also stipulate the means of selecting the committee and any restrictions on its composition. The committee shall function until the completion of its tasks or until discharged by the Senate. A final report of its work shall be presented to the Senate. Members shall serve for the duration of the committee unless otherwise specified by the Senate.

# ARTICLE 6 STANDING COMMITTEE SPECIFICATIONS

#### 6.1 Academic Procedures and Standards Committee:

- 6.1.a Membership: The committee shall consist of an appointed presiding officer; ten (10) faculty members; one (1) staff member; three (3) undergraduate and one (1) graduate student; and the following persons or a representative of each: the Senior Vice President and Provost, the Director of Undergraduate Admissions, the University Registrar, the Associate Provost for Academic Affairs and Dean for Undergraduate Studies, and the Associate Provost for Academic Affairs and Dean of the Graduate School.
- 6.1.b Quorum: A quorum of the Academic Procedures and Standards Committee shall be ten (10) voting members.

- 6.1.c Charge: The committee shall formulate and continually review policies, rules, and regulations governing the admission, readmission, academic standing, and dismissal of all students for academic deficiency.
- 6.1.d Charge: The committee shall formulate and continually review policies and procedures for academic advisement, scheduling of classes, and registration.
- 6.1.e Charge: The committee shall formulate and continually review policies to be observed by the instructional staff in conducting classes, seminars, examinations, students' research, and student evaluations.
- 6.1.f Policies, rules, and regulations exclusively governing admission, readmission, scholastic standing, and dismissal of graduate students for academic deficiency shall be reviewed by an appropriate committee of the Graduate School. Such policies, rules, and regulations will be transmitted by the Graduate School directly to the Senate through the Executive Committee. Policies, rules, and regulations that concern both graduate and undergraduate matters shall be considered by both the Academic Procedures & Standards (APAS) Committee and the appropriate committee of the Graduate School.

### 6.2 Campus Affairs Committee:

### 6.2.a Membership:

- (1) The committee shall consist of an appointed presiding officer; six (6) faculty members; two (2) undergraduate and two (2) graduate students; two (2) staff members, with one exempt and one non-exempt to the extent of availability; the President or a representative of the Student Government Association; the President or a representative of the Graduate Student Government; and the following persons or a representative of each: the Senior Vice President and Provost, the Vice President and Chief Administrative Officer, the Vice President for Student Affairs, the Vice President for Marketing and Communications, the Vice President for Diversity & Inclusion, and the Chair of the Coaches Council.
- (2) When discussions of safety are on the agenda, the Chief of Police, the Office of General Counsel, the Director of Transportation Services, and other campus constituencies, as appropriate, shall be invited to participate or send a representative.
- (3) The Chair of this committee or a faculty member designated by the Chair and approved by the Senate Executive Committee will serve as an ex officio member of the Athletic Council. The Chair, or a committee member designated by the Chair, shall also serve as an ex-officio member of the Campus Transportation Advisory Committee.
- 6.2.b Quorum: A quorum of the Campus Affairs Committee shall be nine (9) voting members.
- 6.2.c Charge: The committee shall formulate and continually review policies and regulations affecting the entire campus, its functions, its facilities, its internal operation and its external relationships, including the awarding of campus prizes and honors, and make recommendations concerning the future of the campus.
- 6.2.d Charge: The committee shall formulate and continually review policies and procedures for the periodic review of campus level administrators.
- 6.2.e Charge: The committee shall periodically gather community input on safety and security issues and shall act as a liaison between the police and the campus community.

### 6.3 Committee on Committees:

- 6.3.a Membership and terms:
  - (1) As set forth in the *Plan* (Article 8.3.a), the Committee on Committees shall be chaired by the Chair-Elect of the Senate.

- (2) The voting membership, as defined in the *Plan* (Article 8.3.a), shall consist of the Chair-Elect of the Senate, six (6) faculty members elected by faculty Senators, with no more than one (1) from any College or School; one (1) non-exempt staff member elected by non-exempt staff Senators; one (1) exempt staff member elected by exempt staff Senators; one (1) undergraduate student elected by undergraduate student Senators; and one (1) graduate student elected by graduate student Senators.
- (3) Students are elected to serve for one (1) year, faculty and staff for two (2) years, whether or not their membership in the Senate continues beyond their first year of service in the committee.
- (4) Terms of faculty and staff members are staggered in such a way that, at any time, no more than three (3) faculty members and one (1) staff member are serving the second year of their term.
- (5) In the event of a vacancy on the Committee on Committees, the available candidate who had received the next highest number of votes in the last annual election for the Committee on Committees shall fill the remainder of the unexpired term. In the event that there is no runner-up, the Executive Committee shall fill the vacant seat.
- (6) A quorum of the Committee on Committees shall be six (6) voting members.

### 6.3.b Charge:

- (1) As set forth in the *Plan* (Article 8.3.b), responsibilities of the Committee on Committees include:
  - (a) Identification and recruitment of individuals for service on Senate committees;
  - (b) Approval of the University Library Council slate of nominees, as mandated in section 2.C of the Bylaws of the University Library Council.
  - (c) Creation of a slate of nominees for the Nominations Committee, for approval by the Senate.
- (2) Additional duties include:
  - (a) As needed, the Committee on Committees may be charged to assess effectiveness of committees, and make recommendations for improvements and changes in their operations and structure.
  - (b) Other such duties as specified by the Executive Committee.
- 6.3.c Operation: The Committee on Committees shall follow the procedures specified for standing committees in Article 5 above, with the exceptions of 5.3.b and 5.5.

### 6.4 Educational Affairs Committee:

- 6.4.a Membership: The committee shall consist of an appointed presiding officer; ten (10) faculty members, of whom at least two (2) must be tenured/tenure-track faculty members and at least two (2) must be professional track faculty members; two (2) staff members, with one exempt and one non-exempt to the extent of availability; two (2) undergraduate students and one (1) graduate student; the President or a representative of the Student Government Association; the President or a representative of the Graduate Student Government; the Associate Dean for General Education; a representative of the Associate Provost for Academic Affairs and Dean for Undergraduate Studies; and the following persons or a representative of each: the Senior Vice President and Provost, the Associate Provost for Academic Affairs and Dean of the Graduate School, and the Vice President of Information Technology and Chief Information Officer (CIO).
- 6.4.b Quorum: A quorum of the Educational Affairs Committee shall be eleven (11) voting members.
- 6.4.c Charge: The committee shall formulate and continually review plans and policies to strengthen the

educational system of the College Park campus. The committee shall receive ideas, recommendations, and plans for educational innovations from members of the campus community and others. The committee shall inform itself of conditions in the Colleges, Schools, and other academic units, and shall propose measures to make effective use of the resources of the campus for educational purposes.

- 6.4.d Charge: The committee shall exercise broad oversight and supervision of the General Education Program at the University of Maryland as described in the 2010 document Transforming General Education at the University of Maryland and the General Education Implementation Plan approved by the University Senate in February 2011. The committee shall review and make recommendations concerning the General Education Program to the Senate and the Associate Provost for Academic Affairs and Dean for Undergraduate Studies. Such recommendations shall include, as the committee deems appropriate, the program's requirements and its vision, especially with regard to evaluating trends, reviewing learning outcomes, and maintaining the balance of courses in the General Education categories.
- 6.4.e Relation of the Educational Affairs Committee to the General Education Program and the Office of the Associate Provost for Academic Affairs and Dean for Undergraduate Studies:
  - (1) The Associate Provost for Academic Affairs and Dean for Undergraduate Studies willprepare an annual report on the status of the General Education Program and will send the report to the Educational Affairs Committee by October 1.
  - (2) The Associate Provost for Academic Affairs and Dean for Undergraduate Studies will meet with the Educational Affairs Committee as needed to discuss or update the report. Topics will include but not be limited to: the membership and ongoing work of the General Education Faculty Boards; the proposal and approval process for General Education courses; the learning outcomes for the different course categories; areas where additional courses or rebalancing may be needed; trends and developments that may impact the General Education Program; and informational resources for students, faculty, and advisors about the General Education Program.
  - (3) The Office of the Associate Provost for Academic Affairs and Dean for Undergraduate Studies shall inform the committee of modifications in the proposal or review process, the disposition of recommendations from the committee, and any other changes regarding the implementation of the General Education Program as specifically delegated to that office.

### 6.5 Elections, Representation, and Governance Committee:

- 6.5.a Membership: The committee shall consist of an appointed presiding officer; six (6) faculty members; one (1) exempt staff member; one (1) non-exempt staff member; two (2) undergraduate and two (2) graduate students; and representatives of the Director of Human Resources and the Associate Vice President for Institutional Research, Planning, and Assessment.
- 6.5.b Quorum: A quorum of the Elections, Representation, and Governance Committee shall be eight (8) voting members.
- 6.5.c Charge: The committee shall review and recommend policies regarding the conduct of elections, determine correct apportionments for all constituencies, and investigate and adjudicate all charges arising from the management and results of Senate elections.
- 6.5.d Charge: The committee shall determine the correct apportionment for all constituencies every five (5) years as stipulated in Article 3.8 of the Plan and following any review or revision of the *Plan* as stipulated in Article 6.3 of the *Plan*.
- 6.5.e Charge: The committee shall supervise all Senatorial elections and referenda in accordance with the *Plan* (Article 4.2), and shall consult with certain constituencies in their nomination and election processes in accordance with the *Plan* (Article 4) as requested by the Executive Committee.

- 6.5.f Charge: The committee shall formulate and review procedures for the tallying and reporting of election results and shall perform other such duties as appropriate (Article 3.3.b of the *Plan*).
- 6.5.g Charge: The committee shall review the Plans of Organization of the Colleges, Schools, and other units, in accordance with the *Plan* (Article 11) and as specified in Appendix 7 of these Bylaws.
- 6.5.h Charge: The committee shall review and observe the operation and effectiveness of the University Senate and make any appropriate recommendations for improvements.
- 6.5.i Charge: The committee shall receive all petitions for impeachment of the Chair or Chair-Elect in accordance with the *Plan* (Article 5.8).
- 6.5.j Charge: The committee shall initiate procedures for expelling Senators in accordance with the *Plan* (Article 4.10).
- 6.5.k Charge: The committee shall receive all petitions for the recall of Senators in accordance with the *Plan* (Article 4.11).

### 6.6 Equity, Diversity, & Inclusion Committee:

- 6.6.a Membership: The committee shall consist of an appointed presiding officer; five (5) faculty members; three (3) exempt staff members; two (2) non-exempt staff members; two (2) undergraduate and two (2) graduate students; and the following persons or a representative of each: the Senior Vice President and Provost, the Vice President for Diversity & Inclusion, the Vice President and Chief Administrative Officer, the Vice President for Student Affairs, and the Director of the Office of Civil Rights and Sexual Misconduct.
- 6.6.b Quorum: A quorum of the Equity, Diversity, & Inclusion Committee shall be ten (10) voting members.
- 6.6.c Charge: The committee shall actively promote an equitable, diverse, and inclusive campus that is free from all forms of discrimination by formulating and continually reviewing policies and procedures pertaining to issues of equity, diversity, and inclusion. These include but are not limited to the University of Maryland Non-Discrimination Policy and Procedures and the University of Maryland Disability & Accessibility Policy and Procedures.
- 6.6.d Charge: The committee shall consider programs and activities for improving equity, diversity, and inclusiveness on campus, and shall make recommendations to appropriate campus bodies.

### 6.7 Faculty Affairs Committee:

- 6.7.a Membership: The committee shall consist of an appointed presiding officer; ten (10) faculty members, of whom four (4) shall be Senators including one (1) assistant professor and one (1) professional track faculty member; one (1) undergraduate student and two (2) graduate students; one (1) staff member; and the following persons or a representative of each: the President, the Senior Vice President and Provost, and the Director of Human Resources. One (1) elected Council of University System Faculty representative from the University shall serve as a voting ex officio member. The Faculty Ombuds Officer shall serve as a non-voting ex officio member.
- 6.7.b Quorum: A quorum of the Faculty Affairs Committee shall be nine (9) voting members.
- 6.7.c Charge: The committee shall formulate and continually review policies pertaining to faculty life, employment, academic freedom, morale, and perquisites.
- 6.7.d Charge: The committee shall work for the advancement of academic freedom and the protection of faculty and research interests.
- 6.7.e Charge: The committee shall, in consultation with Colleges, Schools, and other academic units,

formulate and review procedures for the periodic review of academic administrators below the campus level.

6.7.f Charge: The committee shall review the Appointment, Promotion, and Tenure or Permanent Status section of each College, School, or the Library Plan of Organization in accordance with Appendix 7 of these *Bylaws*. In conjunction with this review, the committee shall also review the professional track faculty Appointment, Evaluation, and Promotion Policy of each College, School, or the Library.

### 6.8 Programs, Curricula, and Courses Committee:

- 6.8.a Membership: The committee shall consist of an appointed presiding officer; ten (10) faculty members; one (1) staff member; two (2) undergraduate students and one (1) graduate student; and the following persons or a representative of each: the Senior Vice President and Provost, the Associate Provost for Academic Affairs and Dean for Undergraduate Studies, the Associate Provost for Academic Affairs and Dean of the Graduate School, and the Dean of Libraries.
- 6.8.b Quorum: A quorum of the Programs, Curricula, and Courses Committee shall be nine (9) voting members.
- 6.8.c Charge: The committee shall formulate, review, and make recommendations to the Senate concerning policies related both (1) to the establishment, modification, or discontinuance of academic programs, curricula, and courses; and (2) to the establishment, reorganization, or abolition of colleges, schools, academic departments, or other units that offer credit-bearing programs of instruction or regularly offer courses for credit.
- 6.8.d Charge: The committee shall review and make recommendations to the Senate in at least the areas designated by (1) through (3) below. Recommendations in these areas are not subject to amendment on the Senate floor unless a detailed objection describing the area of concern has been filed with the Office of the University Senate at least forty-eight (48) hours prior to the meeting at which the recommendations will be introduced. The committee will announce proposed recommendations to the campus community sufficiently in advance of the meeting at which they are to be considered so as to allow time for concerned parties to file their objections.
  - (1) All proposals for the establishment of a new academic program, for the discontinuance of an existing academic program, for the merger or splitting of existing academic programs, or for the renaming of an existing academic program;
  - (2) All proposals for the creation, abolition, merger, splitting, or change of name of Colleges, Schools, departments of instruction, or other units that offer credit-bearing programs of instruction or regularly offer courses for credit; and
  - (3) All proposals to reassign existing units or programs to other units or programs.
- 6.8.e Charge: The committee shall review and shall directly advise the Office of Academic Planning and Programs concerning proposals to modify the curricula of existing academic programs, or to establish citation programs consistent with College rules approved by the Senate. The committee shall inform the Senate of its actions in these cases.
- 6.8.f Charge: The committee shall review, establish, and advise the Vice President's Advisory Committee concerning policies for adding, deleting, or modifying academic courses.
- 6.8.g Charge: The committee shall be especially concerned with the thoroughness and soundness of all proposals, and shall evaluate each according to the mission of the University, the justification for the proposed action, the availability of resources, the appropriateness of the sponsoring group, and the proposal's conformity with existing regulations. The committee shall be informed of any recommendations made by the Academic Planning Advisory Committee concerning resource issues, the consistency of the proposed action with the University's mission and strategic directions, or both.

- 6.8.h Operation: The committee shall follow the procedures specified for standing committees in Article 5 above, with the exception of 5.3.b.
- 6.8.i Relation of the Programs, Curricula, and Courses Committee to the Office of the Senior VicePresident and Provost.
  - (1) The committee, in consultation with the Office of the Senior Vice President and Provost, shall determine the requirements for supporting documentation and the procedures for review for all proposals.
  - (2) The committee shall be informed by the Office of the Senior Vice President and Provost of all proposed modifications to existing programs and curricula. After consulting with the presiding officer of the committee, the Office of the Senior Vice President and Provost shall act on all minor changes that are not of a policy nature.
  - (3) The committee shall be informed by the Office of the Senior Vice President and Provost of all changes made pursuant to 6.8.i(2). The committee shall be informed by the Office of the Senior Vice President and Provost of all other changes in academic curricula whose approval has been specifically delegated to that office. In particular, this includes the approval to offer existing academic programs through distance education or at a new off-campus location.
- 6.8.j Relationship of the Programs, Curricula, and Courses Committee to the Graduate School: Proposals concerned with graduate programs and curricula shall receive the review specified by the Graduate School, in addition to the review of the Programs, Curricula, and Courses Committee. Any such proposal whose approval has been denied by the Graduate School shall not be considered by the committee.

### 6.9 Staff Affairs Committee:

- 6.9.a Membership: The committee shall consist of an appointed presiding officer; eight (8) staff members, with two (2) members from each of the elected staff categories; two (2) Category II contingent employees, with one exempt and one non-exempt to the extent of availability; one (1) faculty member; one (1) student; and one (1) representative each of the Senior Vice President and Provost, the Director of Human Resources, the Vice President and Chief Administrative Officer and the Vice President for Student Affairs. The three (3) elected University representatives to the Council of University System Staff (CUSS) shall serve as voting ex officio members; the alternate University representatives to the Council of University System Staff (CUSS) shall be non-voting ex officio members.
- 6.9.b Quorum: A quorum of the Staff Affairs Committee shall be nine (9) voting members.
- 6.9.c Charge: The committee shall formulate and continually review campus policies affecting staff members, including policies regarding periodic review of campus departments and administrators that employ staff members.
- 6.9.d Charge: The committee shall assist the Office of the University Senate in soliciting nominations and encouraging participation in elections of staff Senators as specified in Article 4.5 of the *Plan*.
- 6.9.e Charge: Staff Affairs shall assist the Committee on Committees and the Senate Executive Committee in identifying and recruiting staff representatives for campus and Senate committees, including system-wide activities involving staff.
- 6.9.f Charge: The committee shall administer the Council of University System Staff (CUSS) nomination and election process. Definitions of eligible staff shall be defined by the Board of Regents and CUSS.
- 6.9.g Charge: The committee shall actively promote and provide orientation and opportunities for staff involvement in shared governance at every administrative level.
- 6.9.h Charge: The committee shall facilitate the annual nomination process for the Board of Regents' Staff

#### 6.10 Student Affairs Committee:

- 6.10.a Membership: The committee shall consist of an appointed presiding officer; eight (8) undergraduate students, of whom four (4) must be Senators; four (4) graduate students, of whom two (2) must be Senators; two (2) faculty members; two (2) staff members with one exempt and one non-exempt to the extent of availability; the President or a representative of the Student Government Association; the President or a representative of the Graduate Student Government; two (2) representatives of the Office of the Vice President for Student Affairs; and one (1) representative each from the Graduate School, and the Department of Resident Life.
- 6.10.b Quorum: A quorum of the Student Affairs Committee shall be ten (10) voting members.
- 6.10.c Charge: The committee shall formulate and continually review policies regarding all non-academic matters of student life including, but not limited to, student organizations, resident life, extracurricular activities, and student concerns in the campus community.
- 6.10.d Charge: The committee shall support the work of other Senate committees by assessing and communicating the student perspective on a range of issues affecting students, including matters outside the purview described in 6.10.c.
- 6.10.e Charge: The committee shall assist the Office of the University Senate and the Colleges and Schools as appropriate in soliciting nominations and encouraging participation in the election of student Senators.

### 6.11 Student Conduct Committee:

- 6.11.a Membership: The committee shall consist of an appointed presiding officer; four (4) faculty members; one (1) staff member; five (5) students, of whom at least three (3) must be undergraduate students and one (1) must be a graduate student; and the Director of the Office of Student Conduct, or a representative, as a non-voting ex officio member.
- 6.11.b Charge: The committee shall formulate and continually review recommendations concerning the rules and codes of student conduct, as well as means of enforcing those rules and codes.
- 6.11.c Charge: The committee acts as an appellate body for infractions of the approved Code of Student Conduct and Code of Academic Integrity. Procedures for the committee's operation in this role are to be developed and filed with the Office of Student Conduct and the Executive Secretary and Director of the Senate. The committee shall also confirm members of all judicial boards listed in the Code of Student Conduct, except conference and ad hoc boards.

# ARTICLE 7 SPECIAL COMMITTEE ON UNIVERSITY FINANCE

### 7.1 Membership and Selection:

- 7.1.a Composition: The special committee shall consist of a presiding officer appointed by the Senate Chair from among the tenured faculty; five (5) tenured or tenure-track faculty members; one (1) professional track faculty member; one (1) exempt staff member; one (1) non-exempt staff member; two (2) undergraduate students; one (1) graduate student; the immediate Past Chair of the Senate; the Vice President for Finance and Chief Financial Officer; the Associate Vice President for Finance and Personnel, Academic Affairs; and the following persons or a representative of each: the President, and the Vice President for Student Affairs. The Senior Vice President and Provost shall also appoint a representative chosen from among current and former unit-level budget officers or former department chairs. All members of the special committee shall be voting members.
- 7.1.b Selection of Members: The regular membership of the special committee shall be selected by the elected members of the Senate Executive Committee. Following the May 7, 2019, Transition Meeting,

current Senators may nominate any member of the campus community. Nominees shall provide a statement indicating their interest in and qualifications for the special committee. Members of the Senate Executive Committee may not be nominated. Elected members of the Senate Executive Committee will vote by constituencies for members of the special committee. In the event of a tie, the Senate Chair will cast the deciding vote.

- 7.1.c Membership—Vacancies: After each Transition Meeting of the Senate, current Senators may nominate members of the campus community for any vacant seats. In the event of a vacancy during the academic year, members of the Senate Executive Committee from the respective constituency will select a replacement from the most recent list of nominees. If there are no interested nominees, a new nomination period will be opened and members of the Senate may submit nominations following the procedures in 7.1.b.
- 7.1.d Membership—Terms: Terms shall be four (4) years for faculty and staff, and one (1) year for students. Student members who wish to continue may be renewed up to two times. Terms shall begin on July 1, 2019.
- 7.2 Charge: The special committee shall exercise the following functions:
  - 7.2.a Develop a deep understanding of the University's budget and budgeting processes and use that knowledge to educate the campus community on these practices.
  - 7.2.b Consult with and advise the President, the Senior Vice President and Provost, and other University administrators on short- and long-term institutional priorities, particularly as they relate to the University's mission and Strategic Plan.
  - 7.2.c Advise Senate-related bodies—including committees, councils, and task forces—on the fiscal implications of any proposed recommendations under consideration.
  - 7.2.d Report to the Senate two times each year on the budgetary and fiscal condition of the University and the administration's response to any special committee recommendations.
  - 7.2.e Regularly report on its activities and the budgetary and fiscal condition of the University to the Senate Executive Committee.

### 7.3 Operations:

- 7.3.a Agenda Determination: The special committee shall have principal responsibility for identifying matters of present and potential concern to the campus community within its area of responsibility. The presiding officer shall place such matters on the agenda of the committee. Agendas shall be made publicly available prior to each meeting.
- 7.3.b Meetings: The special committee shall meet as frequently as is needed to accomplish its charge, but at least monthly throughout the academic year. Additional meetings may be required over the summer months to accommodate the University's budgeting processes. Given the sensitive nature of the special committee's work, meetings will be closed to all but members and invited guests.
- 7.3.c Minutes: Minutes of the special committee's proceedings shall be kept.
- 7.3.d Procedure: The version of *Robert's Rules of Order* that shall govern the special committee shall be *Robert's Rules of Order for Small Committees, Newly Revised.* The special committee shall determine how technology, such as phone and video conferencing and other electronic methods of participation, can be used for its purposes. The special committee may choose to conduct votes via email, and shall agree on any other mechanisms for conducting business outside of meetings, when necessary.
- 7.3.e Quorum: Quorum shall be a majority of the members of the special committee.
- 7.3.f Guests: The special committee may invite guests to participate in its meetings if it is deemed necessary.

#### 7.4 Dissolution:

7.4.a The special committee shall be dissolved following the adjournment of the last regular Senate meeting of the 2022-2023 academic year, at which time the provisions in this article will become inoperative.

## ARTICLE 8 UNIVERSITY COUNCILS

- 8.1 **Definition:** University Councils are established by Article 8.6 of the *Plan* to exercise an integrated advisory role over specified campus units and their associated activities. University Councils are jointly sponsored by the University Senate and the Office of the President or Provost (as appropriate). University Councils may be assigned reporting responsibilities to any member(s) of the College Park administration at the dean level or above (hereafter referred to as the "designated administrative officer").
- 8.2 Creation of University Councils: Proposals to create a University Council shall be evaluated by a task force appointed jointly by the Senate Executive Committee and the designated administrative officer to whom the new Council would report. Following its deliberations, this task force shall present a report (hereafter referred to as the "Task Force Report") to the Senate, the designated administrative officer, and the director of the unit whose activities are the focus of the Council. The Task Force Report shall indicate the specifications that define the working relationship among the Senate, the designated administrative officer, and the director. The Task Force Report shall include at least the following: the scope and purpose of the new Council; a review of the current committees and advisory relationships to be superseded by the proposed Council; identification of the designated administrative officer and unit director to whom the Council reports; the charge to the Council; the size, composition, and appointment process of members of the Council; the Council's relationship to the Senate, the designated administrative officer, and the director including the responsibilities of these three sponsors to the Council and the responsibilities of the Council to these three sponsors; and principles for operation of the Council. The Task Force Report shall be reviewed by the Executive Committee, approved by the designated administrative officer, and then approved by the Senate. At the same time, the Senate shall approve appropriate revisions in its Bylaws to incorporate the Council into its council structure as defined in Article 8 of these Bylaws. The Task Force Report, as approved, shall be preserved with official Senate documents, serving as a record of the original agreements establishing the Council.
- 8.3 **Specifications in Senate Bylaws:** For each Council, Senate Bylaws shall: state its name; specify its responsibilities to the Senate; define its membership, including any voting privileges of ex officio members; and identify any exceptions or additions to the provisions of this Article particular to the Council.

### 8.4 Basic Charge:

- 8.4.a The Council's responsibilities to the University Senate shall include those specified for Senate committees in Article 5.2 of these *Bylaws*. In addition, each Council shall:
  - (1) Sponsor hearings, as appropriate, on issues within its purview that are of concern to the Senate and the campus community.
  - (2) Provide a mechanism for communication with the campus community on major issues facing the unit and its activities.
  - (3) Respond to charges sent to the Council by the Senate Executive Committee in accordance with Article 4.7.
  - (4) Provide an annual written report to the Senate on the Council's activities including the status of unresolved issues.
- 8.4.b Responsibilities to the designated administrative officer shall be specified in the Task Force Report and may include:
  - (1) To advise on the unit's budget, space, and other material resources, in addition to personnel,

- staffing and other human resources.
- (2) To advise on the unit's administrative policies and practices.
- (3) To advise on the charges to be given to periodic internal and external review committees.
- (4) To respond to requests for review, analysis, and advice from the designated administrative officer.
- (5) To meet at least annually with the designated administrative officer to review the major issues facing the unit and its activities on campus.
- (6) To fulfill such other responsibilities as specified in the Task Force Report.
- 8.4.c Responsibilities to the unit's director shall be specified in the Task Force Report and may include:
  - (1) To advise on the needs and concerns of the campus community.
    - (2) To advise on opportunities, policies, and practices related to the unit's ongoing operations.
  - (3) To review and advise on unit reports, studies, and proposed initiatives.
  - (4) To respond to requests for review, analysis, and advice made by the director.
  - (5) To meet at least annually with the director to review the major issues facing the unit and its activities on campus.
  - (6) To fulfill such other responsibilities as specified in the Task Force Report.

### 8.5 Membership and Appointment to University Councils:

- 8.5.a Membership: Councils shall have nine (9) to thirteen (13) members as specified in the appropriate subsection of Article 9 of these *Bylaws*. In addition, each Council shall include an ex officio member designated by the administrative officer, and such other ex officio members as specified in Article 5.5.d of these *Bylaws*. These ex officio members shall have voice but no vote.
- 8.5.b Appointment: Representatives of the designated administrative officer's office and the University Senate shall agree on nominees for vacancies on the Council. These nominations shall be submitted to the designated administrative officer for approval. In addition, these nominations shall be submitted to the University Senate for approval, or for election if specified in the Council's governing documents. In exercising its powers of appointment to the Council, the Senate shall follow procedures for review and approval for Senate committee appointments specified in Article 5.5.e of these *Bylaws*.
- 8.5.c Terms: Rules governing beginning date and length of terms, and restrictions on reappointment shall be specified in the governing documents of each Council. The presiding officer shall serve a three (3) year term and cannot be reappointed, unless otherwise specified in the governing documents of the Council.
- 8.5.d Appointment of Presiding Officer: The designated administrative officer and the Senate Chair shall reach an agreement on a presiding officer, and the joint choice shall be submitted to the Senate for approval. If the presiding officer is selected from among the membership of the Council, a replacement shall be appointed to the vacated seat.

# 8.6 Operational Relationship of University Councils to Sponsors:

- 8.6.a The Office of the University Senate shall provide basic support for the activities of University Councils.
- 8.6.b The office of the designated administrative officer, through its ex officio University Council member,

- shall provide administrative support and serve as a liaison to other administrative units as required.
- 8.6.c The unit director shall provide the University Council with internal data, reports, studies, and any other materials required to support the Council's work. In addition, the director shall also arrange for unit staff to appear before the committee as requested.
- 8.6.d Control of the University Council's agenda shall be the responsibility of the presiding officer of the University Council and the voting members of the University Council in accordance with procedures for standing committees provided in Article 5.3.a, subject to the charges provided in Article 8.4 of these *Bylaws*, the appropriate subsection of Article 9 of these *Bylaws*, and the approved Task Force Report governing the University Council.
- 8.6.e Each University Council shall develop its own bylaws, which must be approved by the designated administrative officer and by the Senate.
- 8.6.f In addition to the required annual report, the presiding officer shall keep the Executive Secretary and Director and the Chair of the Senate informed of the major issues before the University Council and shall indicate when action or information items are likely to be forwarded for Senate consideration. In submitting recommendations for Senate action, the University Council shall inform the unit director and the designated administrative officer in advance of its recommendations. For purposes of conducting Senate business, reports from the University Council and floor privileges of the Senate shall be managed in the same manner as standing committees of the Senate defined in these Bylaws (3.3.c, 4.4.b). In the case where the presiding officer of the University Council is not a member of the Senate, they may report to the Senate and participate in the deliberations of the Senate subject to the provisions of Article 3.3.c of these Bylaws.

### 8.7 Review of University Councils:

- 8.7.a Five (5) years after a University Council is formed, a review of the University Council shall be undertaken jointly by the Senate and administration, and a written report issued. The review may recommend continuation of the University Council in its original form and mode of operation, modification of the University Council structure and/or operations, or discontinuance of the University Council.
- 8.7.b Following the initial review, the University Council and its operations shall be reviewed in conjunction with the periodic review of the *Plan*.

# ARTICLE 9 UNIVERSITY COUNCIL SPECIFICATIONS

### 9.1 University Library Council

- 9.1.a Charge: The University Library Council has the responsibility to provide advice and to report on policy issues concerning the University Libraries to the University Senate, to the Senior Vice President and Provost, and to the Dean of Libraries (see Appendix 1 for additional responsibilities and the Library Council's Bylaws).
- 9.1.b Membership: The Library Council shall consist of thirteen (13) appointed members and four (4) ex officio members. The appointed members shall be: the Chair, ten (10) faculty members including at least one (1) member of the library faculty, one (1) graduate student, and one (1) undergraduate student. The four (4) ex officio members shall be a representative of the Office of the Senior Vice President and Provost, a representative of the Office of the Dean of Libraries, a representative of the Division of Information Technology, and the Chair-Elect of the Senate.
- 9.1.c The Chair shall be a tenured faculty member.
- 9.1.d Reporting Responsibilities: The University Library Council shall report to the University Senate and the Senior Vice President and Provost under the terms of responsibility defined in Article 8.4 of these

Bylaws.

## 9.2 University Research Council:

- 9.2.a Charge: In addition to the charges specified in Articles 5.2 and 8.4 of these *Bylaws*, the Research Council shall be governed by the following: The Research Council is charged to formulate and continually review policies regarding research, its funding, its relation to graduate and undergraduate academic degree programs, and its service to the community. Also, the Research Council is charged to review the research needs of faculty, other researchers and students, and to make recommendations to facilitate the research process and productivity of the University. Further, the Research Council shall formulate and continually review policies on the establishment, naming, reorganization, or abolition of bureaus, centers, or institutes that do not offer programs of instruction or regularly offer courses for credit, including their relationship to graduate and undergraduate academic programs. Additionally, when it perceives problems, the Research Council has the power to undertake investigative studies and recommend solutions.
- 9.2.b Membership: The University Research Council shall consist of thirteen (13) appointed members and ten (10) ex officio members. The appointed members shall be the Chair and eight (8) faculty members; one (1) staff member; and three (3) students, including at least one (1) graduate and one (1) undergraduate student. Eight (8) voting ex officio members include a representative of the Vice President for Research, a representative of the Dean of the Graduate School, a representative of the Dean of Undergraduate Studies, the Director of the Office of Research Administration and Advancement, the Chair of the Intellectual Property (IP) Committee and, and the Chairs of the subcommittees of the University Research Council. A representative of the President and a representative of the Senior Vice President and Provost shall serve as non-voting ex-officio members.
- 9.2.c The Chair shall be a tenured faculty member.
- 9.2.d Reporting Responsibilities: The University Research Council shall report to the University Senate and the Vice President for Research under the terms of responsibility defined in Article 8.4 of these *Bylaws* and the report establishing the University Research Council.

## 9.3 University IT Council:

- 9.3.a Charge: The IT Council shall advise and report on policy issues concerning the Division of Information Technology to the University Senate and the Vice President for Information Technology and CIO. In addition to such responsibilities as are enumerated in Article 8 of these *Bylaws*, and as articulated in the Bylaws of the University IT Council (see Appendix 3), the IT Council shall:
  - 1) Respond to requests from the Division of Information Technology, extra-divisional advisory bodies (such as the Council of Deans or the Campus Student Technology Advisory Fee Committee), the University Senate, or other campus stakeholders for guidance on IT policy and implementation.
  - 2) Advise on the Division of Information Technology's budget, material resources, personnel, staffing and human resources, and administrative policies and practices.
  - Investigate matters concerning the Division of Information Technology and recommend solutions to the University Senate, the Vice President for Information Technology and CIO, or the general campus community.
  - 4) Advise on IT planning, including strategic and other major planning for information technology operation and development.
  - 5) Advise on policy recommendations related to campus technology facilities, equipment, software, and services.

- 9.3.b Membership: The IT Council shall consist of up to thirteen appointed members, and additional non-voting ex-officio members. The appointed members shall be: the chair, one staff member, one undergraduate student, one graduate student, one professional track faculty member, one tenured faculty member, and the chairs of the IT Council Working Groups. The non-voting ex-officio members shall include a representative from the University Libraries; a representative from the Office of the Provost; a representative from the Information Technology Advisory Committee; and the Vice President for Information Technology and CIO. Additional non-voting ex-officio members may be appointed as needed, by agreement between the Vice President for Information Technology and CIO and the Senate Executive Committee.
- 9.3.c The chair of the IT Council shall be appointed by the Vice President for Information Technology and CIO and the Senate, as described in 8.5 of these *Bylaws*. The chair will serve a three-year term.
- 9.3.d Working Groups: The IT Council may create up to seven Working Groups. These groups should carry out research and make recommendations on IT issues, and work with the appropriate Division of Information Technology staff member appointed by the Vice President for Information Technology and CIO. The specific responsibilities of each Working Group shall be described in the Bylaws of the University IT Council. The chair of each Working Group shall be appointed by the Vice President for Information Technology and CIO and the Senate and shall serve a two-year term.
- 9.3.e Reporting Responsibilities: The IT Council shall report to the Vice President for Information Technology and CIO and to the University Senate under the terms of responsibility defined in Article 8.4 of these *Bylaws*.

# ARTICLE 10 THE ATHLETIC COUNCIL

#### 10.1 The Athletic Council

- 10.1.a The Athletic Council exists to help the University develop and maintain the best possible intercollegiate athletic program consistent with the academic integrity of the institution and the academic and social development of student athletes. The Athletic Council shall operate in accordance with its charter (Appendix 4), which shall specify its role, scope, responsibilities, leadership, and membership. Changes to the charter shall be approved by the President of the University.
- 10.1.b Membership: The charter designates its membership. The membership of the Athletic Council elected by the Senate includes:
  - 1) Seven faculty members elected by the Senate at the annual Transition Meeting. Elected faculty representatives shall serve for a three-year term, and faculty who have served a full term shall for a period of one year be ineligible for re-election. The Senate should make very effort to assure diversity among the candidates for election to the Council.
  - 2) One staff member elected by the Senate at the annual Transition Meeting for a three-year term. A staff member who has served a full term shall for a period of one year be ineligible for re-election.
  - 3) The Chair of the Senate Campus Affairs Committee, or a faculty member designated by the Committee, shall serve as an ex-officio member.
- 10.1.c Relationship between the Senate and the Athletic Council:
  - The Council in cooperation with the Athletic Director shall submit an annual report to the Senate on the status of intercollegiate athletics at the University. This report shall at least include an analysis of admissions, academic performance, class attendance, major selection, graduation rates, budget performance, and compliance with NCAA, Conference, and campus rules.

2) The Council shall inform the Senate for its review of any proposed amendments to the Council's charter.

# ARTICLE 11 DUTIES OF THE EXECUTIVE SECRETARY AND DIRECTOR

- 11.1 The Executive Secretary and Director of the Senate shall be responsible for the minutes and audio recordings of all Senate meetings.
  - 11.1.a The minutes shall include actions and business transacted, at a minimum. They shall be submitted to the Senate for approval. Copies of the approved minutes shall be made available to all chief administrative officers of Colleges, Schools, departments, and other units, and to the campus news media.
  - 11.1.b A complete audio recording shall be made of each meeting and shall be maintained by the Office of the University Senate. In accordance with the University's Records Retention and Disposal Schedule, a copy of each audio recording, excluding only those parts recorded during closed sessions, shall be placed with the minutes in the University Archives for open access.
- 11.2 The Executive Secretary and Director shall also maintain the following kinds of Senate records (see Article4.8):
  - (1) All material distributed to Senate members;
  - (2) All material received by or distributed to members of the Executive Committee;
  - (3) Any minutes of the Senate or the Executive Committee not otherwise included under (1) and (2);
  - (4) Annual reports of all committees of the Senate not otherwise included under (1) and (2);
  - (5) The audio records of Senate meetings;
  - (6) The current and all previous versions of the *Plan* and the *Bylaws*:
  - (7) Articles concerned with Senate structure and operation from campus and University publications as they come to the attention of the Executive Secretary and Director; and
  - (8) Other items deemed appropriate by the Executive Secretary and Director or the Chair of the Senate.
- 11.3 The Executive Secretary and Director shall store inactive records of the Senate in the University Archives.
- 11.4 The Executive Secretary and Director shall be responsible for the preparation of the Senate budget in accordance with Article 4.6.
- 11.5 The Executive Secretary and Director shall ensure that a directory of the membership of the new Senate indicating for each member the constituency, term, unit, and email address is made available on the Senate website following the annual transition of the Senate.
- 11.6 The Executive Secretary and Director shall ensure that contact information for all Senate officers and of all presiding officers of all Senate committees is made available on the Senate website for all members of the campus community.
- 11.7 The Executive Secretary and Director will normally provide a copy of the agenda and supporting material one week in advance of each Senate meeting.
- 11.8 The Executive Secretary and Director shall prepare for the members of the Senate and its Executive Committee, as appropriate, all agendas, minutes, reports, and other documents, with the exception of

- proposals relating to the Programs, Curricula, and Courses (PCC) Committee. Nonetheless, the Executive Secretary and Director shall be responsible for the distribution of all items of Senate business, and to other such committees as necessary.
- 11.9 The Executive Secretary and Director shall inform the Executive Committee of the status of all members of the Senate in accordance with the *Plan* (Article 3.4.a(3-4), 3.4.b(3-4), and 3.7) and these *Bylaws* (Articles 2.2, 4.1, 5.5, and 5.6).
- 11.10 The Executive Secretary and Director shall have the privilege of attending the meetings of all standing committees and ad hoc committees of the Senate to assist in the coordination of Senate business.
- 11.11 The Executive Secretary and Director shall provide information or assistance as requested for revision of the undergraduate catalog.

# ARTICLE 12 ANNUAL TRANSITION OF THE SENATE

# 12.1 Preparation for Transition:

- 12.1.a By no later than the scheduled December meeting of the Senate, the Committee on Committees shall present to the Senate eight (8) nominees from among outgoing Senate members to serve on the Nominations Committee. The nominees shall include four (4) faculty members, one (1) exempt staff member, one (1) non-exempt staff member, one (1) graduate student, and one (1) undergraduate student. Further nominations shall not be accepted from the floor of the Senate. The Senate, as a body, shall approve the slate of nominees to serve on the Nominations Committee. The Chair-Elect of the Senate shall serve as a non-voting, ex officio member of the Nominations Committee. The Nominations Committee shall elect its own Chair from within the membership of the committee.
- 12.1.b The Nominations Committee shall solicit nominations from the membership of the Senate and shall present to the Chair of the Senate by April 15:
  - A slate of at least two (2) candidates per seat from each constituency for elected membership on the Executive Committee, including those incumbent elected members who are eligible and willing to stand for reelection,
  - (2) Slates of candidates to replace the outgoing members of the Committee on Committees, the Campus Transportation Advisory Committee (CTAC), the University Athletic Council, and the Council of University System Faculty (CUSF), and any other committees as required by these *Bylaws*, including at least one (1) nominee for each position to be filled, and
  - (3) A minimum of two (2) candidates for the office of Chair-Elect.

Before reporting to the Chair of the Senate, the Nominations Committee shall secure the consent of all candidates in writing.

12.1.c A brief statement of each candidate's qualifications shall be sent to the voting membership of the incoming Senate ten (10) working days before the Transition Meeting of the Senate. Any further nominations made by members of the Senate and accompanied by a brief supporting statement and the consent of the candidate must be received by the Executive Secretary and Director at least twelve (12) working days before the Transition Meeting. These additional nominations shall be sent to the voting membership of the incoming Senate ten (10) working days before the Transition Meeting.

### 12.2 Transition Elections:

- 12.2.a Election of the Chair-Elect shall occur as provided for in section 5.7.a of the Plan.
- 12.2.b The election of members of the Executive Committee, Committee on Committees, Campus Transportation Advisory Committee (CTAC), Athletic Council, Council of University System Faculty

(CUSF), and such other persons elected by the members of the Senate, shall be held after the election of the Chair-Elect.

- (1) Nominations for each of these committees and councils may be submitted in accordance with Article 12.1 above.
- (2) Nominations may also be received from the floor by the Chair at the Transition Meeting. Any such nomination is contingent on the consent of the candidate, which must have been secured beforehand in writing if the nomination is made in the absence of the candidate.
- (3) In the event of a tie vote in the election for members of the Executive Committee or the Committee on Committees, a ballot will be distributed to each Senator in the appropriate constituency. The election to break the tie should end one (1) week from the start date.

#### 12.3 Transition of the Senate:

- 12.3.a The new Senate session will begin at the Transition Meeting, which will be the last regularly scheduled meeting of the Spring semester.
- 12.3.b Newly elected Senators will be inducted at the Transition Meeting. Terms of office of newly elected Senators will begin, and the terms of the outgoing Senators will end, during the Transition Meeting.
- 12.3.c The outgoing Chair will pass the gavel to the previous Chair-Elect, who will assume the Chair role.
- 12.3.d The elected members of the outgoing Executive Committee and the Committee on Committees shall continue to serve until new members are elected.
- 12.3.e After the conclusion of the Transition Meeting, any vacancies on standing committees will be filled by the new Committee on Committees, subject to the approval of the Executive Committee and pending confirmation by the full Senate at its next regularly scheduled meeting.

# APPENDIX 1 BYLAWS OF THE UNIVERSITY LIBRARY COUNCIL

1. **Charge to the Library Council:** The University Library Council has the responsibility to provide advice about policy issues concerning the University Libraries to the University Senate, to the Senior Vice President and Provost, and to the Dean of Libraries.

# A. The Council's Responsibilities to the University Senate:

- (1) Make recommendations for major changes and improvements in policies, operations, and services of the Libraries that represent the concerns and interests of Senate constituencies as well as other users of the Libraries. Such recommendations should specify the resource implications. Reports and recommendations to the University Senate shall be submitted to the Senate Executive Committee for placement on the agenda of the University Senate in the same manner as reports from the Senate's standing committees. It is expected that the Library Council will also inform the Senior Vice President and Provost in advance of these legislative recommendations. In addition to the mandatory annual report, the Chair of the Library Council shall keep the Chair of the Senate informed of the major issues before the Library Council and shall indicate when action or information items are likely to be forwarded for Senate consideration.
- (2) Respond to charges sent to the Library Council by the Senate Executive Committee.
- (3) Provide an annual written report of the Library Council's activities, including the status of recommendations made by the Library Council each year, and of unresolved issues before the Library Council.

# B. The Library Council's Responsibilities to the Senior Vice President and Provost:

- (1) Advise on the Libraries' budget, space, personnel and staffing, and other resources. It is expected that the Senior Vice President and Provost will consult the Library Council before undertaking major reviews of the Libraries with APAC and before preparing the annual budget for the Libraries.
- (2) Advise on the Libraries' administrative policies and practices.
- (3) Advise on the charges to be given to the committees to review the Dean of Libraries and to conduct the unit review of the University Libraries based on University policy
- (4) Advise on matters concerning the Libraries in conjunction with accreditation review and strategic planning.
- (5) Respond to requests for review, analysis, and advice made by the Senior Vice President and Provost.
- (6) Meet at least annually with the Senior Vice President and Provost to review the major issues facing the Libraries and its activities on campus.
- (7) The Library Council is responsible for informing the Senior Vice President and Provost of pending reports and recommendations to the University Senate.

### C. The Library Council's Responsibilities to the Dean of Libraries:

- (1) Advise on the needs and concerns of diverse constituencies within the campus community with respect to Library policies, services, and new resources and technology.
- (2) Advise on strategies to involve Library users in the initiation, evaluation, and integration of new Library policies, practices, procedures, and technology. Such strategies might include forums for the discussion of changes, workshops for adjusting to new technologies, and ongoing programs of Library education.
- (3) Advise on operations, policies and new opportunities.
- (4) Advise on Library planning including strategic planning and other major plans for Library operation and development.

- (5) Review and advise on the Libraries' reports, studies, and proposed initiatives that have significant long-term resource implications for the Libraries.
- (6) Hold at least one (1) meeting each year at which the Dean shall review major issues and plans, summarized in a State of the Libraries report distributed in advance to the Library Council.
- (7) It is expected that the Library Council will adopt a broad campus perspective and that the Dean of the Libraries will inform the Library Council of the University Libraries' needs and concerns and seek advice about major modifications of policies and operations affecting the campus community.

# D. To Fulfill Its Responsibilities, the Library Council May:

- (1) Undertake investigative studies in matters concerning the University Libraries and recommend solutions to the University Senate, the Senior Vice President and Provost, the Dean of Libraries, or the general campus community.
- (2) Conduct open hearings on major issues concerning the University Libraries and their activities.
- (3) Communicate directly with the campus community on concerns related to support for, policies of, and services provided by the University Libraries.
- 2. **Composition of the Library Council:** The Library Council shall consist of thirteen (13) appointed members and four (4) ex officio members. The appointed members shall be: the Chair, ten (10) faculty members including at least one (1) member of the Library faculty, one (1) graduate student, and one (1) undergraduate student. The four (1) ex officio members shall be a representative of the Office of the Senior Vice President and Provost, a representative of the Dean of the Libraries Office, a representative of the Division of Information Technology, and the Chair-Elect of the Senate.

### A. Tenure in Office:

- (1) The Library Council Chair should be a tenured faculty member appointed for a single three-year term. Normally, the Chair shall have served as a member of the Library Council. If the Chair is serving as a regular member of the Library Council at the time of appointment, a new member shall be appointed to serve the remainder of the term the Chair has vacated. The Senior Vice President and Provost and the Senate Executive Committee shall reach an agreement on the Library Council Chair, and the joint choice shall be submitted to the University Senate for its approval.
- (2) The remaining ten (10) faculty members shall be appointed for staggered two-year terms. No faculty member shall serve more than two (2) terms consecutively. For this purpose, members who have served more than a year should be considered to have served a full term.
- (3) The two (2) student members shall be appointed for one-year terms. No student member should serve more than two (2) terms consecutively. For this purpose, student members who have served more than half their term should be considered to have served a full term.
- (4) The Office of the Senior Vice President and Provost will appoint a member of the Provost's staff as an ex officio member of the Library Council who will have voice but not vote.
- (5) The Dean of Libraries' Office will appoint an upper-level member of the Libraries' administrative staff as an ex officio member of the Library Council who will have voice but no vote.
- (6) The Vice President for Information Technology and Chief Information Officer (CIO) will appoint a member of the Division of Information Technology's staff as an ex officio member of the Library Council who will have voice but no vote.
- (7) The Chair-Elect of the Senate shall serve as an ex officio member of the Library Council who will have voice but no vote.

- B. Qualifications of Library Council Members: Successful operation of the Library Council requires that the members of the Library Council understand the nature of the Libraries and represent the best interests of the campus as well as the particular interests of their specific constituencies.
  - 1. The Library Council members should be chosen from people who can bring a campus-wide perspective to their deliberations on Library matters and who have shown interest and willingness to foster a good working relationship between the Libraries and their users.
  - 2. Library Council members should be selected to represent as broad a range of campus disciplines and interests as possible. Faculty members should include representatives from both the professional and arts and sciences colleges, and within these constituencies, representatives of the arts and humanities, social sciences, and physical and biological sciences.
- C. The Appointment Process: In the spring of each year, the Chair of the University Library Council shall notify the representative of the Office of the Senior Vice President and Provost and the Chair-Elect of the Senate of the appointments required for the following academic year. The representative of the Office of the Senior Vice President and Provost and the Chair-Elect of the Senate shall draw up a slate of nominees who will agree to serve, and the slate will be submitted to the Senior Vice President and Provost and the Committee on Committees for approval. The list of nominees for Library Council membership shall be submitted to the University Senate for approval. Ordinarily, the slate will be presented at the same Senate meeting at which other committee slates are approved. Dates of appointment and beginning of terms shall correspond with those of Senate committees. Replacement of Library Council members will take place through the same consultative process as the initial appointment, with submission of names to the Senate occurring as needed.
- 3. **Operation of the Library Council:** Effective and efficient Library Council operation will require adequate support and full cooperation among the Senate, the Senior Vice President and Provost, the Dean, and their offices.
  - A. The Office of the University Senate or its designee will provide normal committee support to the Council, including maintaining mailing lists, reproducing Library Council documents, keeping a copy of Library Council minutes, maintaining files for the Library Council, and arranging meeting rooms.
  - B. The Office of the Senior Vice President and Provost, through its ex officio Library Council member, will provide liaison to other administrative units, such as the Office of Institutional Research, Planning and Assessment, for their reports, data, or assistance. The Office of the University Senate will also provide website space for the Library Council.
  - C. The Dean of the Libraries will provide the Library Council with internal data, reports, studies, etc. as needed to support the Library Council's work. The Dean will also arrange for unit staff to present testimony concerning such reports as the Library Council finds useful in carrying out its responsibilities. The Dean's assistance to the committee shall also include providing the Library Council members with the opportunity to attend an appropriate orientation session dealing with the Libraries.
  - D. Control of the Library Council's agenda will be the responsibility of the Library Council Chair and the voting members of the Library Council.
  - E. While being responsive to the needs of the Senior Vice President and Provost and the Senate in a timely manner is necessary, the sponsoring parties and the Dean of the Libraries must not attempt to micro-manage the ongoing operation of the Library Council. In turn the Library Council must not attempt to micro manage the Libraries.
  - F. The Library Council shall meet as necessary, but in no case less than once per semester. Meetings may be called by the Chair. In addition, upon receiving a request of any three members of the Library Council, the Chair shall call a meeting. A majority of the voting members of the Library Council shall constitute a quorum for the conducting of official business of the Library Council.

### 4. Operational Relationship of the Library Council to its Sponsors:

A. For purposes of University Senate action, a Library Council created through Senate action will appear in essentially the same role as a standing committee of the University Senate.

- B. The Chair may present reports and recommendations to the Senate but will not have a vote in Senate proceedings, unless he or she is a member of the Senate.
- C. Since the committees of the Senior Vice President and Provost range widely in form and function, and do not operate under a formal plan of organization and bylaws, there is no need to specify the Library Council's standing in the same fashion. For other purposes, such as APAC review of the Unit, the Library Council might be consulted like a College Advisory Council (that colleges will have under the shared governance plan) could be.
- D. The Dean of Libraries will ordinarily meet with the Library Council and have a voice in its deliberations. Since one of the three main functions of the Library Council is to advise the Dean, the Dean shall not formally be a member of the Library Council. On formal reports and recommendations of the Library Council to the University Senate or to the Senior Vice President and Provost, the Dean of the Libraries may send a separate memorandum to the Senate or the Senior Vice President and Provost, as appropriate, supporting or opposing the report or the recommendations, and providing rationale for the Dean's position.
- 5. **Review of the Library Council:** The Library Council and its operations will be reviewed in conjunction with the periodic review of the Senate and the *Plan*.

# APPENDIX 2 BYLAWS OF THE UNIVERSITY RESEARCH COUNCIL

{To be inserted once available}

# APPENDIX 3 BYLAWS OF THE UNIVERSITY IT COUNCIL

Charge to the University Information Technology (IT) Council: The IT Council has the responsibility to facilitate
alignment of vision, priorities, and pace of IT investments and to recommend IT policies to the University Senate
and administration. The IT Council is supported by Working Groups, which facilitate campus-wide communication
related to IT matters.

# A. The Council's Responsibilities to the University Senate:

- 1) Advise on strategic issues involving the University's use of IT, information security, access, retrieval and content stewardship, and telecommunication and knowledge dissemination.
- 2) Bring IT initiatives and proposals to the Senate for consideration and review.
- 3) Keep the Senate informed of strategic IT matters through periodic updates.
- 4) Respond to charges sent to the IT Council by the Senate Executive Committee.
- 5) Provide an annual written report of the IT Council's activities.

# B. The IT Council's Responsibilities to the Vice President for Information Technology and Chief Information Officer (CIO):

- 1) Advise on policy recommendations related to campus technology facilities, equipment, software, and services particularly in the areas of computing (both academic and administrative), networking, and telecommunications.
- 2) Advise on IT planning, including strategic and other major planning for information technology operation and development.

- 3) Advise on the Division of Information Technology's budget, space, and other material resources, in addition to personnel, staffing and other human resources.
- 4) Advise on the Division of Information Technology's administrative policies and practices.
- 5) Respond to requests for review, analysis, and advice made by the Vice President for Information Technology and CIO.

# C. The IT Council's Responsibilities to the Deans, the Campus Student Technology Advisory Fee Committee, and the Campus Community:

- 1) Ensure the distribution of information concerning available campus technology services and how they might be best used to serve the campus community.
- 2) Seek input from current and prospective users concerning types of technology services the campus can provide.
- 3) Respond to input from current users concerning the quality of campus technology services.

### D. To Fulfill Its Responsibilities, the IT Council May:

- Investigate matters concerning the Division of Information Technology and recommend solutions to the University Senate, the Vice President for Information Technology and CIO, or the general campus community.
- 2) Conduct open hearings on major issues concerning the Division of Information Technology and its activities.
- 3) Communicate directly with the campus community on concerns related to the Division of Information Technology's services and policies.
- 2. **Organizational Structure of the IT Council:** The IT Council shall include five standing Working Groups, each of which will have a chair.
- 3. Composition of the IT Council: The IT Council shall consist of eleven appointed members and additional non-voting ex-officio members. The appointed members shall be: the chair, one staff member, one undergraduate student, one graduate student, one professional track faculty member, one tenured faculty member, and the chairs of the five IT Council Working Groups. The non-voting ex-officio members shall be a representative from the University Libraries; a representative from the Office of the Provost; a representative from the Information Technology Advisory Committee (ITAC); and the Vice President for Information Technology and CIO. Additional non-voting ex-officio members may be appointed as needed, by agreement between the Vice President for Information Technology and CIO and the Senate Executive Committee.

#### A. Tenure in Office:

- 1) The IT Council chair should be a tenured faculty member, and is appointed for a single, three-year term. Normally, the chair shall have served as a member of the IT Council. If the chair is serving as a regular member of the IT Council at the time of appointment, a new member shall be appointed to serve the remainder of the term the chair has vacated. The Vice President for Information Technology and CIO and the Senate Chair shall reach an agreement on the IT Council chair, and the joint choice shall be submitted to the University Senate for its approval.
- 2) The five Working Group chairs shall be appointed for staggered two-year terms. No working group chair shall serve more than two terms consecutively. For this purpose, members who have served more than a year should be considered to have served a full term.

- 3) The two faculty members (professional track and tenured) shall be appointed for two-year terms. No faculty member shall serve more than two terms consecutively. For this purpose, members who have served more than a year should be considered to have served a full term.
- 4) The staff member shall be appointed for a two-year term. No staff member shall serve more than two terms consecutively. For this purpose, members who have served more than a year should be considered to have served a full term.
- 5) The two student members shall be appointed for one-year terms. No student member should serve more than two terms consecutively. For this purpose, student members who have served more than half their term should be considered to have served a full term.
- 6) The Dean of the Libraries will appoint a representative from the University Libraries as a non-voting ex officio member of the IT Council.
- The Provost will appoint a representative from the Office of the Provost as a non-voting ex-officio member of the IT Council.
- 8) The Information Technology Advisory Committee (ITAC) will appoint a representative from the committee as a non-voting ex-officio member of the IT Council.
- 9) The Vice President for Information Technology and CIO, or a designee, shall serve as a non-voting exofficio member of the IT Council.
- B. **Qualifications of IT Council Members:** Successful operation of the IT Council requires that its members understand the nature of the Division of Information Technology and represent the best interests of the campus as well as the particular interests of their specific constituencies.
  - IT Council members should be chosen from people who can bring a campus-wide perspective to their deliberations on IT matters and who have shown interest and willingness to foster a good working relationship between the Division of Information Technology and its users.
  - 2) IT Council members should be selected to represent as broad a range of campus disciplines and interests as possible. Faculty members should include representatives from the various disciplines on campus ranging from the arts and humanities and social sciences to the physical and biological sciences and engineering.
- C. The Appointment Process: In the spring of each year, the Senate Office shall notify the Vice President for Information Technology and CIO and the Chair of the Senate of the appointments required for the following academic year. The Vice President for Information Technology and CIO and the Chair of the Senate shall draw up a slate of nominees who will agree to serve, and the slate will be submitted to the Committee on Committees for approval. The final slate of nominees for IT Council membership shall be submitted to the University Senate for approval. Ordinarily, the slate will be presented at the same Senate meeting at which other committee slates are approved. Dates of appointment and beginning of terms shall correspond with those of Senate committees. Replacement of IT Council members will take place through the same consultative process as the initial appointment, with the submission of names to the Senate occurring as needed.

### 4. Operation of the IT Council

A. The Division of Information Technology or its designee will provide normal committee support to the Council, including maintaining mailing lists, reproducing IT Council documents, keeping IT Council minutes and agendas on an IT governance website, and arranging meeting rooms.

- B. Control of the IT Council's agenda will be the responsibility of the IT Council chair and the voting members of the IT Council.
- C. While being responsive to the needs of the Vice President for Information Technology and CIO and the Senate in a timely manner is necessary, the Working Groups and the sponsoring parties as well as the Deans, the Campus Student Technology Fee Advisory Committee, and the campus community must not attempt to micro-manage the ongoing operation of the IT Council. In turn, the IT Council must not attempt to micro-manage the Division of Information Technology.
- D. The IT Council should typically meet once every month and shall meet at least once per semester. Meetings will be scheduled by Division of Information Technology staff, in consultation with the IT Council chair and the Vice President for Information Technology and CIO.
- 5. **Working Groups of the IT Council:** The Working Groups will serve in an advisory capacity to the IT Council. These groups should carry out research and make recommendations on IT issues, and shall each work with the appropriate Division of Information Technology staff member, as appointed by the Vice President for Information Technology and CIO.

# A. The five Working Groups shall be:

- 1) IT Infrastructure Working Group, which focuses on building and maintaining a sound, advanced, secure, and productive physical information technology infrastructure (including but not limited to facilities, hardware, networks, and software) capable of supporting broad and effective use by students, faculty, and staff throughout the institution, including remote University members such as agricultural extension offices.
- 2) Learning Technologies Working Group, which provides the vision, priorities, and pace for enterprise learning technology solutions and services to be undertaken on campus. Its work focuses on endorsing the adoption of new learning technology solutions, as well as making recommendations for upgrading or decommissioning existing services. Working group members are nominated by the Deans.
- 3) Research Technologies Working Group, which provides the vision, priorities, and pace for enterprise research technology solutions and services to be undertaken on campus. Its work focuses on endorsing the adoption of new research technology solutions, as well as making recommendations for upgrading or decommissioning existing services. Working group members are nominated by the Deans.
- 4) Administrative Systems Working Group, which advises the Vice President for Information Technology and CIO in matters of enterprise-wide administrative system technology decisions and priorities.
- 5) IT Security Working Group, which advises the Vice President for Information Technology and CIO on IT security matters. The focus is on securing the integrity of information technology resources, safeguarding institutional information, protecting the privacy of University community members in their use of IT, and ensuring the continuity of the institution's IT resources and information repositories in the face of possible disaster scenarios.
- B. Composition of the Working Groups: Each Working Group will have a chair appointed by the Vice President for Information Technology and CIO and the Senate Chair for a two-year term. The membership of each Working Group will be appointed by the Vice President for Information Technology and CIO unless otherwise specified above (5.A.2 and 3), but will be flexible so that additional members can be engaged in the decision-making and review process as appropriate. The membership of each Working Group shall include a combination of faculty, staff, and students.
- C. Terms on Working Groups shall be two (2) years for faculty and staff. Appointments to two-year terms shall be staggered: that is, as far as practical, half of the terms from each faculty or staff constituency shall expire each year. Terms shall begin on July 1 of the appropriate year.

D. Meetings of the Working Groups: The Working Groups usually meet three to four times a semester.

## E. Working Group Responsibilities:

- 1) Provide knowledge in a particular area and serve as an advisory board, by which the IT Council can route items for review and comment.
- 2) Submit proposals and issues to the IT Council for consideration and/or funding.
- 3) Assist in the annual review and update of the Information Technology Strategic Plan.

# 6. Operational Relationship of the IT Council to its Sponsors:

- A. For purposes of University Senate action, the IT Council will appear in essentially the same role as a standing committee of the University Senate.
- B. The IT Council chair may present reports and recommendations to the Senate but will not have a vote in Senate proceedings, unless he or she is a member of the Senate.
- C. The Vice President for Information Technology and CIO is an ex-officio member of the IT Council and has a voice in its deliberations.
- 7. **Review of the IT Council:** The IT Council and its operations will be reviewed in conjunction with the periodic review of the Senate and the Plan.
- 8. Amendments: Amendments to these Bylaws shall be provided to the IT Council members a minimum of seven calendar days in advance of any regular meeting. Approval shall require a two-thirds vote of the present and voting regular membership of the Council. Upon approval, a revised copy of the Bylaws shall be sent to the Senate Office.

# APPENDIX 4 CHARTER OF THE UNIVERSITY ATHLETIC COUNCIL

The University of Maryland at College Park is dedicated to higher learning, research, and public service. An intercollegiate athletic program can significantly contribute to the learning and the public service components of the Campus Mission. The operation of a successful athletic program fosters spirit, identity and a sense of pride within the campus community and provides talented student-athletes with the opportunity to enrich their collegiate experience through participation in a challenging and competitive athletic program. Excellence of the athletic program at College Park stems not only from successful competition, but more importantly, from the general involvement in the Campus milieu of student-athletes who will earn degrees and who in other respects, embody qualities with which the institution can identify. Most importantly, both athletic success and academic integrity are the crucial elements in judging the excellence of the athletic program at the University of Maryland at College Park.

The importance of faculty involvement and influence in the institutional control and operation of an excellent athletic program cannot be overestimated. Faculty advice and participation will enhance the integrity of the athletic program in terms of academic performance, rules compliance, and compatibility of athletic programs with the mission of the campus.

#### PURPOSE OF THE ATHLETIC COUNCIL

First and foremost, the Athletic Council exists to help the University develop and maintain the best possible intercollegiate athletic program consistent with the academic integrity of the institution and the academic and social development of student athletes. The Athletic Council is the primary body, which advises the President on all matters relating to intercollegiate athletics. It is responsible for formulation and recommendation of policy matters affecting intercollegiate athletics and for monitoring the implementation of such policy by the intercollegiate athletics program. The Council, on behalf of the President, provides the necessary faculty input and participation in intercollegiate athletics as required by the Big Ten Conference, National Collegiate Athletic Association and the University of

Maryland at College Park. The Council does not execute policy but serves to influence policy development and administration.

This document delineates the responsibilities, processes, and membership of the Athletic Council at the University of Maryland at College Park. It is expected that the Council will be proactive in its task of preparing policy recommendations and monitoring their implementation by the intercollegiate athletics program. The Council expects to have the full support of the Campus in the responsible performance of its duties.

# **FUNCTION/DUTIES OF THE ATHLETIC COUNCIL**

The major function of the Athletic Council is to assist the President and the Director of Intercollegiate Athletics in the exercise of "institutional responsibility and control of intercollegiate athletics" as required by the constitution of the Big Ten Conference, the National Collegiate Athletic Association and the University of Maryland at College Park. The Council functions in advisory, compliance, liaison, and representative capacities. The Athletic Council shall meet at least four times each year, twice in each semester, and at such other times as needed to carry out the duties of the Council. Specific duties of the Council shall include but not be limited to the following:

- 1. Promote an understanding of intercollegiate athletics among faculty, students, staff, alumni and other members of the University of Maryland at College Park community.
- 2. Promote the adoption and implementation of appropriate policies for the admission and continuing eligibility of student athletes at the University of Maryland at College Park.
- 3. Monitor the preparation of the athletic budget by the Athletic Director during the regular budgetary process and make recommendations to the Athletic Director and the President concerning sources (i.e. student athletic fees) and allocations of funds.
- 4. Participation in the selection process for the Director of Intercollegiate Athletics and the head coaches in all sports including, if possible, informal meetings of the final candidates with the Executive Committee in the interview process. A faculty member from the Athletic Council should be included on all search committees for head coaches.
- Establish criteria and make recommendations, with the advice of the Athletic Director, regarding which sports shall be certified as intercollegiate sports.
- 6. Recommend policies concerning athletic schedules, practice, the number of contests to be played each year in each sport and the NCAA category of schools with which it is desirable to compete.
- 7. Establish guidelines for and make recommendations regarding the acceptance of invitations to post-season events, special holiday games, or other events outside the regular season schedule.
- 8. Review and formulate policies concerned with substance abuse that will provide protection to the health of student-athletes and ensure that such policies have a strong educational emphasis.
- 9. Review and endorse policy on physical facilities necessary for the conduct of a competitive Division I-A program.
- 10. Review and formulate policies on recruitment and the awarding of athletic grants and scholarships to student-athletes who meet eligibility standards.
- 11. Review and approve the criteria for departmental awards in recognition of athletic and academic achievement.
- 12. Review athletic event price schedules, seating priorities and allocation of tickets to various groups.
- 13. Monitor the advisement, academic support and counseling services available to student-athletes.
- 14. Review and formulate policy concerning the conduct of home athletic contests, particularly with respect to the protection and safety of participants and spectators.

- 15. Review and formulate policy regarding the expectations of and behavior of coaches and student-athletes.
- 16. Review and formulate policy regarding the expectations of and behavior of cheerleaders and their advisors.
- 17. Assist with the development of official reports to be submitted by the President for filing with the conference or appropriate associations.
- 18. Review with appropriate authorities the financial audits of the Department of Intercollegiate Athletics.
- 19. Monitor the activities of the Department of Intercollegiate Athletics to make sure that they are in compliance with Conference (Big Ten) and Association (NCAA) bylaws, regulations and legislation.

In fulfilling its functions/duties, the Athletic Council

- must maintain confidentiality;
- shall have available to it complete information on all items which appear for its consideration and shall have full opportunity for discussion of each item before action is taken;
- shall have available full and current information on the financial, academic and related activities of the intercollegiate athletics program; and
- is authorized to recommend to the President the employment of experts from outside the Campus when their advice is needed.

#### RESPONSIBILITIES OF THE CHAIR OF THE ATHLETIC COUNCIL

The Athletic Council has a Chair who is selected by the President from the faculty. The duties of the Chair shall include:

- 1. Serve as a spokesperson for the Council in all contacts with the media.
- 2. Serve as the Faculty Representative to the Big Ten Conference and the National Collegiate Athletic Association (NCAA).
- 3. Chair meetings of the Athletic Council and the Executive Committee of the Council.
- 4. Call regular meetings of the Athletic Council and such special meetings as may be necessary.
- 5. Prepare the agenda for meetings of the Athletic Council and of the Executive Committee of the Council.
- 6. Represent the campus, as authorized by the President, at meetings of the NCAA, Big Ten, United States Intercollegiate Lacrosse Association, United States Olympic Committee, Intercollegiate Athletic Association of America, College Football Association and other groups which establish international, national and regional policies for intercollegiate athletics.
- 7. Advise the President and serve as spokesperson to the faculty on behalf of the President on appropriate matters.
- 8. Report to the President on all actions taken by the Athletic Council.
- 9. Work with the Director of Intercollegiate Athletics in coordinating and carrying out the functions of the Athletic Council.
- 10. Monitor activities of the Department of Intercollegiate Athletics and confer regularly with the President on matters which should come to the President's attention.
- 11. Ensure that required reports and recommendations from the Athletic Council are provided to the President.
- 12. Report to the President and the Athletic Director on the concerns of the Athletic Council relative to athletics and to interpret to the faculty and other groups the University's athletic policies and activities.

- 13. Ensure that all actions of the Chair and the Executive Committee made on behalf of the Council are properly recorded and reported to the full membership of the Council in a timely manner.
- 14. Coordinate with the President's Office all financial support necessary to carry out the duties of Chair, including the development of an annual budget for this support; and the approval of all requests for expenditures and expense reimbursements made for this purpose. The President's Office is the administrative unit responsible for providing appropriate financial support to the Chair of the Athletic Council/Faculty Athletic Representative, and for approving both the annual budget request for this support as well as all expenditures, and expense reimbursements made for this purpose.
- 15. Know, recognize, and comply with the laws, policies, rules and regulations governing the University and its employees, and the rules of the National Collegiate Athletic Association (the "NCAA") and the Big Ten. Inform the Athletic Department Compliance Officer immediately of any suspected violation. Assist, as requested, in the investigation and reporting of those violations.

### RESPONSIBILITIES OF THE VICE CHAIR OF THE ATHLETIC COUNCIL

The Athletic Council has a Vice-Chair who is selected by the President from the faculty. The duties of the Vice-Chair shall include:

- 1. Assist the Chair of the Council with conducting the business and meeting of the Council.
- 2. Conduct meetings of the Council in the absence of the Chair.
- 3. Write periodic articles for University publications about the actions of the Council.
- 4. Serve on the Executive Committee of the Council.
- 5. Coordinate the activities of and serve as an ex officio member to standing committees of the Council.
- 6. Know, recognize, and comply with the laws, policies, rules and regulations governing the University and its employees, and the rules of the National Collegiate Athletic Association (the "NCAA") and the Big Ten Inform the Athletic Department Compliance Officer immediately of any suspected violation. Assist, as requested, in the investigation and reporting of those violations.

# ATHLETIC COUNCIL MEMBERSHIP

Intercollegiate Athletics plays an important role in fostering pride and spirit in the University community. The Athletic Council membership is designed to be representative of this community and shall consist of faculty, administration, staff, students and alumni. The membership shall include minorities, women and men, and thorough consideration will be given to ensure a balanced representation on the Council. The Athletic Council shall consist of twenty voting and five non-voting members appointed by the President or elected by the Senate as follows:

Voting Members of the Athletic Council

- The Athletic Council has a Chair who is selected by the President from the faculty. The duration of the Chair's
  membership on the Council is determined by the President. The initial appointment is for a five year term
  which may be renewed by the President.
- The Athletic Council has a Vice-Chair who is selected by the President from the faculty. The duration of the Vice-Chair's membership on the Council is determined by the President. The initial appointment is for a three year term which may be renewed by the President.
- Seven faculty members of the Athletic Council will be elected by the Senate. These elected faculty members
  will serve for a three year period and are not eligible to serve a second consecutive three year period. The
  Senate should make every effort to assure diversity among the elected members.
- The Faculty member who is Chair of the Campus Affairs Committee of the Senate or a designee from the Committee who must be a faculty member is a member of the Athletic Council.

- One Academic Dean appointed by the Provost. The appointment is for a one year term which may be renewed
  by the Provost.
- Two staff members, one who is appointed by the President for a three year period and one who is elected for a three year period by the Senate. These staff members will serve on a staggered basis and are not eligible to serve a second consecutive three year period.
- The Vice President for Student Affairs.
- One representative from the "M" Club. The appointment is for one year.
- One representative from the Terrapin Club. The appointment is for one year.
- One student representative from the Student Government Association. The appointment is for one year.
- One undergraduate female athlete. The appointment is for two years and the student should maintain eligibility in her sport.
- One undergraduate male athlete. The appointment is for two years and the student should maintain eligibility in his sport.
- One graduate student. The appointment is for two years and the student should maintain good standing in the Graduate School.

Non-Voting Members of the Athletic Council

- The Director of Intercollegiate Athletics.
- A Representative from the President's Office.
- A Representative of the Office of General Counsel.
- The Director of the Student Health Services.
- The Director of the Office of Alumni Programs for the University of Maryland at College Park.
- A current head coach selected by the coaches as their representative. This appointment will be a one-year
  appointment with a three year limit.

In making all non-elected appointments to the Athletic Council, the President should solicit recommendations from the following advisory groups or persons: Executive Committee of the Athletic Council, President of the Student Government Association, President of the Graduate Student Government, Dean of the Graduate School, and Director of Intercollegiate Athletics. The term of office of all members of the Council shall begin with the first meeting of the new academic year.

Vacancies occurring on the Council due to resignation or other cause will be filled as they occur. If the vacancy is one of the members of the Council elected by the Senate, the Senate will be asked to elect a replacement to fill the vacancy. For all other vacancies, the President will solicit nominations from the appropriate groups and appoint a replacement to fill the remainder of the unexpired term. Persons appointed to fill a partial term on the Council will be eligible for election or appointment to a full term as appropriate for their membership category.

### **COMMITTEES OF THE ATHLETIC COUNCIL**

Committees of the Athletic Council shall include an Executive Committee, Standing Committees of the Council, and Ad-Hoc Committees as needed. The major responsibilities and membership of these Committees of the Athletic Council follow.

1. Executive Committee. The membership of the Committee is as follows: The Chair of the Athletic Council who will

serve as chair, the Vice-Chair of the Athletic Council, chairs of the five standing committees of the Athletic Council, the representative from the President's office, and a staff or student member of the Athletic Council. If one or more of the Chairs of the standing committees are not faculty, the membership of the Executive Committee will be adjusted to include four faculty in addition to the Chair. Total membership of the Executive Committee will not exceed eight at any time. The responsibilities of the Executive Committee shall include the following:

- Meet on matters calling for immediate action and at times when meetings of the full Athletic Council are not possible.
- Identify and assign problems to standing subcommittees and ad-hoc committees for study and receive reports from these committees.
- Serve as the personnel committee of the Council upon request of the President.
- Review compliance reports submitted by the Department of Intercollegiate Athletics and ensure that the Department is in compliance with all Conference and Association policies.
- Advise the President on an emergency basis.
- Recommend faculty and staff for membership on the Athletic Council.
- Standing Committees of the Athletic Council. The Chair of the Athletic Council will select the Chairs of the Standing Committees and will appoint each committee and, with the exception of the Academic Committee, will appoint each committee after soliciting volunteers from the Council membership.
  - a. Academic Committee. All faculty members of the Council are members of the committee. The general role of the Academic Committee is to ensure that appropriate academic standards are established and maintained for all student-athletes and that all participants recognize the priority of successful academic performance by all student-athletes. In fulfilling this function, the Committee shall make appropriate recommendations to the Athletic Council. In particular, the Committee shall have the following duties:
    - Recommend policies and procedures regarding standards and criteria for admission of student-athletes.
    - Recommend academic policies and procedures regarding standards and criteria for continuing eligibility of student-athletes to participate in intercollegiate sports.
    - Consider and decide academic appeals of student-athletes concerned with eligibility.
    - Review every semester the academic program and progress of student-athletes.
    - Recommend policies for and monitor the activities of the academic support services provided to the student-athletes.
    - Recommend policies regarding post-season and tournament participation by athletic teams.
    - Recommend policies regarding scheduling and practice time.
  - b. **Budget and Facilities Committee.** The general purpose of this Committee is to monitor but not manage those activities of the Athletic Department pertaining to budget and facilities. In fulfilling this function, the Committee shall make appropriate recommendations to the Athletic Council. More specifically, responsibilities of the Committee shall include the following:
    - Monitor the preparation of the athletic budget(s) by the Director of Intercollegiate Athletics.
    - Review and analyze for the Council the final budget(s) submitted by the Director of Intercollegiate Athletics to the President.
    - Establish criteria and make recommendations, with the advice of the Director of Intercollegiate Athletics, regarding which sports shall be certified as intercollegiate sports.

- Review policies regarding the number and distribution of athletic scholarships to be awarded annually.
- Review and recommend policies for athletic event price schedules, seating priorities and allocation of tickets to various groups.
- Review and recommend policies regarding utilization and development of intercollegiate athletic facilities.
- Monitor the financial accountability of the Department of Intercollegiate Athletics.
- c. Student Life Committee. This Committee is concerned with all non-academic aspects of the student-athlete's involvement with the University. In fulfilling this function, the Committee shall make appropriate recommendations to the Athletic Council. More specifically, the responsibilities of the Committee shall include the following:
  - Review and recommend policies concerning the nature and type of health screening and drug testing.
  - Review and recommend policies regarding practice schedules.
  - Review and recommend policies for determining when health and other non-academic factors will be used to restrict a student's involvement in intercollegiate athletics.
  - Review and recommend policies for and monitor activities of non-academic support programs and placement services.
  - Review and recommend policies regarding scholarship awards and retention of these awards.
  - Review and recommend policies for housing assignments.
  - Assist the Athletic Council in assuring the personal and social development of all student-athletes and their full integration into campus life.
- d. **External Affairs Committee.** This Committee is concerned with external activities of the Department of Intercollegiate Athletics. In fulfilling this function, the Committee shall make appropriate recommendations to the Athletic Council. More specifically, the responsibilities of the Committee shall include the following:
  - Review and endorse fundraising activities.
  - Review and recommend policies for complementary distribution of tickets to athletic events.
  - Review and recommend guidelines and/or policies for all sports marketing activities (i.e. sports camps, special events, endorsements, etc.)
  - Review and recommend guidelines and/or policies for interactions with alumni and friends of the Athletic Department including the Terrapin Club, the "M" Club, and the Maryland Education Foundation.
  - Review and recommend policies and/or guidelines for all media interactions.
- e. **Professional Sports Counseling Panel (PSCP).** The PSCP is a committee of the Athletic Council authorized under NCAA by-law 12.3.4 to advise and assist student athletes in preparation for professional athletic careers. Consonant with its charge, the University of Maryland, College Park PSCP provides:
  - Education and advice to student athletes about NCAA amateurism rules and professional sports careers.
  - Oversight to the Athletic Department's implementation of University and NCAA regulations regarding contacts between student athletes and agents.
  - Advice to the Athletic Council on matters related to its charge.
- 3. Ad-Hoc Committees. The Chair of the Athletic Council, upon advice of the Council, will appoint Ad-Hoc

Committees as needed. Membership on these committees will be on a volunteer basis or by appointment by the Chair of the Council after seeking advice from the Executive Committee.

#### MEETINGS OF THE ATHLETIC COUNCIL

The Chair of the Council serves as the spokesperson for the Council. Meetings of the Council are open only to Council members and invited guests. Individuals who are not members of the Council, but who wish to attend a specific meeting should seek the prior approval of the Chair. Information provided to Council members concerning specific personnel or compliance matters will not be divulged by individual members without permission of the Chair.

# APPENDIX 5 PROCEDURES FOR ELECTIONS OF UMCP REPRESENTATIVES TO THE COUNCIL OF UNIVERSITY SYSTEM FACULTY (CUSF)

The Chair of CUSF is not a member of CUSF. Thus, if the Chair is from College Park, a replacement must be named. At the end of his/her term as Chair, if his/her term on CUSF is not finished, he/she resumes his/her position as a CUSF member.

The normal term for CUSF representatives is three (3) years, with two alternates serving three (3) year terms; if both alternates are elected at the same time, priority to be a replacement shall be in order of votes received. If a regular representative is unable to serve out his/her term, an alternate replaces him/her for the remainder of the term, and a new alternate is named. The replacement representative shall be chosen in order of number of votes received. The Office of the University Senate will identify a replacement alternate subject to confirmation by the Senate Executive Committee.

The University Senate will elect representatives to CUSF each spring. The Senate Nominations Committee will solicit candidates and will present a slate to the Chair of the Senate with at least one (1) candidate for each vacant position to be filled. At the Transitional Meeting of the Senate, faculty Senators will vote to elect representatives to CUSF. Each faculty Senator shall have as many votes as there are open positions. If there are more candidates than positions, the person(s) receiving the most votes, in order, are declared representatives. The person receiving the next most votes is declared alternate. The remaining person, in order of vote tally, will be asked to move into the alternate position if the previous paragraph comes in to play. A record of the outcome of the election will be retained by the Executive Secretary and Director of the University Senate. If there are not sufficient candidates, or the pool of candidates is exhausted, representatives are chosen by the Executive Committee.

# APPENDIX 6 PROCEDURES FOR ELECTIONS OF UMCP REPRESENTATIVES TO THE COUNCIL OF UNIVERSITY SYSTEM STAFF (CUSS)

The mission of the Council of University System Staff (CUSS) is to provide a voice for Staff employee concerns in reference to basic decisions that affect the welfare of the University System of Maryland (USM) and its employees. CUSS speaks for all non-exempt and exempt staff employees on Regular and Contingent II Status, who are not represented by a union under collective bargaining.

CUSS is comprised of Staff employees representing each USM institution and the USM Office (USMO). Institution membership is proportionate to the number of Staff employees at the individual institutions, with a minimum of two (2) primary members and two (2) alternate members per institution. Representation on CUSS from each constituent institution is apportioned according to the following formula: 1 to 999 eligible employees, 2 representatives; over 1000 eligible employees, 3 representatives. Staff at each constituent institution shall also select an alternate who shall substitute for a regular member of CUSS when needed. Alternates should be selected at the same time and in the same manner as regular members. A delegation may include more than one (1) alternate who is eligible to cast a vote for an absent member provided the member has given prior notification to the Chair of CUSS. The University of Maryland, College Park is entitled to three (3) representatives, and up to three (3) alternates.

As defined in 6.10.f of the Senate *Bylaws*, the Senate Staff Affairs Committee is responsible for administering the CUSS nomination and election process. Definitions of eligible staff shall be determined by the Board of Regents and CUSS. The CUSS elections will be administered in the spring semester every other year, as the terms of the current CUSS representatives are expiring. The Staff Affairs Committee will solicit candidates from the eligible staff population and will present ballots to the same population with at least one (1) candidate for each vacant position to be filled.

Eligible staff employees will vote to elect representatives to CUSS. If there are more candidates than positions, the person(s) receiving the most votes, in order, are declared representatives. The person(s) receiving the next most votes are declared alternate(s). A record of the outcome of the election will be retained by the Executive Secretary and Director of the University Senate.

New members shall begin their terms August 1. The normal term for CUSS representatives and alternates is two (2) years. If a regular representative is unable to serve out his/her term, an alternate replaces him/her for the remainder of the term, and a new alternate is named. The replacement representative shall be chosen in order of number of votes received.

# APPENDIX 7 PROCEDURES FOR REVIEW OF COLLEGE AND SCHOOL PLANS OF ORGANIZATION

- 1. In accordance with Article 11 of the Plan, each College, School, Department and other Academic Program, and the Library, shall have a Plan of Organization.
  - a. The Plan of Organization of each College, School, and the Library shall be reviewed by the University Senate according to the procedures detailed in section 2 of this appendix. All revisions to such Plans of Organization must be approved by the University Senate and the President prior to taking effect.
  - b. The Plan of Organization of a Department or other Academic Program shall be reviewed and revised by the Faculty Advisory Committee of the College to which it belongs. In the review and revision of such Plans, the University Senate may act in an advisory capacity if asked to do so by the College.
- 2. Each College, School, and the Library shall review and revise its Plan of Organization in accordance with Article 11.3 of the Plan and shall submit it to the University Senate for review.
  - a. Revised Plans of Organization shall be reviewed by the Senate Elections, Representation, and Governance (ERG) Committee for compliance with the University's Plan of Organization, University policy, and best practices of shared governance.
  - b. The Senate Faculty Affairs Committee shall review the Appointment, Promotion, and Tenure or Permanent Status section of each Plan and any related documentation for compliance with the University's APT Policy. The Senate Faculty Affairs Committee shall also review the Appointment, Evaluation, and Promotion Policy and any related documentation for compliance with University policies on professional track faculty and the University's Guidelines for Appointment, Evaluation, and Promotion of Professional Track Faculty.
  - c. The ERG and Faculty Affairs Committees shall communicate any concerns or requested revisions to the respective College, School, or Library.
  - d. Once all necessary revisions have been made, the ERG and Faculty Affairs Committees shall certify that they find the Plan to be in compliance and the revised Plan of Organization shall be submitted to the College Assembly or equivalent for approval.
  - e. Upon approval of the College Assembly or equivalent, the ERG Committee shall submit the revised Plan and its accompanying report to the Senate Executive Committee for review and placement on the Senate Agenda.
  - f. The revised Plan of Organization shall require final approval by the University Senate and the President.
- 3. During the initial implementation of a recently approved Plan of Organization, a College, School, or the Library may submit additional minimal or technical amendments to the Senate within one year of final approval by the University President. These revisions will undergo an expedited review process by the Senate ERG Committee, and by the Faculty Affairs Committee if appropriate. The committee(s) shall review only those amendments submitted by the College, School, or the Library, and shall not conduct a full review of the Plan. Upon approval by the ERG Committee (and the Faculty Affairs Committee, if necessary), the amendments shall be submitted to the College Assembly, the Senate Executive Committee, the Senate, and the President according to the procedures outlined above in section 2 d-f.

4. Until a revised Plan of Organization is approved by the University Senate and President, the version of the Plan of Organization of each College, School, and the Library that was most recently approved by the University Senate and President remains in effect, and provides the rules under which the College, School, or the Library must review and approve future revisions to its Plan. The University Plan of Organization supersedes any provisions in the Plan of any College, School, the Library, Department, or Academic Program that are in conflict with the purpose, applicability, or intent of the University Plan.

# Dates of Approval, Updates and Amendments to the Senate Bylaws

Approved, Campus Senate, October 9, 1986 Approved, Board of Regents, February 6, 1987

Updated, July 11, 1988

Amended, February 13, 1986 Amended, December 7, 1986

Amended, May 7, 1990

Amended, September 13, 1990 Amended, November 15, 1990

Amended, October 14, 1993

Amended, December 6, 1993

Amended, March 31, 1994

Amended, April 18, 1994

Amended, May 5, 1994

Amended, November 10, 1994

Amended, August 28, 1996

Amended, May 15, 1997

Amended, March 5, 1998

Amended, April 2, 1998

Amended, April 6, 2000

Amended, February 12, 2001

Amended, September 19, 2002

Amended, February 3, 2003

Amended, October 16, 2003

Amended, April 19, 2004

Amended, April 4, 2005

Amended, May 15, 2007

Amended, May 8, 2008

Amended, October 16, 2008

Amended, February 9, 2009

Amended, May 4, 2009

Amended, November 12, 2009

Amended, March 3, 2010

Amended, February 9, 2011

Amended, May 4, 2011

Amended, March 8, 2012

Amended, April 19, 2012

Amended, May 2, 2013

Amended September 18, 2013

Amended, April 15, 2015

Approved after 2015 Plan of Org Review, May 4, 2015

Amended, November 20, 2015

Amended, December 14, 2015

Amended, February 18, 2016

Amended, March 18, 2016

Amended March 24, 2017

Amended November 8, 2017

Amended May 3, 2019

Amended February 7, 2020

Amended March 30, 2020

Amended November 12, 2020

Amended December 10, 2020

Amended April 9, 2021

Amended September 10, 2021

Amended September 29, 2023

# REPORT | #19-20-15

Senate Elections, Representation, & Governance (ERG) Committee

# Campus Transportation Advisory Committee (CTAC) Senate Representation

# 2024-2025 Committee Members

Kim Gonzalez (Chair)
Janet Adesina (Graduate Student)
Paul Brown (Faculty)
Sophia Conrad (Undergraduate Student)
Jon Crocker (Faculty)
Tom Hatcher (Faculty)
Lian Kish (Exempt Staff)
Suraj Krishnamurti (Graduate Student)
Jessica O'Hara (Faculty)

Polly O'Rourke (Faculty)
Pamela Phillips (Ex-Officio- Associate VP IRPA Rep)
Madeline Reynoso (Undergraduate Student)
Sajeela Yagub (Faculty)

Date of Submission February 28, 2025

# **BACKGROUND**

At the October 10, 2024 University Senate Meeting, Senator Lyons, GRAD, JOUR inquired about the status of legislation (Senate Document 19-20-15) regarding the Senate representation on the Campus Transportation Advisory Committee (CTAC). The Senate voted in favor of charging the Elections, Representation, & Governance (ERG) Committee with the review and re-evaluation of the relationship between CTAC and the Senate.

On November 15, 2024, the Senate Executive Committee (SEC) provided a charge document (Appendix 1) to the ERG Committee that included reviewing the University Senate Bylaws Article 12 on the Annual Transition of the Senate, consulting with stakeholders, and considering the representative and practical implications of any changes to the University Bylaws (Appendix 2) and CTAC election processes.

# **COMMITTEE WORK**

The ERG Committee began its consideration of the charge in November 2024. The committee reviewed supplementary documentation, provided by the Vice President for Administration and Finance, following the Senate Meeting action. The first described the context and history of the charge, dating back to 2019, as well as the current status of proposed changes, which indicated that there was no need or desire to change the way in which CTAC members were selected. The document fully supported the ERG Committee's ongoing process to consult with stakeholders and submitting a full report to the Senate.

The ERG Committee also reviewed the original memo, dated November 12, 2019, which described a hope that the Senate could consider amending the structure of CTAC, s it is not a Senate Council, and a different structure may assist with practical, sustainable transportation practices on campus.

After familiarizing themselves with the history of the proposal, the committee then invested time in consulting with stakeholders about the current status of the CTAC representation and election process. This included consultations with Senator Lyons who brought forward the Senate action, the University Senate's Nominations Committee, Faculty Affairs Committee, Staff Affairs Committee, and Student Affairs Committee.

The general feedback from all committees was that the current process as outlined in the University Bylaws had little to no effect on campus communities, and that any concerns with transportation or safety lay outside the representation process of CTAC. The ERG Committee discussed that some comments from the consultative committees indicated that more education or outreach about the Senate-Elected process may be necessary, as some members were unaware of the details about CTAC or Senate-Elected Committee and Council processes.

The ERG Committee also consulted with members of the current Campus Transportation Advising Committee (CTAC) to elicit responses about any potential changes to representation. The feedback from CTAC was consistent with the other committees, with responses indicating that there was no need to adjust the current representation, or process by which representatives were selected. Members direct responses include the following.

- "I believe that the members on CTAC for the past 4 years have been competent and enthusiastic in their advocacy positions. At times, experts and guests are invited to share more information, which I believe is the right amount of flexibility in membership/ attendance for this group."
- "I don't see what advantage this [change in appointment] would provide to the current work on the CTAC."

On January 28, 2025 the ERG Committee discussed the compilation of consultations, and reviewed the work of the semester. Understanding the context that the 2019 initiative to change representation was not reflective of current administrative or membership goals, and that campus constituency consultation revealed no desire for change, the committee agreed that no change to the current Bylaws was necessary.

The University Senate Bylaws were not reviewed by the Office of General Counsel, as the ERG Committee was proposing no change to the current Bylaws.

After due consideration, the ERG Committee voted in favor of "no Senate action" for the Campus Transportation Advisory Committee (CTAC) Senate Representation (Senate Document #19-20-15).

# RECOMMENDATIONS

The ERG Committee recommends that no change be made to the <u>University of Maryland Senate</u> <u>Bylaws</u>, and that there be a vote of No Senate Action on remaining legislation related to this item (Senate Document #19-20-15).

### **APPENDICES**

Appendix 1 - Charge from the Senate Executive Committee

Appendix 2 - University Senate Bylaws



# CHARGE

Charged: November 15, 2024 | Deadline: February 28. 2025

# Review of The Campus Transportation Advisory Committee (CTAC) Senate Representation (Senate Document #19-20-15)

# Elections, Representation & Governance Committee | Chair: Kim Gonzalez

The Senate Executive Committee (SEC) and Senate Chair Sly request that the Elections, Representation & Governance (ERG) Committee review the proposal entitled Campus Transportation Advisory Committee (CTAC) Senate Representation.

Specifically, the ERG Committee should:

- 1. Review the following:
  - a. The proposal entitled *Campus Transportation Advisory Committee (CTAC)*Senate Representation.
  - b. The <u>University Senate Bylaws</u>, Article 12 *Annual Transition Of The Senate*.

# 2. Consult:

- a. Members or representatives of the Campus Transportation Advisory Committee.
- b. A representative from the Office of General Counsel on any proposed changes to the University Senate Bylaws, Article 12.
- c. The University Senate's Nominations, Faculty Affairs, Staff Affairs, and Student Affairs Committees to solicit feedback on CTAC member elections.

# 3. Consider:

- a. The representative and practical implications of any changes to the University Bylaws and CTAC election processes.
- 4. If appropriate, recommend whether the policy should be revised and if so, provide suggested revisions.

We ask that you submit a report to the University Senate Office no later than **February 28**, **2025**. If you have questions or need assistance, please contact the University Senate Office, <a href="mailto:senate-admin@umd.edu">senate-admin@umd.edu</a>.

Appendix 2- University Senate Bylaws

# **BYLAWS**

# **OF THE**

# **UNIVERSITY SENATE**

# **AT THE**

# **UNIVERSITY OF MARYLAND**



# BYLAWS OF THE UNIVERSITY SENATE University of Maryland, College Park TABLE OF CONTENTS

ARTICLE 1 - AUTHORIZATION	3
ARTICLE 2 - MEMBERSHIP	3
ARTICLE 3 - MEETINGS	5
ARTICLE 4 - EXECUTIVE COMMITTEE	6
ARTICLE 5 - COMMITTEES OF THE SENATE	8
ARTICLE 6 - STANDING COMMITTEE SPECIFICATIONS  6.1 Academic Procedures and Standards Committee	
6.2 Campus Affairs Committee	12
6.4 Educational Affairs Committee	13
6.5 Elections, Representation, and Governance Committee	15
6.7 Faculty Affairs Committee	
6.9 Staff Affairs Committee	17
6.11 Student Conduct Committee	
ARTICLE 7 - SPECIAL COMMITTEE ON UNIVERSITY FINANCE	18
ARTICLE 8 - UNIVERSITY COUNCILS	20
ARTICLE 9 - UNIVERSITY COUNCIL SPECIFICATIONS	
9.1 University Library Council	23
9.3 University IT Council	
ARTICLE 10 - THE ATHLETIC COUNCIL	
ARTICLE 11 - DUTIES OF THE EXECUTIVE SECRETARY AND DIRECTOR	
ARTICLE 12 - ANNUAL TRANSITION OF THE SENATE	
APPENDIX 1 - BYLAWS OF THE UNIVERSITY LIBRARY COUNCIL	
APPENDIX 3 - BYLAWS OF THE UNIVERSITY RESEARCH COUNCIL	
APPENDIX 4 - CHARTER OF THE UNIVERSITY ATHLETIC COUNCIL	
APPENDIX 5 - PROCEDURES FOR ELECTIONS OF UMCP REPRESENTATIVES TO TI	
OF UNIVERSITY SYSTEM FACULTY (CUSF)	
APPENDIX 6 - PROCEDURES FOR ELECTIONS OF UMCP REPRESENTATIVES TO TUNIVERSITY SYSTEM STAFF (CUSS)	
APPENDIX 7 - PROCEDURES FOR REVIEW OF COLLEGE AND SCHOOL PLANS ORGANIZATION	OF
Dates of Approval, Updates and Amendments to the Senate Bylaws	

# BYLAWS OF THE UNIVERSITY SENATE The University of Maryland, College Park

# ARTICLE 1 AUTHORIZATION

1.1 These *Bylaws of the University Senate* (hereafter referred to as the *Bylaws*) are adopted according to Article 7 of the *University of Maryland Plan of Organization for Shared Governance* (hereafter referred to as the *Plan*), and are subject to amendment as provided for in the *Plan*.

# ARTICLE 2 MEMBERSHIP

2.1 The members of the Senate are as designated in Article 3 of the *Plan and* further specified in 2.1 and 2.2 below. All elected members are subject to the conditions stated in the *Plan,* including its provisions for expulsion, recall, and impeachment (Articles 4.10, 4.11, and 5.8 of the *Plan* and Article 2.3, 2.4, and 2.5 below).

#### 2.1.a Staff Senators

For the purpose of Senate representation, the Staff Constituency is divided into the following categories. Each category shall elect one Senator from among its ranks for each 200 staff members or major fraction thereof.

- 1. Exempt Staff with appointment in Colleges, Schools, and Academic Affairs
- 2. Exempt Staff with appointment in Divisions
- 3. Non-Exempt Staff with appointment in Colleges, Schools, and Academic Affairs
- 4. Non-Exempt Staff with appointment in Divisions
- 2.1.b Staff member job categories will not include the category designated for the President, vice presidents, provosts, and deans if they hold faculty rank.
- 2.1.c Any individual within the faculty member voting constituency cannot be included in the staff member voting constituency or nominated for election as a staff Senator. Staff candidates for the Senate must have been employed at the University of Maryland College Park for 12 months prior to standing as candidates for the Senate. Staff members may not stand for Senate elections while in the probationary period of employment.
- 2.1.d An ex officio member denoted in the *Plan* (Article 3.6.a.) who is not precluded from staff member categories as noted in Articles 2.1.b and 2.1.c may be elected as a voting member of the Senate by an appropriate constituency. Such ex officio members should also have been employed by the University of Maryland College Park for 12 months prior to standing as candidates for the Senate.
- 2.1.e As noted in the *Plan* (Article 3.3.c), the term of each staff Senator shall be three (3) years. Terms of staff members will be staggered in such a way that for each term, one-third of the total members from a job category are serving the first year of their term. Not every member of a specific staff job category shall be elected in the same year. However, if the University or these *Bylaws* redefine the staff job categories outside of a normal reapportionment, the staff Senate seats will be vacated. A subsequent election will be held to populate all staff Senate seats within the new categories with staggered terms as follows:
  - (1) One-third of the members in a job category who received the lowest number of votes will serve a one-year term.
  - (2) One-third of the members in a job category who received the second lowest number of votes will serve two-year terms,
  - (3) One-third of the members in a job category who received the highest number of votes will serve three year-terms.

A person serving less than a three-year term is defined as not to have served a full term and is eligible for re-election to a full term the following year.

### 2.2 Single Member Constituencies

The Senators defined in (a)-(g) below shall be voting members of the Senate. All elections held pursuant to this section shall be organized by the Office of the University Senate.

- (a) Part-Time Research, Part-Time Teaching, Adjunct, and both Full-Time and Part-Time Visiting Faculty who are not members of the Faculty Constituency as defined in Section 3.2 of the Plan shall together elect one (1) Senator, for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as a Faculty Senator.
  - (b) Emeriti Faculty who are not members of the Faculty Constituency as defined in Section 3.2 of the *Plan* shall elect one (1) Senator from among their ranks for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as a Faculty Senator.
- (c) Head Coaches who are not members of the Faculty Constituency as defined in Section 3.2 of the University Plan of Organization together shall elect one Senator from among their ranks to serve for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as a faculty Senator.
- (d) Post-Doctoral Scholars, Post-Doctoral Associates (formerly Research Associates), Junior Lecturers, and Faculty Assistants (formerly Faculty Research Assistants) who are not members of any Senate constituency as defined in Article 3 of the *Plan* together shall elect one (1) Senator, for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as a Faculty Senator.
- (e) The Contingent II staff shall elect one (1) Senator from among their ranks for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as all other staff Senators. The Contingent II staff Senator shall have been employed by the University for twelve months prior to their election.
- (f) The part-time undergraduate students shall elect one (1) Senator from among their ranks for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as all other student Senators.
- (g) The part-time graduate students shall elect one (1) Senator from among their ranks for a term of one (1) year, renewable for up to three (3) years. When the Senate votes by constituencies, that Senator shall have the same voting rights as all other student Senators.
- 2.3 Elected Senators shall not be absent from two (2) consecutive regularly scheduled meetings of the Senate without notifying the Office of the University Senate that they will require an excused absence (Article 4.10.a of the *Plan*). The Senator shall be counted in the total membership when a quorum is defined for a meeting unless that Senator is expelled.
- 2.4 If an elected Senator is no longer a member of the constituency by which they were elected, the seat may be vacated and the Senator may be replaced according to the following guidelines:
  - 2.4.a If there was a runner-up in the election in which the Senator was elected, the runner-up shall replace that Senator immediately, provided they are still eligible.
  - 2.4.b If there was no runner-up in the election in which the Senator was elected and the vacancy occurs in the spring semester, that Senator shall serve for the remainder of the Senate year and shall be replaced in the next election cycle for the remainder of the term.
  - 2.4.c If there was no runner-up in the election in which the Senator was elected and the vacancy occurs prior to the spring semester, or if the Senator is unable to serve the remainder of the Senate year, the

Senate Executive Committee, in consultation with the appropriate constituency, shall appoint a replacement for that Senator.

- 2.5 If an elected Senator is no longer in satisfactory standing at the University, they shall be replaced immediately upon notification to and verification by the Office of the University Senate in accordance with 2.4.a or 2.4.c above.
- 2.6 All elections shall be completed by the Transition Meeting of the Senate.

# ARTICLE 3 MEETINGS

# 3.1 Regular Meetings:

The Senate shall schedule at least four (4) regular meetings each semester. The notice, agenda, and supporting documents will normally be provided by the Office of the University Senate to the membership one week prior to each regular meeting unless otherwise approved by the Executive Committee.

# 3.2 Special Meetings:

- 3.2.a Special meetings of the Senate may be called in any of the following ways, with the matter(s) to be considered to be specified in the call:
  - (1) By the presiding officer of the Senate;
  - (2) By a majority vote of the Executive Committee of the Senate;
  - (3) By written petition of a majority of the elected members of the Senate. The petition shall be delivered to the Chair or the Executive Secretary and Director of the Senate. The Chair shall give notice of arrangements for the meeting within seventy-two (72) hours of receipt of a valid petition; or
  - (4) By resolution of the Senate.
- 3.2.b The notice of a special meeting shall include the agenda and shall be sent to the members of the Senate as far in advance of the meeting as possible. The agenda of a special meeting may specify a scheduled time of adjournment.
- 3.2.c The scheduling of a special meeting shall reflect the urgency of the matter(s) specified in the call, the requirement of reasonable notice, and the availability of the membership.

### 3.3 Openness of Meetings and Floor Privileges:

- 3.3.a Meetings of the Senate shall be open to all members of the campus community except when the meetings are being conducted in closed session.
- 3.3.b Representatives of the news media shall be admitted to all meetings of the Senate except when the meetings are conducted in closed session. The use of television, video, or recording equipment shall not be permitted except by express consent of the Senate.
- 3.3.c When a report of a committee of the Senate is being considered, members of that committee who are not members of the Senate may have a voice but not a vote in the deliberations of the Senate on that report.
- 3.3.d Any Senator may request the privilege of the floor for any member of the campus community to speak on the subject before the Senate. The Chair shall rule on such requests.
- 3.3.e By vote of the Senate, by ruling of the Chair, or by order of the Executive Committee included in the agenda of the meeting, the Senate shall go into closed session. The ruling of the Chair and the order

- of the Executive Committee shall be subject to appeal, but the Chair shall determine whether such appeal shall be considered in open or closed session.
- 3.3.f While in closed session, the meeting shall be restricted to voting members of the Senate (Article 3 in the *Plan*), members granted a voice but not a vote (Articles 3.6, 5.2.c, and 5.5.c. of the *Plan*), the Executive Secretary and Director, the Parliamentarian, the immediate Past Chair of the Senate, any staff required for meeting operations, and other persons expressly invited by the Senate Chair.

### 3.4 Rules for Procedure:

- 3.4.a The version of *Robert's Rules of Order* that shall govern the conduct of Senate meetings shall be *Robert's Rules of Order*, *Newly Revised*.
- 3.4.b A quorum for meetings shall be defined as a majority of elected Senators who have not submitted an excused absence to the Office of the University Senate, or sixty (60) Senators, whichever number is higher. For the purpose of determining a quorum, ex officio members without vote shall not be considered.
- 3.4.c Voting shall be restricted to eligible members of the Senate (Article 3 in the *Plan*) who are participating in the Senate meeting at the time of the vote.

# ARTICLE 4 EXECUTIVE COMMITTEE

### 4.1 Membership and Election:

- 4.1.a As set forth in the *Plan* (Article 8.2), the members of the Executive Committee shall include the Chair and Chair-Elect of the Senate, thirteen (13) members elected from the voting membership of the Senate, and five (5) non-voting ex officio members.
- 4.1.b The election of the Executive Committee should follow the election of the Chair-Elect as provided for in the *Plan* (Article 5.3 and 5.7.a). In the event of a tie vote in the election for members of the Executive Committee, a ballot will be made available to each Senator in the appropriate constituency as soon as the votes are counted and the tie discovered. The election to break the tie should end one (1) week from the start date.
- 4.1.c In the event of a vacancy on the Executive Committee, the available candidate who had received the next highest number of votes in the annual election for the Executive Committee shall fill the remainder of the unexpired term.

# 4.2 Charge: The Executive Committee shall exercise the following functions:

- 4.2.a Assist in carrying into effect the actions of the Senate;
- 4.2.b Act for the Senate as provided for by and subject to the limitations stated in Article 4.3;
- 4.2.c Act as an initiating body suggesting possible action by the Senate:
- 4.2.d Assist in the administrative implementation of policies adopted by the Senate, as needed;
- 4.2.e Prepare the agenda for each Senate meeting as provided for by and subject to limitations stated in Article 4.4;
- 4.2.f Serve as a channel through which any member of the campus community may introduce matters for consideration by the Senate or its committees;
- 4.2.g Ensure that information on the Senate's work each year is available to the campus community through the Senate website and reported to the President;

- 4.2.h Provide feedback on the operations of the Office of the University Senate, and make recommendations to the President or his or her designee for improvements in those operations and for suitable candidates for the role of the Executive Secretary and Director;
- 4.2.i Serve as the channel through which the Senate and the campus community may participate in the selection of administrators at the University;
  - 4.2.j Perform such other functions as maybe given it in other provisions of these *Bylaws* and the *Plan*; and
- 4.2.k Make recommendations on nominees for campus-wide and system-wide committees and councils requiring representatives, when necessary.

# 4.3 Rules Governing Executive Committee Action for the Senate:

- 4.3.a Where time or the availability of the membership precludes a meeting of the Senate, as, for example, during the summer or between semesters, the Executive Committee may act on behalf of the Senate.
- 4.3.b A report of all actions taken by the Executive Committee when acting on behalf of the Senate, with supporting material, shall be included with the agenda of the next regular meeting of the Senate. By written request of ten (10) Senators, received by the Chair of the Senate prior to the call to order of that meeting, any Executive Committee action on behalf of the Senate shall be vacated and the item in question placed on the agenda for that meeting. If any such item is not petitioned to the floor, it shall stand as an approved action of the Senate.

# 4.4 Rules Governing Preparation of the Senate Agenda:

- 4.4.a The order of business for regular meetings of the Senate shall be set by the Executive Committee. Typically, the order of the agenda will follow *Robert's Rules of Order*, but the Executive Committee maintains authority to reorder the agenda based on the needs of the Senate.
- 4.4.b For regular meetings the Executive Committee shall consider all submissions for inclusion on the Senate agenda. The Executive Committee may not alter a submission, but may delay its inclusion, may include it on the agenda of a special meeting, may submit the material directly to a committee of the Senate, or may refuse to place it on the agenda if the material is inappropriate, incomplete, or unclear. The party making a submission shall be notified of the action taken in this regard by the Executive Committee.
- 4.4.c The order of business for a special meeting of the Senate shall be set by the Executive Committee.
- 4.4.d For a special meeting the agenda shall include the matter(s) specified in the call of that meeting as the Special Order. Other items may be included on the agenda as the Executive Committee deems appropriate.
- 4.5 **Meetings of the Executive Committee**: A quorum of the Executive Committee shall be eight (8) voting members. Minutes of the meetings shall be kept. The agenda shall be made publicly available prior to each meeting. The Executive Committee shall meet at the call of the Chair or by petition of eight (8) voting members of the Executive Committee, or by petition of twenty-five (25) voting members of the Senate.
- 4.6 **The Senate Budget**: The Executive Secretary and Director shall be responsible for the Senate budget, shall consult with the Senate Chair on the preparation of the budget request, and shall report to the Executive Committee on the status of the budget.
  - 4.6.a The Executive Secretary and Director shall make an annual report to the Associate Vice President for Personnel and Budget on expenditure of the Senate budget.
  - 4.6.b Consent of the Associate Vice President for Personnel and Budget shall be required before any change in the budgeted use of Senate funds involving more than ten percent (10%) of the total may be undertaken.

- 4.7 Referral of Items to Standing Committees: The Executive Committee shall refer items to the standing committees.
  - 4.7.a The Executive Committee shall refer an item to an appropriate committee when instructed by the Senate or when requested by the President, or when petitioned by 150 members of the Senate electorate.
  - 4.7.b The Executive Committee may also refer any item it deems appropriate, and the standing committee shall give due consideration to such requests from the Executive Committee.
  - 4.7.c The Chair of the Senate may, as need requires, act for the Executive Committee and refer items to standing committees. All such actions shall be reported at the next meeting of the Executive Committee.
- 4.8 To the extent permitted by law and University policy, the records of the Senate shall be open.

# ARTICLE 5 COMMITTEES OF THE SENATE

- 5.1 **Standing Committees Specifications:** The specifications of each standing committee of the Senate shall state its name, its specific charge, and any exceptions or additions to the basic charge to standing committees stated in Article 5.2. The specifications shall list all voting ex officio members and shall define committee composition.
  - 5.1.a Standing Committees: In an appropriate section of Article 6 there shall be specifications for each committee.
- 5.2 **Standing Committees Basic Charge:** In its area of responsibility, as defined in its specifications, each committee shall be an arm of the Senate with the following powers:
  - (1) To formulate and review policies to be established by the Senate according to the *Plan* (Article 1);
  - (2) To review established policies and their administration and to recommend any changes in policies or their administration that may be desirable;
  - (3) To serve in an advisory capacity, upon request, regarding the administration of policies;
  - (4) To function on request of the President or of the Executive Committee as a board of appeal with reference to actions and/or decisions made in the application of policies; and
  - (5) To recommend the creation of special subcommittees (Article 5.7-5.9) when deemed necessary.

# 5.3 Standing Committees - Committee Operation:

- 5.3.a Agenda Determination:
  - (1) Issues within the committee's purview shall be referred and charged to the committee by the Executive Committee (Article 4.7). Such matters should take priority on the committee's agenda.
  - (2) A committee shall have principal responsibility for identifying matters of present and potential concern to the campus community within its area of responsibility. Such matters should be placed on the agenda of the committee.
  - (3) Committee agendas shall be made publicly available prior to each meeting.
- 5.3.b Minutes of the proceedings of each committee meeting shall be kept.

- 5.3.c Rules for Procedure of Standing Committees: Standing Committees are typically governed by *Robert's Rules of Order for Small Committees, Newly Revised.* Standing Committees shall determine how technology, such as video conferencing and other electronic methods of participation, can be used for their purposes. Standing Committees may choose to conduct votes online or via email, and shall agree on any other mechanisms for conducting business outside of meetings, when necessary.
- 5.3.d Quorum Requirements of Standing Committees: Unless a quorum number is specified in the membership description of a committee, the quorum shall be a majority of voting members of the committee.
- 5.4 **Standing Committees Reporting Responsibilities:** Each committee shall be responsible through its presiding officer for the timely delivery of the following reports.
  - 5.4.a Information on the schedule of committee meetings stating the date, time, and location should be made available on the Senate. The agenda for each meeting shall be made available on the Senate website as far in advance of the meeting as possible.
  - 5.4.b The committee shall report its progress on agenda items as required by the Executive Secretary and Director or the Chair of the Senate.
  - 5.4.c Reports providing information and/or recommendations to the Senate shall be submitted to the Executive Committee for inclusion on the Senate agenda. Reports resulting from the committee's advisory or board of appeals function shall be submitted to the appropriate Senate or campus officer, and the Executive Committee notified of the submission.
  - 5.4.d Upon written request of at least four (4) members of a committee, the presiding officer of that committee shall include a minority statement with any committee report. Those requesting inclusion need not support the substance of the minority statement.
  - 5.4.e An annual report shall be presented to the Chair of the Senate at the end of the academic year, or, if approved by the Chair, no later than August 16, for submission to the Executive Committee. The report shall include a list of all items placed on the committee's agenda, noting the disposition of each and a summary of the committee's deliberations. An overview of the committee's past work shall be made available on the Senate website. In the case of committees with little activity, the committee may recommend inactive status the ensuing year until charged by the Executive Committee to address a specific matter:
    - (1) A committee maybe placed on inactive status with approval of the Executive Committee. No presiding officer or members shall be appointed to the committee while on inactive status.
    - (2) A committee on inactive status may be reactivated by the Executive Committee when matters within its purview, as stipulated in Article 6, are brought to the Executive Committee for review. Following reactivation, the Office of the University Senate shall solicit volunteers for the committee in its annual volunteer period, and the Committee on Committees shall select members for the committee, in accordance with the provisions of 5.5 below.
    - (3) A Special Committee (Article 5.9) may not be appointed to consider matters within standing committee specifications in lieu of reactivating an inactive committee.
- 5.5 **Standing Committees Selecting Members:** Persons shall be named to standing committees in accordance with the procedures listed below.
  - 5.5.a The Office of the University Senate shall provide information on the charge and membership specifications of each committee.
  - 5.5.b The Office of the University Senate shall solicit volunteers for the Senate's standing committees on an annual basis through an online process. During this volunteer period, all faculty, staff, and students shall be eligible to indicate their top three preferences for any committees with vacancies in their

- constituency and include a candidacy statement for consideration by the Committee on Committees. The Office of the University Senate will maintain these records for potential future use.
- 5.5.c The Committee on Committees shall develop slates of nominees to fill vacancies on the standing committees and University Councils. No person shall be nominated for a committee position without consenting to serve on that committee, either through indicated preference or explicit agreement. In making nominations, the Committee on Committees shall keep in view the continuing membership of the committee to ensure that the full membership complies with specifications of the *Plan* and these *Bylaws*. Committee members shall be nominated consistent with requirements for diversity specified in Section 8.1 of the *Plan*.
- 5.5.d Ex officio members named in a committee's specifications shall be voting members unless otherwise specified in the *Bylaws*. Upon recommendation of the Committee on Committees, the Executive Committee may appoint ex officio members with particular expertise or benefit to the committee. Such members shall serve with voice, but without vote. The Executive Committee is empowered to make such changes in non-voting ex officio membership as appropriate.
- 5.5.e The Committee on Committees shall forward a slate of nominees for committee service to the Executive Committee to place on the Senate agenda for approval. Each nominee shall be identified by name and constituency. The notice of nomination shall also include the name and constituency of continuing members of the committee, and the name and office of the ex officio members, listed for information only. The nominations shall be subject to action by the Senate consistent with the *Plan* and the specifications of these *Bylaws*.
- 5.5.f Terms on standing committees shall be two (2) years for faculty and staff, and one (1) year for students. Appointments to two-year terms shall be staggered: that is, as far as practical, half of the terms from each faculty or staff constituency shall expire each year. Terms shall begin on July 1 of the appropriate year.
- 5.5.g A member of a standing committee whose term is expiring may be appointed to another term, subject to restrictions (1) and (2) below. The Committee on Committees is particularly charged to consider the reappointment of active student members.
  - (1) No reappointment shall be made that would cause the appointee to serve longer than four consecutive years on the same committee.
  - (2) At most, half of the non-student members of a committee whose terms are expiring in any given year may be reappointed.
- 5.5.h Terms as presiding officer of a committee shall be one year. A presiding officer may be reappointed if his/her tenure as a Senator is continuing; however, no one shall serve as presiding officer of a committee for longer than two (2) consecutive years.
- 5.5.i Appointments of the presiding officers of committees shall be made by the Chair of the Senate, designated on the annual committee slate, and shall be approved by the Senate.
- 5.6 **Standing Committees Replacing Presiding Officers and Members:** The presiding officer and members of any active standing committee may be replaced for cause after inquiry by the Office of the University Senate with approval of the Executive Committee.
  - 5.6.a Cause, for presiding officers, is defined as the following:
    - (1) Failure to activate the committee during the first semester after appointment in order to organize its business and determine an agenda; or
    - (2) Failure to activate the committee in order to respond to communications referred from the Executive Committee; or

- (3) Failure to activate the committee in order to carry out specific charges required in Article 6 or other Senate documents; or
- (4) Continual absence from scheduled committee meetings.
- 5.6.b Cause, for members, is defined as the following:
  - Continual absence from committee meetings and/or lack of participation in committee activities;
     or
  - (2) Lack of registration on campus for students or termination of employment on campus for faculty and staff.
- 5.6.c Procedure for replacing presiding officers and members:
  - (1) The decision to replace a presiding officer rests with the Senate Chair; and
  - (2) The presiding officer of a committee shall submit the request to replace a committee member to the Chair of the Committee on Committees.
- 5.6.d The Senate Chair and the Chair of the Committee on Committees shall consult with the Office of the University Senate to identify a replacement when a decision is made to replace a presiding officer or a committee member.
- 5.7 **Standing Committees Appointing Special Subcommittees:** A standing committee of the Senate may appoint special subcommittees to assist in the effective performance of its responsibilities. Persons appointed to special subcommittees who are not members of standing committees must be approved by the Executive Committee. The Chair of any special subcommittee must be a member of the standing committee making the appointment.
- Standing Committees Appointing Special Joint Subcommittees: Two or more standing committees of the Senate may appoint special joint subcommittees to assist in the effective review of issues that pertain to the charge of multiple committees. Persons appointed to serve who are not members of associated standing committees must be approved by the Executive Committee. The Chair of any such subcommittee must be a member of one of the associated standing committees making the appointment. Special Joint Subcommittees will report directly to the full associated standing committees for final action.
- 5.9 **Special Committees:** A special committee of the Senate may be established by resolution of the Senate to carry out a specified task. The empowering resolution shall also stipulate the means of selecting the committee and any restrictions on its composition. The committee shall function until the completion of its tasks or until discharged by the Senate. A final report of its work shall be presented to the Senate. Members shall serve for the duration of the committee unless otherwise specified by the Senate.

### ARTICLE 6 STANDING COMMITTEE SPECIFICATIONS

#### 6.1 Academic Procedures and Standards Committee:

- 6.1.a Membership: The committee shall consist of an appointed presiding officer; ten (10) faculty members; one (1) staff member; three (3) undergraduate and one (1) graduate student; and the following persons or a representative of each: the Senior Vice President and Provost, the Director of Undergraduate Admissions, the University Registrar, the Associate Provost for Academic Affairs and Dean for Undergraduate Studies, and the Associate Provost for Academic Affairs and Dean of the Graduate School.
- 6.1.b Quorum: A quorum of the Academic Procedures and Standards Committee shall be ten (10) voting members.

- 6.1.c Charge: The committee shall formulate and continually review policies, rules, and regulations governing the admission, readmission, academic standing, and dismissal of all students for academic deficiency.
- 6.1.d Charge: The committee shall formulate and continually review policies and procedures for academic advisement, scheduling of classes, and registration.
- 6.1.e Charge: The committee shall formulate and continually review policies to be observed by the instructional staff in conducting classes, seminars, examinations, students' research, and student evaluations.
- 6.1.f Policies, rules, and regulations exclusively governing admission, readmission, scholastic standing, and dismissal of graduate students for academic deficiency shall be reviewed by an appropriate committee of the Graduate School. Such policies, rules, and regulations will be transmitted by the Graduate School directly to the Senate through the Executive Committee. Policies, rules, and regulations that concern both graduate and undergraduate matters shall be considered by both the Academic Procedures & Standards (APAS) Committee and the appropriate committee of the Graduate School.

### 6.2 Campus Affairs Committee:

#### 6.2.a Membership:

- (1) The committee shall consist of an appointed presiding officer; six (6) faculty members; two (2) undergraduate and two (2) graduate students; two (2) staff members, with one exempt and one non-exempt to the extent of availability; the President or a representative of the Student Government Association; the President or a representative of the Graduate Student Government; and the following persons or a representative of each: the Senior Vice President and Provost, the Vice President and Chief Administrative Officer, the Vice President for Student Affairs, the Vice President for Marketing and Communications, the Vice President for Diversity & Inclusion, and the Chair of the Coaches Council.
- (2) When discussions of safety are on the agenda, the Chief of Police, the Office of General Counsel, the Director of Transportation Services, and other campus constituencies, as appropriate, shall be invited to participate or send a representative.
- (3) The Chair of this committee or a faculty member designated by the Chair and approved by the Senate Executive Committee will serve as an ex officio member of the Athletic Council. The Chair, or a committee member designated by the Chair, shall also serve as an ex-officio member of the Campus Transportation Advisory Committee.
- 6.2.b Quorum: A quorum of the Campus Affairs Committee shall be nine (9) voting members.
- 6.2.c Charge: The committee shall formulate and continually review policies and regulations affecting the entire campus, its functions, its facilities, its internal operation and its external relationships, including the awarding of campus prizes and honors, and make recommendations concerning the future of the campus.
- 6.2.d Charge: The committee shall formulate and continually review policies and procedures for the periodic review of campus level administrators.
- 6.2.e Charge: The committee shall periodically gather community input on safety and security issues and shall act as a liaison between the police and the campus community.

### 6.3 Committee on Committees:

- 6.3.a Membership and terms:
  - (1) As set forth in the *Plan* (Article 8.3.a), the Committee on Committees shall be chaired by the Chair-Elect of the Senate.

- (2) The voting membership, as defined in the Plan (Article 8.3.a), shall consist of the Chair-Elect of the Senate, six (6) faculty members elected by faculty Senators, with no more than one (1) from any College or School; one (1) non-exempt staff member elected by non-exempt staff Senators; one (1) exempt staff member elected by exempt staff Senators; one (1) undergraduate student elected by undergraduate student Senators; and one (1) graduate student elected by graduate student Senators.
- (3) Students are elected to serve for one (1) year, faculty and staff for two (2) years, whether or not their membership in the Senate continues beyond their first year of service in the committee.
- (4) Terms of faculty and staff members are staggered in such a way that, at any time, no more than three (3) faculty members and one (1) staff member are serving the second year of their term.
- (5) In the event of a vacancy on the Committee on Committees, the available candidate who had received the next highest number of votes in the last annual election for the Committee on Committees shall fill the remainder of the unexpired term. In the event that there is no runner-up, the Executive Committee shall fill the vacant seat.
- (6) A quorum of the Committee on Committees shall be six (6) voting members.

### 6.3.b Charge:

- (1) As set forth in the *Plan* (Article 8.3.b), responsibilities of the Committee on Committees include:
  - (a) Identification and recruitment of individuals for service on Senate committees;
  - (b) Approval of the University Library Council slate of nominees, as mandated in section 2.C of the Bylaws of the University Library Council.
  - (c) Creation of a slate of nominees for the Nominations Committee, for approval by the Senate.
- (2) Additional duties include:
  - (a) As needed, the Committee on Committees may be charged to assess effectiveness of committees, and make recommendations for improvements and changes in their operations and structure.
  - (b) Other such duties as specified by the Executive Committee.
- 6.3.c Operation: The Committee on Committees shall follow the procedures specified for standing committees in Article 5 above, with the exceptions of 5.3.b and 5.5.

#### 6.4 Educational Affairs Committee:

- 6.4.a Membership: The committee shall consist of an appointed presiding officer; ten (10) faculty members, of whom at least two (2) must be tenured/tenure-track faculty members and at least two (2) must be professional track faculty members; two (2) staff members, with one exempt and one non-exempt to the extent of availability; two (2) undergraduate students and one (1) graduate student; the President or a representative of the Student Government Association; the President or a representative of the Graduate Student Government; the Associate Dean for General Education; a representative of the Associate Provost for Academic Affairs and Dean for Undergraduate Studies; and the following persons or a representative of each: the Senior Vice President and Provost, the Associate Provost for Academic Affairs and Dean of the Graduate School, and the Vice President of Information Technology and Chief Information Officer (CIO).
- 6.4.b Quorum: A quorum of the Educational Affairs Committee shall be eleven (11) voting members.
- 6.4.c Charge: The committee shall formulate and continually review plans and policies to strengthen the

educational system of the College Park campus. The committee shall receive ideas, recommendations, and plans for educational innovations from members of the campus community and others. The committee shall inform itself of conditions in the Colleges, Schools, and other academic units, and shall propose measures to make effective use of the resources of the campus for educational purposes.

- 6.4.d Charge: The committee shall exercise broad oversight and supervision of the General Education Program at the University of Maryland as described in the 2010 document Transforming General Education at the University of Maryland and the General Education Implementation Plan approved by the University Senate in February 2011. The committee shall review and make recommendations concerning the General Education Program to the Senate and the Associate Provost for Academic Affairs and Dean for Undergraduate Studies. Such recommendations shall include, as the committee deems appropriate, the program's requirements and its vision, especially with regard to evaluating trends, reviewing learning outcomes, and maintaining the balance of courses in the General Education categories.
- 6.4.e Relation of the Educational Affairs Committee to the General Education Program and the Office of the Associate Provost for Academic Affairs and Dean for Undergraduate Studies:
  - (1) The Associate Provost for Academic Affairs and Dean for Undergraduate Studies willprepare an annual report on the status of the General Education Program and will send the report to the Educational Affairs Committee by October 1.
  - (2) The Associate Provost for Academic Affairs and Dean for Undergraduate Studies will meet with the Educational Affairs Committee as needed to discuss or update the report. Topics will include but not be limited to: the membership and ongoing work of the General Education Faculty Boards; the proposal and approval process for General Education courses; the learning outcomes for the different course categories; areas where additional courses or rebalancing may be needed; trends and developments that may impact the General Education Program; and informational resources for students, faculty, and advisors about the General Education Program.
  - (3) The Office of the Associate Provost for Academic Affairs and Dean for Undergraduate Studies shall inform the committee of modifications in the proposal or review process, the disposition of recommendations from the committee, and any other changes regarding the implementation of the General Education Program as specifically delegated to that office.

#### 6.5 Elections, Representation, and Governance Committee:

- 6.5.a Membership: The committee shall consist of an appointed presiding officer; six (6) faculty members; one (1) exempt staff member; one (1) non-exempt staff member; two (2) undergraduate and two (2) graduate students; and representatives of the Director of Human Resources and the Associate Vice President for Institutional Research, Planning, and Assessment.
- 6.5.b Quorum: A quorum of the Elections, Representation, and Governance Committee shall be eight (8) voting members.
- 6.5.c Charge: The committee shall review and recommend policies regarding the conduct of elections, determine correct apportionments for all constituencies, and investigate and adjudicate all charges arising from the management and results of Senate elections.
- 6.5.d Charge: The committee shall determine the correct apportionment for all constituencies every five (5) years as stipulated in Article 3.8 of the Plan and following any review or revision of the *Plan* as stipulated in Article 6.3 of the *Plan*.
- 6.5.e Charge: The committee shall supervise all Senatorial elections and referenda in accordance with the *Plan* (Article 4.2), and shall consult with certain constituencies in their nomination and election processes in accordance with the *Plan* (Article 4) as requested by the Executive Committee.

- 6.5.f Charge: The committee shall formulate and review procedures for the tallying and reporting of election results and shall perform other such duties as appropriate (Article 3.3.b of the *Plan*).
- 6.5.g Charge: The committee shall review the Plans of Organization of the Colleges, Schools, and other units, in accordance with the *Plan* (Article 11) and as specified in Appendix 7 of these Bylaws.
- 6.5.h Charge: The committee shall review and observe the operation and effectiveness of the University Senate and make any appropriate recommendations for improvements.
- 6.5.i Charge: The committee shall receive all petitions for impeachment of the Chair or Chair-Elect in accordance with the *Plan* (Article 5.8).
- 6.5.j Charge: The committee shall initiate procedures for expelling Senators in accordance with the *Plan* (Article 4.10).
- 6.5.k Charge: The committee shall receive all petitions for the recall of Senators in accordance with the *Plan* (Article 4.11).

#### 6.6 Equity, Diversity, & Inclusion Committee:

- 6.6.a Membership: The committee shall consist of an appointed presiding officer; five (5) faculty members; three (3) exempt staff members; two (2) non-exempt staff members; two (2) undergraduate and two (2) graduate students; and the following persons or a representative of each: the Senior Vice President and Provost, the Vice President for Diversity & Inclusion, the Vice President and Chief Administrative Officer, the Vice President for Student Affairs, and the Director of the Office of Civil Rights and Sexual Misconduct.
- 6.6.b Quorum: A quorum of the Equity, Diversity, & Inclusion Committee shall be ten (10) voting members.
- 6.6.c Charge: The committee shall actively promote an equitable, diverse, and inclusive campus that is free from all forms of discrimination by formulating and continually reviewing policies and procedures pertaining to issues of equity, diversity, and inclusion. These include but are not limited to the University of Maryland Non-Discrimination Policy and Procedures and the University of Maryland Disability & Accessibility Policy and Procedures.
- 6.6.d Charge: The committee shall consider programs and activities for improving equity, diversity, and inclusiveness on campus, and shall make recommendations to appropriate campus bodies.

#### 6.7 Faculty Affairs Committee:

- 6.7.a Membership: The committee shall consist of an appointed presiding officer; ten (10) faculty members, of whom four (4) shall be Senators including one (1) assistant professor and one (1) professional track faculty member; one (1) undergraduate student and two (2) graduate students; one (1) staff member; and the following persons or a representative of each: the President, the Senior Vice President and Provost, and the Director of Human Resources. One (1) elected Council of University System Faculty representative from the University shall serve as a voting ex officio member. The Faculty Ombuds Officer shall serve as a non-voting ex officio member.
- 6.7.b Quorum: A quorum of the Faculty Affairs Committee shall be nine (9) voting members.
- 6.7.c Charge: The committee shall formulate and continually review policies pertaining to faculty life, employment, academic freedom, morale, and perguisites.
- 6.7.d Charge: The committee shall work for the advancement of academic freedom and the protection of faculty and research interests.
- 6.7.e Charge: The committee shall, in consultation with Colleges, Schools, and other academic units,

formulate and review procedures for the periodic review of academic administrators below the campus level.

6.7.f Charge: The committee shall review the Appointment, Promotion, and Tenure or Permanent Status section of each College, School, or the Library Plan of Organization in accordance with Appendix 7 of these *Bylaws*. In conjunction with this review, the committee shall also review the professional track faculty Appointment, Evaluation, and Promotion Policy of each College, School, or the Library.

#### 6.8 Programs, Curricula, and Courses Committee:

- 6.8.a Membership: The committee shall consist of an appointed presiding officer; ten (10) faculty members; one (1) staff member; two (2) undergraduate students and one (1) graduate student; and the following persons or a representative of each: the Senior Vice President and Provost, the Associate Provost for Academic Affairs and Dean for Undergraduate Studies, the Associate Provost for Academic Affairs and Dean of the Graduate School, and the Dean of Libraries.
- 6.8.b Quorum: A quorum of the Programs, Curricula, and Courses Committee shall be nine (9) voting members.
- 6.8.c Charge: The committee shall formulate, review, and make recommendations to the Senate concerning policies related both (1) to the establishment, modification, or discontinuance of academic programs, curricula, and courses; and (2) to the establishment, reorganization, or abolition of colleges, schools, academic departments, or other units that offer credit-bearing programs of instruction or regularly offer courses for credit.
- 6.8.d Charge: The committee shall review and make recommendations to the Senate in at least the areas designated by (1) through (3) below. Recommendations in these areas are not subject to amendment on the Senate floor unless a detailed objection describing the area of concern has been filed with the Office of the University Senate at least forty-eight (48) hours prior to the meeting at which the recommendations will be introduced. The committee will announce proposed recommendations to the campus community sufficiently in advance of the meeting at which they are to be considered so as to allow time for concerned parties to file their objections.
  - (1) All proposals for the establishment of a new academic program, for the discontinuance of an existing academic program, for the merger or splitting of existing academic programs, or for the renaming of an existing academic program;
  - (2) All proposals for the creation, abolition, merger, splitting, or change of name of Colleges, Schools, departments of instruction, or other units that offer credit-bearing programs of instruction or regularly offer courses for credit; and
  - (3) All proposals to reassign existing units or programs to other units or programs.
- 6.8.e Charge: The committee shall review and shall directly advise the Office of Academic Planning and Programs concerning proposals to modify the curricula of existing academic programs, or to establish citation programs consistent with College rules approved by the Senate. The committee shall inform the Senate of its actions in these cases.
- 6.8.f Charge: The committee shall review, establish, and advise the Vice President's Advisory Committee concerning policies for adding, deleting, or modifying academic courses.
- 6.8.g Charge: The committee shall be especially concerned with the thoroughness and soundness of all proposals, and shall evaluate each according to the mission of the University, the justification for the proposed action, the availability of resources, the appropriateness of the sponsoring group, and the proposal's conformity with existing regulations. The committee shall be informed of any recommendations made by the Academic Planning Advisory Committee concerning resource issues, the consistency of the proposed action with the University's mission and strategic directions, or both.

- 6.8.h Operation: The committee shall follow the procedures specified for standing committees in Article 5 above, with the exception of 5.3.b.
- 6.8.i Relation of the Programs, Curricula, and Courses Committee to the Office of the Senior VicePresident and Provost.
  - (1) The committee, in consultation with the Office of the Senior Vice President and Provost, shall determine the requirements for supporting documentation and the procedures for review for all proposals.
  - (2) The committee shall be informed by the Office of the Senior Vice President and Provost of all proposed modifications to existing programs and curricula. After consulting with the presiding officer of the committee, the Office of the Senior Vice President and Provost shall act on all minor changes that are not of a policy nature.
  - (3) The committee shall be informed by the Office of the Senior Vice President and Provost of all changes made pursuant to 6.8.i(2). The committee shall be informed by the Office of the Senior Vice President and Provost of all other changes in academic curricula whose approval has been specifically delegated to that office. In particular, this includes the approval to offer existing academic programs through distance education or at a new off-campus location.
- 6.8.j Relationship of the Programs, Curricula, and Courses Committee to the Graduate School: Proposals concerned with graduate programs and curricula shall receive the review specified by the Graduate School, in addition to the review of the Programs, Curricula, and Courses Committee. Any such proposal whose approval has been denied by the Graduate School shall not be considered by the committee.

#### 6.9 Staff Affairs Committee:

- 6.9.a Membership: The committee shall consist of an appointed presiding officer; eight (8) staff members, with two (2) members from each of the elected staff categories; two (2) Category II contingent employees, with one exempt and one non-exempt to the extent of availability; one (1) faculty member; one (1) student; and one (1) representative each of the Senior Vice President and Provost, the Director of Human Resources, the Vice President and Chief Administrative Officer and the Vice President for Student Affairs. The three (3) elected University representatives to the Council of University System Staff (CUSS) shall serve as voting ex officio members; the alternate University representatives to the Council of University System Staff (CUSS) shall be non-voting ex officio members.
- 6.9.b Quorum: A quorum of the Staff Affairs Committee shall be nine (9) voting members.
- 6.9.c Charge: The committee shall formulate and continually review campus policies affecting staff members, including policies regarding periodic review of campus departments and administrators that employ staff members.
- 6.9.d Charge: The committee shall assist the Office of the University Senate in soliciting nominations and encouraging participation in elections of staff Senators as specified in Article 4.5 of the *Plan*.
- 6.9.e Charge: Staff Affairs shall assist the Committee on Committees and the Senate Executive Committee in identifying and recruiting staff representatives for campus and Senate committees, including system-wide activities involving staff.
- 6.9.f Charge: The committee shall administer the Council of University System Staff (CUSS) nomination and election process. Definitions of eligible staff shall be defined by the Board of Regents and CUSS.
- 6.9.g Charge: The committee shall actively promote and provide orientation and opportunities for staff involvement in shared governance at every administrative level.
- 6.9.h Charge: The committee shall facilitate the annual nomination process for the Board of Regents' Staff

#### 6.10 Student Affairs Committee:

- 6.10.a Membership: The committee shall consist of an appointed presiding officer; eight (8) undergraduate students, of whom four (4) must be Senators; four (4) graduate students, of whom two (2) must be Senators; two (2) faculty members; two (2) staff members with one exempt and one non-exempt to the extent of availability; the President or a representative of the Student Government Association; the President or a representative of the Graduate Student Government; two (2) representatives of the Office of the Vice President for Student Affairs; and one (1) representative each from the Graduate School, and the Department of Resident Life.
- 6.10.b Quorum: A quorum of the Student Affairs Committee shall be ten (10) voting members.
- 6.10.c Charge: The committee shall formulate and continually review policies regarding all non-academic matters of student life including, but not limited to, student organizations, resident life, extracurricular activities, and student concerns in the campus community.
- 6.10.d Charge: The committee shall support the work of other Senate committees by assessing and communicating the student perspective on a range of issues affecting students, including matters outside the purview described in 6.10.c.
- 6.10.e Charge: The committee shall assist the Office of the University Senate and the Colleges and Schools as appropriate in soliciting nominations and encouraging participation in the election of student Senators.

#### 6.11 Student Conduct Committee:

- 6.11.a Membership: The committee shall consist of an appointed presiding officer; four (4) faculty members; one (1) staff member; five (5) students, of whom at least three (3) must be undergraduate students and one (1) must be a graduate student; and the Director of the Office of Student Conduct, or a representative, as a non-voting ex officio member.
- 6.11.b Charge: The committee shall formulate and continually review recommendations concerning the rules and codes of student conduct, as well as means of enforcing those rules and codes.
- 6.11.c Charge: The committee acts as an appellate body for infractions of the approved Code of Student Conduct and Code of Academic Integrity. Procedures for the committee's operation in this role are to be developed and filed with the Office of Student Conduct and the Executive Secretary and Director of the Senate. The committee shall also confirm members of all judicial boards listed in the Code of Student Conduct, except conference and ad hoc boards.

# ARTICLE 7 SPECIAL COMMITTEE ON UNIVERSITY FINANCE

#### 7.1 Membership and Selection:

- 7.1.a Composition: The special committee shall consist of a presiding officer appointed by the Senate Chair from among the tenured faculty; five (5) tenured or tenure-track faculty members; one (1) professional track faculty member; one (1) exempt staff member; one (1) non-exempt staff member; two (2) undergraduate students; one (1) graduate student; the immediate Past Chair of the Senate; the Vice President for Finance and Chief Financial Officer; the Associate Vice President for Finance and Personnel, Academic Affairs; and the following persons or a representative of each: the President, and the Vice President for Student Affairs. The Senior Vice President and Provost shall also appoint a representative chosen from among current and former unit-level budget officers or former department chairs. All members of the special committee shall be voting members.
- 7.1.b Selection of Members: The regular membership of the special committee shall be selected by the elected members of the Senate Executive Committee. Following the May 7, 2019, Transition Meeting,

current Senators may nominate any member of the campus community. Nominees shall provide a statement indicating their interest in and qualifications for the special committee. Members of the Senate Executive Committee may not be nominated. Elected members of the Senate Executive Committee will vote by constituencies for members of the special committee. In the event of a tie, the Senate Chair will cast the deciding vote.

- 7.1.c Membership—Vacancies: After each Transition Meeting of the Senate, current Senators may nominate members of the campus community for any vacant seats. In the event of a vacancy during the academic year, members of the Senate Executive Committee from the respective constituency will select a replacement from the most recent list of nominees. If there are no interested nominees, a new nomination period will be opened and members of the Senate may submit nominations following the procedures in 7.1.b.
- 7.1.d Membership—Terms: Terms shall be four (4) years for faculty and staff, and one (1) year for students. Student members who wish to continue may be renewed up to two times. Terms shall begin on July 1, 2019.
- 7.2 Charge: The special committee shall exercise the following functions:
  - 7.2.a Develop a deep understanding of the University's budget and budgeting processes and use that knowledge to educate the campus community on these practices.
  - 7.2.b Consult with and advise the President, the Senior Vice President and Provost, and other University administrators on short- and long-term institutional priorities, particularly as they relate to the University's mission and Strategic Plan.
  - 7.2.c Advise Senate-related bodies—including committees, councils, and task forces—on the fiscal implications of any proposed recommendations under consideration.
  - 7.2.d Report to the Senate two times each year on the budgetary and fiscal condition of the University and the administration's response to any special committee recommendations.
  - 7.2.e Regularly report on its activities and the budgetary and fiscal condition of the University to the Senate Executive Committee.

#### 7.3 Operations:

- 7.3.a Agenda Determination: The special committee shall have principal responsibility for identifying matters of present and potential concern to the campus community within its area of responsibility. The presiding officer shall place such matters on the agenda of the committee. Agendas shall be made publicly available prior to each meeting.
- 7.3.b Meetings: The special committee shall meet as frequently as is needed to accomplish its charge, but at least monthly throughout the academic year. Additional meetings may be required over the summer months to accommodate the University's budgeting processes. Given the sensitive nature of the special committee's work, meetings will be closed to all but members and invited guests.
- 7.3.c Minutes: Minutes of the special committee's proceedings shall be kept.
- 7.3.d Procedure: The version of *Robert's Rules of Order* that shall govern the special committee shall be *Robert's Rules of Order for Small Committees, Newly Revised*. The special committee shall determine how technology, such as phone and video conferencing and other electronic methods of participation, can be used for its purposes. The special committee may choose to conduct votes via email, and shall agree on any other mechanisms for conducting business outside of meetings, when necessary.
- 7.3.e Quorum: Quorum shall be a majority of the members of the special committee.
- 7.3.f Guests: The special committee may invite guests to participate in its meetings if it is deemed necessary.

#### 7.4 Dissolution:

7.4.a The special committee shall be dissolved following the adjournment of the last regular Senate meeting of the 2022-2023 academic year, at which time the provisions in this article will become inoperative.

### ARTICLE 8 UNIVERSITY COUNCILS

- 8.1 **Definition:** University Councils are established by Article 8.6 of the *Plan* to exercise an integrated advisory role over specified campus units and their associated activities. University Councils are jointly sponsored by the University Senate and the Office of the President or Provost (as appropriate). University Councils may be assigned reporting responsibilities to any member(s) of the College Park administration at the dean level or above (hereafter referred to as the "designated administrative officer").
- 8.2 Creation of University Councils: Proposals to create a University Council shall be evaluated by a task force appointed jointly by the Senate Executive Committee and the designated administrative officer to whom the new Council would report. Following its deliberations, this task force shall present a report (hereafter referred to as the "Task Force Report") to the Senate, the designated administrative officer, and the director of the unit whose activities are the focus of the Council. The Task Force Report shall indicate the specifications that define the working relationship among the Senate, the designated administrative officer, and the director. The Task Force Report shall include at least the following: the scope and purpose of the new Council; a review of the current committees and advisory relationships to be superseded by the proposed Council; identification of the designated administrative officer and unit director to whom the Council reports; the charge to the Council; the size, composition, and appointment process of members of the Council; the Council's relationship to the Senate, the designated administrative officer, and the director including the responsibilities of these three sponsors to the Council and the responsibilities of the Council to these three sponsors; and principles for operation of the Council. The Task Force Report shall be reviewed by the Executive Committee, approved by the designated administrative officer, and then approved by the Senate. At the same time, the Senate shall approve appropriate revisions in its Bylaws to incorporate the Council into its council structure as defined in Article 8 of these Bylaws. The Task Force Report, as approved, shall be preserved with official Senate documents, serving as a record of the original agreements establishing the Council.
- 8.3 **Specifications in Senate Bylaws:** For each Council, Senate Bylaws shall: state its name; specify its responsibilities to the Senate; define its membership, including any voting privileges of ex officio members; and identify any exceptions or additions to the provisions of this Article particular to the Council.

#### 8.4 Basic Charge:

- 8.4.a The Council's responsibilities to the University Senate shall include those specified for Senate committees in Article 5.2 of these *Bylaws*. In addition, each Council shall:
  - (1) Sponsor hearings, as appropriate, on issues within its purview that are of concern to the Senate and the campus community.
  - (2) Provide a mechanism for communication with the campus community on major issues facing the unit and its activities.
  - (3) Respond to charges sent to the Council by the Senate Executive Committee in accordance with Article 4.7.
  - (4) Provide an annual written report to the Senate on the Council's activities including the status of unresolved issues.
- 8.4.b Responsibilities to the designated administrative officer shall be specified in the Task Force Report and may include:
  - (1) To advise on the unit's budget, space, and other material resources, in addition to personnel,

- staffing and other human resources.
- (2) To advise on the unit's administrative policies and practices.
- (3) To advise on the charges to be given to periodic internal and external review committees.
- (4) To respond to requests for review, analysis, and advice from the designated administrative officer.
- (5) To meet at least annually with the designated administrative officer to review the major issues facing the unit and its activities on campus.
- (6) To fulfill such other responsibilities as specified in the Task Force Report.
- 8.4.c Responsibilities to the unit's director shall be specified in the Task Force Report and may include:
  - (1) To advise on the needs and concerns of the campus community.
    - (2) To advise on opportunities, policies, and practices related to the unit's ongoing operations.
  - (3) To review and advise on unit reports, studies, and proposed initiatives.
  - (4) To respond to requests for review, analysis, and advice made by the director.
  - (5) To meet at least annually with the director to review the major issues facing the unit and its activities on campus.
  - (6) To fulfill such other responsibilities as specified in the Task Force Report.

#### 8.5 Membership and Appointment to University Councils:

- 8.5.a Membership: Councils shall have nine (9) to thirteen (13) members as specified in the appropriate subsection of Article 9 of these *Bylaws*. In addition, each Council shall include an ex officio member designated by the administrative officer, and such other ex officio members as specified in Article 5.5.d of these *Bylaws*. These ex officio members shall have voice but no vote.
- 8.5.b Appointment: Representatives of the designated administrative officer's office and the University Senate shall agree on nominees for vacancies on the Council. These nominations shall be submitted to the designated administrative officer for approval. In addition, these nominations shall be submitted to the University Senate for approval, or for election if specified in the Council's governing documents. In exercising its powers of appointment to the Council, the Senate shall follow procedures for review and approval for Senate committee appointments specified in Article 5.5.e of these *Bylaws*.
- 8.5.c Terms: Rules governing beginning date and length of terms, and restrictions on reappointment shall be specified in the governing documents of each Council. The presiding officer shall serve a three (3) year term and cannot be reappointed, unless otherwise specified in the governing documents of the Council.
- 8.5.d Appointment of Presiding Officer: The designated administrative officer and the Senate Chair shall reach an agreement on a presiding officer, and the joint choice shall be submitted to the Senate for approval. If the presiding officer is selected from among the membership of the Council, a replacement shall be appointed to the vacated seat.

#### 8.6 Operational Relationship of University Councils to Sponsors:

- 8.6.a The Office of the University Senate shall provide basic support for the activities of University Councils.
- 8.6.b The office of the designated administrative officer, through its ex officio University Council member,

- shall provide administrative support and serve as a liaison to other administrative units as required.
- 8.6.c The unit director shall provide the University Council with internal data, reports, studies, and any other materials required to support the Council's work. In addition, the director shall also arrange for unit staff to appear before the committee as requested.
- 8.6.d Control of the University Council's agenda shall be the responsibility of the presiding officer of the University Council and the voting members of the University Council in accordance with procedures for standing committees provided in Article 5.3.a, subject to the charges provided in Article 8.4 of these *Bylaws*, the appropriate subsection of Article 9 of these *Bylaws*, and the approved Task Force Report governing the University Council.
- 8.6.e Each University Council shall develop its own bylaws, which must be approved by the designated administrative officer and by the Senate.
- 8.6.f In addition to the required annual report, the presiding officer shall keep the Executive Secretary and Director and the Chair of the Senate informed of the major issues before the University Council and shall indicate when action or information items are likely to be forwarded for Senate consideration. In submitting recommendations for Senate action, the University Council shall inform the unit director and the designated administrative officer in advance of its recommendations. For purposes of conducting Senate business, reports from the University Council and floor privileges of the Senate shall be managed in the same manner as standing committees of the Senate defined in these Bylaws (3.3.c, 4.4.b). In the case where the presiding officer of the University Council is not a member of the Senate, they may report to the Senate and participate in the deliberations of the Senate subject to the provisions of Article 3.3.c of these Bylaws.

### 8.7 Review of University Councils:

- 8.7.a Five (5) years after a University Council is formed, a review of the University Council shall be undertaken jointly by the Senate and administration, and a written report issued. The review may recommend continuation of the University Council in its original form and mode of operation, modification of the University Council structure and/or operations, or discontinuance of the University Council.
- 8.7.b Following the initial review, the University Council and its operations shall be reviewed in conjunction with the periodic review of the *Plan*.

### ARTICLE 9 UNIVERSITY COUNCIL SPECIFICATIONS

### 9.1 University Library Council

- 9.1.a Charge: The University Library Council has the responsibility to provide advice and to report on policy issues concerning the University Libraries to the University Senate, to the Senior Vice President and Provost, and to the Dean of Libraries (see Appendix 1 for additional responsibilities and the Library Council's Bylaws).
- 9.1.b Membership: The Library Council shall consist of thirteen (13) appointed members and four (4) ex officio members. The appointed members shall be: the Chair, ten (10) faculty members including at least one (1) member of the library faculty, one (1) graduate student, and one (1) undergraduate student. The four (4) ex officio members shall be a representative of the Office of the Senior Vice President and Provost, a representative of the Office of the Dean of Libraries, a representative of the Division of Information Technology, and the Chair-Elect of the Senate.
- 9.1.c The Chair shall be a tenured faculty member.
- 9.1.d Reporting Responsibilities: The University Library Council shall report to the University Senate and the Senior Vice President and Provost under the terms of responsibility defined in Article 8.4 of these

Bylaws.

#### 9.2 University Research Council:

- 9.2.a Charge: In addition to the charges specified in Articles 5.2 and 8.4 of these *Bylaws*, the Research Council shall be governed by the following: The Research Council is charged to formulate and continually review policies regarding research, its funding, its relation to graduate and undergraduate academic degree programs, and its service to the community. Also, the Research Council is charged to review the research needs of faculty, other researchers and students, and to make recommendations to facilitate the research process and productivity of the University. Further, the Research Council shall formulate and continually review policies on the establishment, naming, reorganization, or abolition of bureaus, centers, or institutes that do not offer programs of instruction or regularly offer courses for credit, including their relationship to graduate and undergraduate academic programs. Additionally, when it perceives problems, the Research Council has the power to undertake investigative studies and recommend solutions.
- 9.2.b Membership: The University Research Council shall consist of thirteen (13) appointed members and ten (10) ex officio members. The appointed members shall be the Chair and eight (8) faculty members; one (1) staff member; and three (3) students, including at least one (1) graduate and one (1) undergraduate student. Eight (8) voting ex officio members include a representative of the Vice President for Research, a representative of the Dean of the Graduate School, a representative of the Dean of Undergraduate Studies, the Director of the Office of Research Administration and Advancement, the Chair of the Intellectual Property (IP) Committee and, and the Chairs of the subcommittees of the University Research Council. A representative of the President and a representative of the Senior Vice President and Provost shall serve as non-voting ex-officio members.
- 9.2.c The Chair shall be a tenured faculty member.
- 9.2.d Reporting Responsibilities: The University Research Council shall report to the University Senate and the Vice President for Research under the terms of responsibility defined in Article 8.4 of these *Bylaws* and the report establishing the University Research Council.

#### 9.3 University IT Council:

- 9.3.a Charge: The IT Council shall advise and report on policy issues concerning the Division of Information Technology to the University Senate and the Vice President for Information Technology and CIO. In addition to such responsibilities as are enumerated in Article 8 of these *Bylaws*, and as articulated in the Bylaws of the University IT Council (see Appendix 3), the IT Council shall:
  - 1) Respond to requests from the Division of Information Technology, extra-divisional advisory bodies (such as the Council of Deans or the Campus Student Technology Advisory Fee Committee), the University Senate, or other campus stakeholders for guidance on IT policy and implementation.
  - 2) Advise on the Division of Information Technology's budget, material resources, personnel, staffing and human resources, and administrative policies and practices.
  - Investigate matters concerning the Division of Information Technology and recommend solutions to the University Senate, the Vice President for Information Technology and CIO, or the general campus community.
  - 4) Advise on IT planning, including strategic and other major planning for information technology operation and development.
  - 5) Advise on policy recommendations related to campus technology facilities, equipment, software, and services.

- 9.3.b Membership: The IT Council shall consist of up to thirteen appointed members, and additional non-voting ex-officio members. The appointed members shall be: the chair, one staff member, one undergraduate student, one graduate student, one professional track faculty member, one tenured faculty member, and the chairs of the IT Council Working Groups. The non-voting ex-officio members shall include a representative from the University Libraries; a representative from the Office of the Provost; a representative from the Information Technology Advisory Committee; and the Vice President for Information Technology and CIO. Additional non-voting ex-officio members may be appointed as needed, by agreement between the Vice President for Information Technology and CIO and the Senate Executive Committee.
- 9.3.c The chair of the IT Council shall be appointed by the Vice President for Information Technology and CIO and the Senate, as described in 8.5 of these *Bylaws*. The chair will serve a three-year term.
- 9.3.d Working Groups: The IT Council may create up to seven Working Groups. These groups should carry out research and make recommendations on IT issues, and work with the appropriate Division of Information Technology staff member appointed by the Vice President for Information Technology and CIO. The specific responsibilities of each Working Group shall be described in the Bylaws of the University IT Council. The chair of each Working Group shall be appointed by the Vice President for Information Technology and CIO and the Senate and shall serve a two-year term.
- 9.3.e Reporting Responsibilities: The IT Council shall report to the Vice President for Information Technology and CIO and to the University Senate under the terms of responsibility defined in Article 8.4 of these *Bylaws*.

# ARTICLE 10 THE ATHLETIC COUNCIL

#### 10.1 The Athletic Council

- 10.1.a The Athletic Council exists to help the University develop and maintain the best possible intercollegiate athletic program consistent with the academic integrity of the institution and the academic and social development of student athletes. The Athletic Council shall operate in accordance with its charter (Appendix 4), which shall specify its role, scope, responsibilities, leadership, and membership. Changes to the charter shall be approved by the President of the University.
- 10.1.b Membership: The charter designates its membership. The membership of the Athletic Council elected by the Senate includes:
  - 1) Seven faculty members elected by the Senate at the annual Transition Meeting. Elected faculty representatives shall serve for a three-year term, and faculty who have served a full term shall for a period of one year be ineligible for re-election. The Senate should make very effort to assure diversity among the candidates for election to the Council.
  - 2) One staff member elected by the Senate at the annual Transition Meeting for a three-year term. A staff member who has served a full term shall for a period of one year be ineligible for re-election.
  - 3) The Chair of the Senate Campus Affairs Committee, or a faculty member designated by the Committee, shall serve as an ex-officio member.
- 10.1.c Relationship between the Senate and the Athletic Council:
  - 1) The Council in cooperation with the Athletic Director shall submit an annual report to the Senate on the status of intercollegiate athletics at the University. This report shall at least include an analysis of admissions, academic performance, class attendance, major selection, graduation rates, budget performance, and compliance with NCAA, Conference, and campus rules.

2) The Council shall inform the Senate for its review of any proposed amendments to the Council's charter.

# ARTICLE 11 DUTIES OF THE EXECUTIVE SECRETARY AND DIRECTOR

- 11.1 The Executive Secretary and Director of the Senate shall be responsible for the minutes and audio recordings of all Senate meetings.
  - 11.1.a The minutes shall include actions and business transacted, at a minimum. They shall be submitted to the Senate for approval. Copies of the approved minutes shall be made available to all chief administrative officers of Colleges, Schools, departments, and other units, and to the campus news media.
  - 11.1.b A complete audio recording shall be made of each meeting and shall be maintained by the Office of the University Senate. In accordance with the University's Records Retention and Disposal Schedule, a copy of each audio recording, excluding only those parts recorded during closed sessions, shall be placed with the minutes in the University Archives for open access.
- 11.2 The Executive Secretary and Director shall also maintain the following kinds of Senate records (see Article4.8):
  - (1) All material distributed to Senate members;
  - (2) All material received by or distributed to members of the Executive Committee;
  - (3) Any minutes of the Senate or the Executive Committee not otherwise included under (1) and (2);
  - (4) Annual reports of all committees of the Senate not otherwise included under (1) and (2);
  - (5) The audio records of Senate meetings;
  - (6) The current and all previous versions of the *Plan* and the *Bylaws*;
  - (7) Articles concerned with Senate structure and operation from campus and University publications as they come to the attention of the Executive Secretary and Director; and
  - (8) Other items deemed appropriate by the Executive Secretary and Director or the Chair of the Senate.
- 11.3 The Executive Secretary and Director shall store inactive records of the Senate in the University Archives.
- 11.4 The Executive Secretary and Director shall be responsible for the preparation of the Senate budget in accordance with Article 4.6.
- 11.5 The Executive Secretary and Director shall ensure that a directory of the membership of the new Senate indicating for each member the constituency, term, unit, and email address is made available on the Senate website following the annual transition of the Senate.
- 11.6 The Executive Secretary and Director shall ensure that contact information for all Senate officers and of all presiding officers of all Senate committees is made available on the Senate website for all members of the campus community.
- 11.7 The Executive Secretary and Director will normally provide a copy of the agenda and supporting material one week in advance of each Senate meeting.
- 11.8 The Executive Secretary and Director shall prepare for the members of the Senate and its Executive Committee, as appropriate, all agendas, minutes, reports, and other documents, with the exception of

- proposals relating to the Programs, Curricula, and Courses (PCC) Committee. Nonetheless, the Executive Secretary and Director shall be responsible for the distribution of all items of Senate business, and to other such committees as necessary.
- 11.9 The Executive Secretary and Director shall inform the Executive Committee of the status of all members of the Senate in accordance with the *Plan* (Article 3.4.a(3-4), 3.4.b(3-4), and 3.7) and these *Bylaws* (Articles 2.2, 4.1, 5.5, and 5.6).
- 11.10 The Executive Secretary and Director shall have the privilege of attending the meetings of all standing committees and ad hoc committees of the Senate to assist in the coordination of Senate business.
- 11.11 The Executive Secretary and Director shall provide information or assistance as requested for revision of the undergraduate catalog.

### ARTICLE 12 ANNUAL TRANSITION OF THE SENATE

#### 12.1 Preparation for Transition:

- 12.1.a By no later than the scheduled December meeting of the Senate, the Committee on Committees shall present to the Senate eight (8) nominees from among outgoing Senate members to serve on the Nominations Committee. The nominees shall include four (4) faculty members, one (1) exempt staff member, one (1) non-exempt staff member, one (1) graduate student, and one (1) undergraduate student. Further nominations shall not be accepted from the floor of the Senate. The Senate, as a body, shall approve the slate of nominees to serve on the Nominations Committee. The Chair-Elect of the Senate shall serve as a non-voting, ex officio member of the Nominations Committee. The Nominations Committee shall elect its own Chair from within the membership of the committee.
- 12.1.b The Nominations Committee shall solicit nominations from the membership of the Senate and shall present to the Chair of the Senate by April 15:
  - (1) A slate of at least two (2) candidates per seat from each constituency for elected membership on the Executive Committee, including those incumbent elected members who are eligible and willing to stand for reelection,
  - (2) Slates of candidates to replace the outgoing members of the Committee on Committees, the Campus Transportation Advisory Committee (CTAC), the University Athletic Council, and the Council of University System Faculty (CUSF), and any other committees as required by these *Bylaws*, including at least one (1) nominee for each position to be filled, and
  - (3) A minimum of two (2) candidates for the office of Chair-Elect.

Before reporting to the Chair of the Senate, the Nominations Committee shall secure the consent of all candidates in writing.

12.1.c A brief statement of each candidate's qualifications shall be sent to the voting membership of the incoming Senate ten (10) working days before the Transition Meeting of the Senate. Any further nominations made by members of the Senate and accompanied by a brief supporting statement and the consent of the candidate must be received by the Executive Secretary and Director at least twelve (12) working days before the Transition Meeting. These additional nominations shall be sent to the voting membership of the incoming Senate ten (10) working days before the Transition Meeting.

#### 12.2 Transition Elections:

- 12.2.a Election of the Chair-Elect shall occur as provided for in section 5.7.a of the Plan.
- 12.2.b The election of members of the Executive Committee, Committee on Committees, Campus Transportation Advisory Committee (CTAC), Athletic Council, Council of University System Faculty

(CUSF), and such other persons elected by the members of the Senate, shall be held after the election of the Chair-Elect.

- (1) Nominations for each of these committees and councils may be submitted in accordance with Article 12.1 above.
- (2) Nominations may also be received from the floor by the Chair at the Transition Meeting. Any such nomination is contingent on the consent of the candidate, which must have been secured beforehand in writing if the nomination is made in the absence of the candidate.
- (3) In the event of a tie vote in the election for members of the Executive Committee or the Committee on Committees, a ballot will be distributed to each Senator in the appropriate constituency. The election to break the tie should end one (1) week from the start date.

#### 12.3 Transition of the Senate:

- 12.3.a The new Senate session will begin at the Transition Meeting, which will be the last regularly scheduled meeting of the Spring semester.
- 12.3.b Newly elected Senators will be inducted at the Transition Meeting. Terms of office of newly elected Senators will begin, and the terms of the outgoing Senators will end, during the Transition Meeting.
- 12.3.c The outgoing Chair will pass the gavel to the previous Chair-Elect, who will assume the Chair role.
- 12.3.d The elected members of the outgoing Executive Committee and the Committee on Committees shall continue to serve until new members are elected.
- 12.3.e After the conclusion of the Transition Meeting, any vacancies on standing committees will be filled by the new Committee on Committees, subject to the approval of the Executive Committee and pending confirmation by the full Senate at its next regularly scheduled meeting.

# APPENDIX 1 BYLAWS OF THE UNIVERSITY LIBRARY COUNCIL

1. **Charge to the Library Council:** The University Library Council has the responsibility to provide advice about policy issues concerning the University Libraries to the University Senate, to the Senior Vice President and Provost, and to the Dean of Libraries.

#### A. The Council's Responsibilities to the University Senate:

- (1) Make recommendations for major changes and improvements in policies, operations, and services of the Libraries that represent the concerns and interests of Senate constituencies as well as other users of the Libraries. Such recommendations should specify the resource implications. Reports and recommendations to the University Senate shall be submitted to the Senate Executive Committee for placement on the agenda of the University Senate in the same manner as reports from the Senate's standing committees. It is expected that the Library Council will also inform the Senior Vice President and Provost in advance of these legislative recommendations. In addition to the mandatory annual report, the Chair of the Library Council shall keep the Chair of the Senate informed of the major issues before the Library Council and shall indicate when action or information items are likely to be forwarded for Senate consideration.
- (2) Respond to charges sent to the Library Council by the Senate Executive Committee.
- (3) Provide an annual written report of the Library Council's activities, including the status of recommendations made by the Library Council each year, and of unresolved issues before the Library Council.

### B. The Library Council's Responsibilities to the Senior Vice President and Provost:

- (1) Advise on the Libraries' budget, space, personnel and staffing, and other resources. It is expected that the Senior Vice President and Provost will consult the Library Council before undertaking major reviews of the Libraries with APAC and before preparing the annual budget for the Libraries.
- (2) Advise on the Libraries' administrative policies and practices.
- (3) Advise on the charges to be given to the committees to review the Dean of Libraries and to conduct the unit review of the University Libraries based on University policy
- (4) Advise on matters concerning the Libraries in conjunction with accreditation review and strategic planning.
- (5) Respond to requests for review, analysis, and advice made by the Senior Vice President and Provost.
- (6) Meet at least annually with the Senior Vice President and Provost to review the major issues facing the Libraries and its activities on campus.
- (7) The Library Council is responsible for informing the Senior Vice President and Provost of pending reports and recommendations to the University Senate.

#### C. The Library Council's Responsibilities to the Dean of Libraries:

- (1) Advise on the needs and concerns of diverse constituencies within the campus community with respect to Library policies, services, and new resources and technology.
- (2) Advise on strategies to involve Library users in the initiation, evaluation, and integration of new Library policies, practices, procedures, and technology. Such strategies might include forums for the discussion of changes, workshops for adjusting to new technologies, and ongoing programs of Library education.
- (3) Advise on operations, policies and new opportunities.
- (4) Advise on Library planning including strategic planning and other major plans for Library operation and development.

- (5) Review and advise on the Libraries' reports, studies, and proposed initiatives that have significant long-term resource implications for the Libraries.
- (6) Hold at least one (1) meeting each year at which the Dean shall review major issues and plans, summarized in a State of the Libraries report distributed in advance to the Library Council.
- (7) It is expected that the Library Council will adopt a broad campus perspective and that the Dean of the Libraries will inform the Library Council of the University Libraries' needs and concerns and seek advice about major modifications of policies and operations affecting the campus community.

### D. To Fulfill Its Responsibilities, the Library Council May:

- (1) Undertake investigative studies in matters concerning the University Libraries and recommend solutions to the University Senate, the Senior Vice President and Provost, the Dean of Libraries, or the general campus community.
- (2) Conduct open hearings on major issues concerning the University Libraries and their activities.
- (3) Communicate directly with the campus community on concerns related to support for, policies of, and services provided by the University Libraries.
- 2. **Composition of the Library Council:** The Library Council shall consist of thirteen (13) appointed members and four (4) ex officio members. The appointed members shall be: the Chair, ten (10) faculty members including at least one (1) member of the Library faculty, one (1) graduate student, and one (1) undergraduate student. The four (1) ex officio members shall be a representative of the Office of the Senior Vice President and Provost, a representative of the Dean of the Libraries Office, a representative of the Division of Information Technology, and the Chair-Elect of the Senate.

#### A. Tenure in Office:

- (1) The Library Council Chair should be a tenured faculty member appointed for a single three-year term. Normally, the Chair shall have served as a member of the Library Council. If the Chair is serving as a regular member of the Library Council at the time of appointment, a new member shall be appointed to serve the remainder of the term the Chair has vacated. The Senior Vice President and Provost and the Senate Executive Committee shall reach an agreement on the Library Council Chair, and the joint choice shall be submitted to the University Senate for its approval.
- (2) The remaining ten (10) faculty members shall be appointed for staggered two-year terms. No faculty member shall serve more than two (2) terms consecutively. For this purpose, members who have served more than a year should be considered to have served a full term.
- (3) The two (2) student members shall be appointed for one-year terms. No student member should serve more than two (2) terms consecutively. For this purpose, student members who have served more than half their term should be considered to have served a full term.
- (4) The Office of the Senior Vice President and Provost will appoint a member of the Provost's staff as an ex officio member of the Library Council who will have voice but not vote.
- (5) The Dean of Libraries' Office will appoint an upper-level member of the Libraries' administrative staff as an ex officio member of the Library Council who will have voice but no vote.
- (6) The Vice President for Information Technology and Chief Information Officer (CIO) will appoint a member of the Division of Information Technology's staff as an ex officio member of the Library Council who will have voice but no vote.
- (7) The Chair-Elect of the Senate shall serve as an ex officio member of the Library Council who will have voice but no vote.

- B. Qualifications of Library Council Members: Successful operation of the Library Council requires that the members of the Library Council understand the nature of the Libraries and represent the best interests of the campus as well as the particular interests of their specific constituencies.
  - 1. The Library Council members should be chosen from people who can bring a campus-wide perspective to their deliberations on Library matters and who have shown interest and willingness to foster a good working relationship between the Libraries and their users.
  - Library Council members should be selected to represent as broad a range of campus disciplines and
    interests as possible. Faculty members should include representatives from both the professional and arts
    and sciences colleges, and within these constituencies, representatives of the arts and humanities, social
    sciences, and physical and biological sciences.
- C. The Appointment Process: In the spring of each year, the Chair of the University Library Council shall notify the representative of the Office of the Senior Vice President and Provost and the Chair-Elect of the Senate of the appointments required for the following academic year. The representative of the Office of the Senior Vice President and Provost and the Chair-Elect of the Senate shall draw up a slate of nominees who will agree to serve, and the slate will be submitted to the Senior Vice President and Provost and the Committee on Committees for approval. The list of nominees for Library Council membership shall be submitted to the University Senate for approval. Ordinarily, the slate will be presented at the same Senate meeting at which other committee slates are approved. Dates of appointment and beginning of terms shall correspond with those of Senate committees. Replacement of Library Council members will take place through the same consultative process as the initial appointment, with submission of names to the Senate occurring as needed.
- 3. **Operation of the Library Council:** Effective and efficient Library Council operation will require adequate support and full cooperation among the Senate, the Senior Vice President and Provost, the Dean, and their offices.
  - A. The Office of the University Senate or its designee will provide normal committee support to the Council, including maintaining mailing lists, reproducing Library Council documents, keeping a copy of Library Council minutes, maintaining files for the Library Council, and arranging meeting rooms.
  - B. The Office of the Senior Vice President and Provost, through its ex officio Library Council member, will provide liaison to other administrative units, such as the Office of Institutional Research, Planning and Assessment, for their reports, data, or assistance. The Office of the University Senate will also provide website space for the Library Council.
  - C. The Dean of the Libraries will provide the Library Council with internal data, reports, studies, etc. as needed to support the Library Council's work. The Dean will also arrange for unit staff to present testimony concerning such reports as the Library Council finds useful in carrying out its responsibilities. The Dean's assistance to the committee shall also include providing the Library Council members with the opportunity to attend an appropriate orientation session dealing with the Libraries.
  - D. Control of the Library Council's agenda will be the responsibility of the Library Council Chair and the voting members of the Library Council.
  - E. While being responsive to the needs of the Senior Vice President and Provost and the Senate in a timely manner is necessary, the sponsoring parties and the Dean of the Libraries must not attempt to micro-manage the ongoing operation of the Library Council. In turn the Library Council must not attempt to micro manage the Libraries.
  - F. The Library Council shall meet as necessary, but in no case less than once per semester. Meetings may be called by the Chair. In addition, upon receiving a request of any three members of the Library Council, the Chair shall call a meeting. A majority of the voting members of the Library Council shall constitute a quorum for the conducting of official business of the Library Council.

#### 4. Operational Relationship of the Library Council to its Sponsors:

A. For purposes of University Senate action, a Library Council created through Senate action will appear in essentially the same role as a standing committee of the University Senate.

- B. The Chair may present reports and recommendations to the Senate but will not have a vote in Senate proceedings, unless he or she is a member of the Senate.
- C. Since the committees of the Senior Vice President and Provost range widely in form and function, and do not operate under a formal plan of organization and bylaws, there is no need to specify the Library Council's standing in the same fashion. For other purposes, such as APAC review of the Unit, the Library Council might be consulted like a College Advisory Council (that colleges will have under the shared governance plan) could be.
- D. The Dean of Libraries will ordinarily meet with the Library Council and have a voice in its deliberations. Since one of the three main functions of the Library Council is to advise the Dean, the Dean shall not formally be a member of the Library Council. On formal reports and recommendations of the Library Council to the University Senate or to the Senior Vice President and Provost, the Dean of the Libraries may send a separate memorandum to the Senate or the Senior Vice President and Provost, as appropriate, supporting or opposing the report or the recommendations, and providing rationale for the Dean's position.
- 5. **Review of the Library Council:** The Library Council and its operations will be reviewed in conjunction with the periodic review of the Senate and the *Plan*.

### APPENDIX 2 BYLAWS OF THE UNIVERSITY RESEARCH COUNCIL

{To be inserted once available}

# APPENDIX 3 BYLAWS OF THE UNIVERSITY IT COUNCIL

Charge to the University Information Technology (IT) Council: The IT Council has the responsibility to facilitate
alignment of vision, priorities, and pace of IT investments and to recommend IT policies to the University Senate
and administration. The IT Council is supported by Working Groups, which facilitate campus-wide communication
related to IT matters.

#### A. The Council's Responsibilities to the University Senate:

- 1) Advise on strategic issues involving the University's use of IT, information security, access, retrieval and content stewardship, and telecommunication and knowledge dissemination.
- 2) Bring IT initiatives and proposals to the Senate for consideration and review.
- 3) Keep the Senate informed of strategic IT matters through periodic updates.
- 4) Respond to charges sent to the IT Council by the Senate Executive Committee.
- 5) Provide an annual written report of the IT Council's activities.

# B. The IT Council's Responsibilities to the Vice President for Information Technology and Chief Information Officer (CIO):

- Advise on policy recommendations related to campus technology facilities, equipment, software, and services - particularly in the areas of computing (both academic and administrative), networking, and telecommunications.
- 2) Advise on IT planning, including strategic and other major planning for information technology operation and development.

- 3) Advise on the Division of Information Technology's budget, space, and other material resources, in addition to personnel, staffing and other human resources.
- 4) Advise on the Division of Information Technology's administrative policies and practices.
- 5) Respond to requests for review, analysis, and advice made by the Vice President for Information Technology and CIO.

# C. The IT Council's Responsibilities to the Deans, the Campus Student Technology Advisory Fee Committee, and the Campus Community:

- 1) Ensure the distribution of information concerning available campus technology services and how they might be best used to serve the campus community.
- 2) Seek input from current and prospective users concerning types of technology services the campus can provide.
- 3) Respond to input from current users concerning the quality of campus technology services.

#### D. To Fulfill Its Responsibilities, the IT Council May:

- Investigate matters concerning the Division of Information Technology and recommend solutions to the University Senate, the Vice President for Information Technology and CIO, or the general campus community.
- Conduct open hearings on major issues concerning the Division of Information Technology and its activities.
- 3) Communicate directly with the campus community on concerns related to the Division of Information Technology's services and policies.
- 2. **Organizational Structure of the IT Council:** The IT Council shall include five standing Working Groups, each of which will have a chair.
- 3. Composition of the IT Council: The IT Council shall consist of eleven appointed members and additional non-voting ex-officio members. The appointed members shall be: the chair, one staff member, one undergraduate student, one graduate student, one professional track faculty member, one tenured faculty member, and the chairs of the five IT Council Working Groups. The non-voting ex-officio members shall be a representative from the University Libraries; a representative from the Office of the Provost; a representative from the Information Technology Advisory Committee (ITAC); and the Vice President for Information Technology and CIO. Additional non-voting ex-officio members may be appointed as needed, by agreement between the Vice President for Information Technology and CIO and the Senate Executive Committee.

#### A. Tenure in Office:

- The IT Council chair should be a tenured faculty member, and is appointed for a single, three-year term. Normally, the chair shall have served as a member of the IT Council. If the chair is serving as a regular member of the IT Council at the time of appointment, a new member shall be appointed to serve the remainder of the term the chair has vacated. The Vice President for Information Technology and CIO and the Senate Chair shall reach an agreement on the IT Council chair, and the joint choice shall be submitted to the University Senate for its approval.
- 2) The five Working Group chairs shall be appointed for staggered two-year terms. No working group chair shall serve more than two terms consecutively. For this purpose, members who have served more than a year should be considered to have served a full term.

- 3) The two faculty members (professional track and tenured) shall be appointed for two-year terms. No faculty member shall serve more than two terms consecutively. For this purpose, members who have served more than a year should be considered to have served a full term.
- 4) The staff member shall be appointed for a two-year term. No staff member shall serve more than two terms consecutively. For this purpose, members who have served more than a year should be considered to have served a full term.
- 5) The two student members shall be appointed for one-year terms. No student member should serve more than two terms consecutively. For this purpose, student members who have served more than half their term should be considered to have served a full term.
- 6) The Dean of the Libraries will appoint a representative from the University Libraries as a non-voting ex officio member of the IT Council.
- The Provost will appoint a representative from the Office of the Provost as a non-voting ex-officio member of the IT Council.
- 8) The Information Technology Advisory Committee (ITAC) will appoint a representative from the committee as a non-voting ex-officio member of the IT Council.
- The Vice President for Information Technology and CIO, or a designee, shall serve as a non-voting exofficio member of the IT Council.
- B. **Qualifications of IT Council Members:** Successful operation of the IT Council requires that its members understand the nature of the Division of Information Technology and represent the best interests of the campus as well as the particular interests of their specific constituencies.
  - IT Council members should be chosen from people who can bring a campus-wide perspective to their deliberations on IT matters and who have shown interest and willingness to foster a good working relationship between the Division of Information Technology and its users.
  - 2) IT Council members should be selected to represent as broad a range of campus disciplines and interests as possible. Faculty members should include representatives from the various disciplines on campus ranging from the arts and humanities and social sciences to the physical and biological sciences and engineering.
- C. The Appointment Process: In the spring of each year, the Senate Office shall notify the Vice President for Information Technology and CIO and the Chair of the Senate of the appointments required for the following academic year. The Vice President for Information Technology and CIO and the Chair of the Senate shall draw up a slate of nominees who will agree to serve, and the slate will be submitted to the Committee on Committees for approval. The final slate of nominees for IT Council membership shall be submitted to the University Senate for approval. Ordinarily, the slate will be presented at the same Senate meeting at which other committee slates are approved. Dates of appointment and beginning of terms shall correspond with those of Senate committees. Replacement of IT Council members will take place through the same consultative process as the initial appointment, with the submission of names to the Senate occurring as needed.

### 4. Operation of the IT Council

A. The Division of Information Technology or its designee will provide normal committee support to the Council, including maintaining mailing lists, reproducing IT Council documents, keeping IT Council minutes and agendas on an IT governance website, and arranging meeting rooms.

- B. Control of the IT Council's agenda will be the responsibility of the IT Council chair and the voting members of the IT Council.
- C. While being responsive to the needs of the Vice President for Information Technology and CIO and the Senate in a timely manner is necessary, the Working Groups and the sponsoring parties as well as the Deans, the Campus Student Technology Fee Advisory Committee, and the campus community must not attempt to micro-manage the ongoing operation of the IT Council. In turn, the IT Council must not attempt to micro-manage the Division of Information Technology.
- D. The IT Council should typically meet once every month and shall meet at least once per semester. Meetings will be scheduled by Division of Information Technology staff, in consultation with the IT Council chair and the Vice President for Information Technology and CIO.
- 5. Working Groups of the IT Council: The Working Groups will serve in an advisory capacity to the IT Council. These groups should carry out research and make recommendations on IT issues, and shall each work with the appropriate Division of Information Technology staff member, as appointed by the Vice President for Information Technology and CIO.

#### A. The five Working Groups shall be:

- 1) IT Infrastructure Working Group, which focuses on building and maintaining a sound, advanced, secure, and productive physical information technology infrastructure (including but not limited to facilities, hardware, networks, and software) capable of supporting broad and effective use by students, faculty, and staff throughout the institution, including remote University members such as agricultural extension offices.
- 2) Learning Technologies Working Group, which provides the vision, priorities, and pace for enterprise learning technology solutions and services to be undertaken on campus. Its work focuses on endorsing the adoption of new learning technology solutions, as well as making recommendations for upgrading or decommissioning existing services. Working group members are nominated by the Deans.
- 3) Research Technologies Working Group, which provides the vision, priorities, and pace for enterprise research technology solutions and services to be undertaken on campus. Its work focuses on endorsing the adoption of new research technology solutions, as well as making recommendations for upgrading or decommissioning existing services. Working group members are nominated by the Deans.
- 4) Administrative Systems Working Group, which advises the Vice President for Information Technology and CIO in matters of enterprise-wide administrative system technology decisions and priorities.
- 5) IT Security Working Group, which advises the Vice President for Information Technology and CIO on IT security matters. The focus is on securing the integrity of information technology resources, safeguarding institutional information, protecting the privacy of University community members in their use of IT, and ensuring the continuity of the institution's IT resources and information repositories in the face of possible disaster scenarios.
- B. Composition of the Working Groups: Each Working Group will have a chair appointed by the Vice President for Information Technology and CIO and the Senate Chair for a two-year term. The membership of each Working Group will be appointed by the Vice President for Information Technology and CIO unless otherwise specified above (5.A.2 and 3), but will be flexible so that additional members can be engaged in the decision-making and review process as appropriate. The membership of each Working Group shall include a combination of faculty, staff, and students.
- C. Terms on Working Groups shall be two (2) years for faculty and staff. Appointments to two-year terms shall be staggered: that is, as far as practical, half of the terms from each faculty or staff constituency shall expire each year. Terms shall begin on July 1 of the appropriate year.

D. Meetings of the Working Groups: The Working Groups usually meet three to four times a semester.

#### E. Working Group Responsibilities:

- 1) Provide knowledge in a particular area and serve as an advisory board, by which the IT Council can route items for review and comment.
- 2) Submit proposals and issues to the IT Council for consideration and/or funding.
- 3) Assist in the annual review and update of the Information Technology Strategic Plan.

#### 6. Operational Relationship of the IT Council to its Sponsors:

- A. For purposes of University Senate action, the IT Council will appear in essentially the same role as a standing committee of the University Senate.
- B. The IT Council chair may present reports and recommendations to the Senate but will not have a vote in Senate proceedings, unless he or she is a member of the Senate.
- C. The Vice President for Information Technology and CIO is an ex-officio member of the IT Council and has a voice in its deliberations.
- 7. **Review of the IT Council:** The IT Council and its operations will be reviewed in conjunction with the periodic review of the Senate and the Plan.
- 8. Amendments: Amendments to these Bylaws shall be provided to the IT Council members a minimum of seven calendar days in advance of any regular meeting. Approval shall require a two-thirds vote of the present and voting regular membership of the Council. Upon approval, a revised copy of the Bylaws shall be sent to the Senate Office.

# APPENDIX 4 CHARTER OF THE UNIVERSITY ATHLETIC COUNCIL

The University of Maryland at College Park is dedicated to higher learning, research, and public service. An intercollegiate athletic program can significantly contribute to the learning and the public service components of the Campus Mission. The operation of a successful athletic program fosters spirit, identity and a sense of pride within the campus community and provides talented student-athletes with the opportunity to enrich their collegiate experience through participation in a challenging and competitive athletic program. Excellence of the athletic program at College Park stems not only from successful competition, but more importantly, from the general involvement in the Campus milieu of student-athletes who will earn degrees and who in other respects, embody qualities with which the institution can identify. Most importantly, both athletic success and academic integrity are the crucial elements in judging the excellence of the athletic program at the University of Maryland at College Park.

The importance of faculty involvement and influence in the institutional control and operation of an excellent athletic program cannot be overestimated. Faculty advice and participation will enhance the integrity of the athletic program in terms of academic performance, rules compliance, and compatibility of athletic programs with the mission of the campus.

#### PURPOSE OF THE ATHLETIC COUNCIL

First and foremost, the Athletic Council exists to help the University develop and maintain the best possible intercollegiate athletic program consistent with the academic integrity of the institution and the academic and social development of student athletes. The Athletic Council is the primary body, which advises the President on all matters relating to intercollegiate athletics. It is responsible for formulation and recommendation of policy matters affecting intercollegiate athletics and for monitoring the implementation of such policy by the intercollegiate athletics program. The Council, on behalf of the President, provides the necessary faculty input and participation in intercollegiate athletics as required by the Big Ten Conference, National Collegiate Athletic Association and the University of

Maryland at College Park. The Council does not execute policy but serves to influence policy development and administration.

This document delineates the responsibilities, processes, and membership of the Athletic Council at the University of Maryland at College Park. It is expected that the Council will be proactive in its task of preparing policy recommendations and monitoring their implementation by the intercollegiate athletics program. The Council expects to have the full support of the Campus in the responsible performance of its duties.

#### FUNCTION/DUTIES OF THE ATHLETIC COUNCIL

The major function of the Athletic Council is to assist the President and the Director of Intercollegiate Athletics in the exercise of "institutional responsibility and control of intercollegiate athletics" as required by the constitution of the Big Ten Conference, the National Collegiate Athletic Association and the University of Maryland at College Park. The Council functions in advisory, compliance, liaison, and representative capacities. The Athletic Council shall meet at least four times each year, twice in each semester, and at such other times as needed to carry out the duties of the Council. Specific duties of the Council shall include but not be limited to the following:

- 1. Promote an understanding of intercollegiate athletics among faculty, students, staff, alumni and other members of the University of Maryland at College Park community.
- 2. Promote the adoption and implementation of appropriate policies for the admission and continuing eligibility of student athletes at the University of Maryland at College Park.
- 3. Monitor the preparation of the athletic budget by the Athletic Director during the regular budgetary process and make recommendations to the Athletic Director and the President concerning sources (i.e. student athletic fees) and allocations of funds.
- 4. Participation in the selection process for the Director of Intercollegiate Athletics and the head coaches in all sports including, if possible, informal meetings of the final candidates with the Executive Committee in the interview process. A faculty member from the Athletic Council should be included on all search committees for head coaches.
- 5. Establish criteria and make recommendations, with the advice of the Athletic Director, regarding which sports shall be certified as intercollegiate sports.
- 6. Recommend policies concerning athletic schedules, practice, the number of contests to be played each year in each sport and the NCAA category of schools with which it is desirable to compete.
- 7. Establish guidelines for and make recommendations regarding the acceptance of invitations to post-season events, special holiday games, or other events outside the regular season schedule.
- 8. Review and formulate policies concerned with substance abuse that will provide protection to the health of student-athletes and ensure that such policies have a strong educational emphasis.
- 9. Review and endorse policy on physical facilities necessary for the conduct of a competitive Division I-A program.
- 10. Review and formulate policies on recruitment and the awarding of athletic grants and scholarships to student-athletes who meet eligibility standards.
- 11. Review and approve the criteria for departmental awards in recognition of athletic and academic achievement.
- 12. Review athletic event price schedules, seating priorities and allocation of tickets to various groups.
- 13. Monitor the advisement, academic support and counseling services available to student-athletes.
- 14. Review and formulate policy concerning the conduct of home athletic contests, particularly with respect to the protection and safety of participants and spectators.

- 15. Review and formulate policy regarding the expectations of and behavior of coaches and student-athletes.
- 16. Review and formulate policy regarding the expectations of and behavior of cheerleaders and their advisors.
- 17. Assist with the development of official reports to be submitted by the President for filing with the conference or appropriate associations.
- 18. Review with appropriate authorities the financial audits of the Department of Intercollegiate Athletics.
- 19. Monitor the activities of the Department of Intercollegiate Athletics to make sure that they are in compliance with Conference (Big Ten) and Association (NCAA) bylaws, regulations and legislation.

In fulfilling its functions/duties, the Athletic Council

- · must maintain confidentiality;
- shall have available to it complete information on all items which appear for its consideration and shall have full opportunity for discussion of each item before action is taken;
- shall have available full and current information on the financial, academic and related activities of the intercollegiate athletics program; and
- is authorized to recommend to the President the employment of experts from outside the Campus when their advice is needed.

#### RESPONSIBILITIES OF THE CHAIR OF THE ATHLETIC COUNCIL

The Athletic Council has a Chair who is selected by the President from the faculty. The duties of the Chair shall include:

- 1. Serve as a spokesperson for the Council in all contacts with the media.
- 2. Serve as the Faculty Representative to the Big Ten Conference and the National Collegiate Athletic Association (NCAA).
- 3. Chair meetings of the Athletic Council and the Executive Committee of the Council.
- 4. Call regular meetings of the Athletic Council and such special meetings as may be necessary.
- 5. Prepare the agenda for meetings of the Athletic Council and of the Executive Committee of the Council.
- 6. Represent the campus, as authorized by the President, at meetings of the NCAA, Big Ten, United States Intercollegiate Lacrosse Association, United States Olympic Committee, Intercollegiate Athletic Association of America, College Football Association and other groups which establish international, national and regional policies for intercollegiate athletics.
- 7. Advise the President and serve as spokesperson to the faculty on behalf of the President on appropriate matters.
- 8. Report to the President on all actions taken by the Athletic Council.
- 9. Work with the Director of Intercollegiate Athletics in coordinating and carrying out the functions of the Athletic Council.
- 10. Monitor activities of the Department of Intercollegiate Athletics and confer regularly with the President on matters which should come to the President's attention.
- 11. Ensure that required reports and recommendations from the Athletic Council are provided to the President.
- 12. Report to the President and the Athletic Director on the concerns of the Athletic Council relative to athletics and to interpret to the faculty and other groups the University's athletic policies and activities.

- 13. Ensure that all actions of the Chair and the Executive Committee made on behalf of the Council are properly recorded and reported to the full membership of the Council in a timely manner.
- 14. Coordinate with the President's Office all financial support necessary to carry out the duties of Chair, including the development of an annual budget for this support; and the approval of all requests for expenditures and expense reimbursements made for this purpose. The President's Office is the administrative unit responsible for providing appropriate financial support to the Chair of the Athletic Council/Faculty Athletic Representative, and for approving both the annual budget request for this support as well as all expenditures, and expense reimbursements made for this purpose.
- 15. Know, recognize, and comply with the laws, policies, rules and regulations governing the University and its employees, and the rules of the National Collegiate Athletic Association (the "NCAA") and the Big Ten. Inform the Athletic Department Compliance Officer immediately of any suspected violation. Assist, as requested, in the investigation and reporting of those violations.

#### RESPONSIBILITIES OF THE VICE CHAIR OF THE ATHLETIC COUNCIL

The Athletic Council has a Vice-Chair who is selected by the President from the faculty. The duties of the Vice-Chair shall include:

- 1. Assist the Chair of the Council with conducting the business and meeting of the Council.
- 2. Conduct meetings of the Council in the absence of the Chair.
- Write periodic articles for University publications about the actions of the Council.
- 4. Serve on the Executive Committee of the Council.
- 5. Coordinate the activities of and serve as an ex officio member to standing committees of the Council.
- 6. Know, recognize, and comply with the laws, policies, rules and regulations governing the University and its employees, and the rules of the National Collegiate Athletic Association (the "NCAA") and the Big Ten Inform the Athletic Department Compliance Officer immediately of any suspected violation. Assist, as requested, in the investigation and reporting of those violations.

### ATHLETIC COUNCIL MEMBERSHIP

Intercollegiate Athletics plays an important role in fostering pride and spirit in the University community. The Athletic Council membership is designed to be representative of this community and shall consist of faculty, administration, staff, students and alumni. The membership shall include minorities, women and men, and thorough consideration will be given to ensure a balanced representation on the Council. The Athletic Council shall consist of twenty voting and five non-voting members appointed by the President or elected by the Senate as follows:

Voting Members of the Athletic Council

- The Athletic Council has a Chair who is selected by the President from the faculty. The duration of the Chair's
  membership on the Council is determined by the President. The initial appointment is for a five year term
  which may be renewed by the President.
- The Athletic Council has a Vice-Chair who is selected by the President from the faculty. The duration of the Vice-Chair's membership on the Council is determined by the President. The initial appointment is for a three year term which may be renewed by the President.
- Seven faculty members of the Athletic Council will be elected by the Senate. These elected faculty members will serve for a three year period and are not eligible to serve a second consecutive three year period. The Senate should make every effort to assure diversity among the elected members.
- The Faculty member who is Chair of the Campus Affairs Committee of the Senate or a designee from the Committee who must be a faculty member is a member of the Athletic Council.

- One Academic Dean appointed by the Provost. The appointment is for a one year term which may be renewed
  by the Provost.
- Two staff members, one who is appointed by the President for a three year period and one who is elected for a three year period by the Senate. These staff members will serve on a staggered basis and are not eligible to serve a second consecutive three year period.
- The Vice President for Student Affairs.
- One representative from the "M" Club. The appointment is for one year.
- One representative from the Terrapin Club. The appointment is for one year.
- One student representative from the Student Government Association. The appointment is for one year.
- One undergraduate female athlete. The appointment is for two years and the student should maintain eligibility in her sport.
- One undergraduate male athlete. The appointment is for two years and the student should maintain eligibility in his sport.
- One graduate student. The appointment is for two years and the student should maintain good standing in the Graduate School.

#### Non-Voting Members of the Athletic Council

- The Director of Intercollegiate Athletics.
- A Representative from the President's Office.
- A Representative of the Office of General Counsel.
- · The Director of the Student Health Services.
- The Director of the Office of Alumni Programs for the University of Maryland at College Park.
- A current head coach selected by the coaches as their representative. This appointment will be a one-year appointment with a three year limit.

In making all non-elected appointments to the Athletic Council, the President should solicit recommendations from the following advisory groups or persons: Executive Committee of the Athletic Council, President of the Student Government Association, President of the Graduate Student Government, Dean of the Graduate School, and Director of Intercollegiate Athletics. The term of office of all members of the Council shall begin with the first meeting of the new academic year.

Vacancies occurring on the Council due to resignation or other cause will be filled as they occur. If the vacancy is one of the members of the Council elected by the Senate, the Senate will be asked to elect a replacement to fill the vacancy. For all other vacancies, the President will solicit nominations from the appropriate groups and appoint a replacement to fill the remainder of the unexpired term. Persons appointed to fill a partial term on the Council will be eligible for election or appointment to a full term as appropriate for their membership category.

### **COMMITTEES OF THE ATHLETIC COUNCIL**

Committees of the Athletic Council shall include an Executive Committee, Standing Committees of the Council, and Ad-Hoc Committees as needed. The major responsibilities and membership of these Committees of the Athletic Council follow.

1. Executive Committee. The membership of the Committee is as follows: The Chair of the Athletic Council who will

serve as chair, the Vice-Chair of the Athletic Council, chairs of the five standing committees of the Athletic Council, the representative from the President's office, and a staff or student member of the Athletic Council. If one or more of the Chairs of the standing committees are not faculty, the membership of the Executive Committee will be adjusted to include four faculty in addition to the Chair. Total membership of the Executive Committee will not exceed eight at any time. The responsibilities of the Executive Committee shall include the following:

- Meet on matters calling for immediate action and at times when meetings of the full Athletic Council are not possible.
- Identify and assign problems to standing subcommittees and ad-hoc committees for study and receive reports from these committees.
- Serve as the personnel committee of the Council upon request of the President.
- Review compliance reports submitted by the Department of Intercollegiate Athletics and ensure that the Department is in compliance with all Conference and Association policies.
- Advise the President on an emergency basis.
- Recommend faculty and staff for membership on the Athletic Council.
- 2. **Standing Committees of the Athletic Council.** The Chair of the Athletic Council will select the Chairs of the Standing Committees and will appoint each committee and, with the exception of the Academic Committee, will appoint each committee after soliciting volunteers from the Council membership.
  - a. Academic Committee. All faculty members of the Council are members of the committee. The general role of the Academic Committee is to ensure that appropriate academic standards are established and maintained for all student-athletes and that all participants recognize the priority of successful academic performance by all student-athletes. In fulfilling this function, the Committee shall make appropriate recommendations to the Athletic Council. In particular, the Committee shall have the following duties:
    - Recommend policies and procedures regarding standards and criteria for admission of student-athletes.
    - Recommend academic policies and procedures regarding standards and criteria for continuing eligibility of student-athletes to participate in intercollegiate sports.
    - Consider and decide academic appeals of student-athletes concerned with eligibility.
    - Review every semester the academic program and progress of student-athletes.
    - Recommend policies for and monitor the activities of the academic support services provided to the student-athletes.
    - Recommend policies regarding post-season and tournament participation by athletic teams.
    - Recommend policies regarding scheduling and practice time.
  - b. **Budget and Facilities Committee.** The general purpose of this Committee is to monitor but not manage those activities of the Athletic Department pertaining to budget and facilities. In fulfilling this function, the Committee shall make appropriate recommendations to the Athletic Council. More specifically, responsibilities of the Committee shall include the following:
    - Monitor the preparation of the athletic budget(s) by the Director of Intercollegiate Athletics.
    - Review and analyze for the Council the final budget(s) submitted by the Director of Intercollegiate Athletics to the President.
    - Establish criteria and make recommendations, with the advice of the Director of Intercollegiate Athletics, regarding which sports shall be certified as intercollegiate sports.

- Review policies regarding the number and distribution of athletic scholarships to be awarded annually.
- Review and recommend policies for athletic event price schedules, seating priorities and allocation of tickets to various groups.
- Review and recommend policies regarding utilization and development of intercollegiate athletic facilities.
- Monitor the financial accountability of the Department of Intercollegiate Athletics.
- c. Student Life Committee. This Committee is concerned with all non-academic aspects of the student-athlete's involvement with the University. In fulfilling this function, the Committee shall make appropriate recommendations to the Athletic Council. More specifically, the responsibilities of the Committee shall include the following:
  - Review and recommend policies concerning the nature and type of health screening and drug testing.
  - Review and recommend policies regarding practice schedules.
  - Review and recommend policies for determining when health and other non-academic factors will be used to restrict a student's involvement in intercollegiate athletics.
  - Review and recommend policies for and monitor activities of non-academic support programs and placement services.
  - Review and recommend policies regarding scholarship awards and retention of these awards.
  - Review and recommend policies for housing assignments.
  - Assist the Athletic Council in assuring the personal and social development of all student-athletes and their full integration into campus life.
- d. **External Affairs Committee.** This Committee is concerned with external activities of the Department of Intercollegiate Athletics. In fulfilling this function, the Committee shall make appropriate recommendations to the Athletic Council. More specifically, the responsibilities of the Committee shall include the following:
  - Review and endorse fundraising activities.
  - Review and recommend policies for complementary distribution of tickets to athletic events.
  - Review and recommend guidelines and/or policies for all sports marketing activities (i.e. sports camps, special events, endorsements, etc.)
  - Review and recommend guidelines and/or policies for interactions with alumni and friends of the Athletic Department including the Terrapin Club, the "M" Club, and the Maryland Education Foundation.
  - Review and recommend policies and/or guidelines for all media interactions.
- e. **Professional Sports Counseling Panel (PSCP).** The PSCP is a committee of the Athletic Council authorized under NCAA by-law 12.3.4 to advise and assist student athletes in preparation for professional athletic careers. Consonant with its charge, the University of Maryland, College Park PSCP provides:
  - Education and advice to student athletes about NCAA amateurism rules and professional sports careers.
  - Oversight to the Athletic Department's implementation of University and NCAA regulations regarding contacts between student athletes and agents.
  - Advice to the Athletic Council on matters related to its charge.
- 3. Ad-Hoc Committees. The Chair of the Athletic Council, upon advice of the Council, will appoint Ad-Hoc

Committees as needed. Membership on these committees will be on a volunteer basis or by appointment by the Chair of the Council after seeking advice from the Executive Committee.

#### **MEETINGS OF THE ATHLETIC COUNCIL**

The Chair of the Council serves as the spokesperson for the Council. Meetings of the Council are open only to Council members and invited guests. Individuals who are not members of the Council, but who wish to attend a specific meeting should seek the prior approval of the Chair. Information provided to Council members concerning specific personnel or compliance matters will not be divulged by individual members without permission of the Chair.

# APPENDIX 5 PROCEDURES FOR ELECTIONS OF UMCP REPRESENTATIVES TO THE COUNCIL OF UNIVERSITY SYSTEM FACULTY (CUSF)

The Chair of CUSF is not a member of CUSF. Thus, if the Chair is from College Park, a replacement must be named. At the end of his/her term as Chair, if his/her term on CUSF is not finished, he/she resumes his/her position as a CUSF member.

The normal term for CUSF representatives is three (3) years, with two alternates serving three (3) year terms; if both alternates are elected at the same time, priority to be a replacement shall be in order of votes received. If a regular representative is unable to serve out his/her term, an alternate replaces him/her for the remainder of the term, and a new alternate is named. The replacement representative shall be chosen in order of number of votes received. The Office of the University Senate will identify a replacement alternate subject to confirmation by the Senate Executive Committee.

The University Senate will elect representatives to CUSF each spring. The Senate Nominations Committee will solicit candidates and will present a slate to the Chair of the Senate with at least one (1) candidate for each vacant position to be filled. At the Transitional Meeting of the Senate, faculty Senators will vote to elect representatives to CUSF. Each faculty Senator shall have as many votes as there are open positions. If there are more candidates than positions, the person(s) receiving the most votes, in order, are declared representatives. The person receiving the next most votes is declared alternate. The remaining person, in order of vote tally, will be asked to move into the alternate position if the previous paragraph comes in to play. A record of the outcome of the election will be retained by the Executive Secretary and Director of the University Senate. If there are not sufficient candidates, or the pool of candidates is exhausted, representatives are chosen by the Executive Committee.

# APPENDIX 6 PROCEDURES FOR ELECTIONS OF UMCP REPRESENTATIVES TO THE COUNCIL OF UNIVERSITY SYSTEM STAFF (CUSS)

The mission of the Council of University System Staff (CUSS) is to provide a voice for Staff employee concerns in reference to basic decisions that affect the welfare of the University System of Maryland (USM) and its employees. CUSS speaks for all non-exempt and exempt staff employees on Regular and Contingent II Status, who are not represented by a union under collective bargaining.

CUSS is comprised of Staff employees representing each USM institution and the USM Office (USMO). Institution membership is proportionate to the number of Staff employees at the individual institutions, with a minimum of two (2) primary members and two (2) alternate members per institution. Representation on CUSS from each constituent institution is apportioned according to the following formula: 1 to 999 eligible employees, 2 representatives; over 1000 eligible employees, 3 representatives. Staff at each constituent institution shall also select an alternate who shall substitute for a regular member of CUSS when needed. Alternates should be selected at the same time and in the same manner as regular members. A delegation may include more than one (1) alternate who is eligible to cast a vote for an absent member provided the member has given prior notification to the Chair of CUSS. The University of Maryland, College Park is entitled to three (3) representatives, and up to three (3) alternates.

As defined in 6.10.f of the Senate *Bylaws*, the Senate Staff Affairs Committee is responsible for administering the CUSS nomination and election process. Definitions of eligible staff shall be determined by the Board of Regents and CUSS. The CUSS elections will be administered in the spring semester every other year, as the terms of the current CUSS representatives are expiring. The Staff Affairs Committee will solicit candidates from the eligible staff population and will present ballots to the same population with at least one (1) candidate for each vacant position to be filled.

Eligible staff employees will vote to elect representatives to CUSS. If there are more candidates than positions, the person(s) receiving the most votes, in order, are declared representatives. The person(s) receiving the next most votes are declared alternate(s). A record of the outcome of the election will be retained by the Executive Secretary and Director of the University Senate.

New members shall begin their terms August 1. The normal term for CUSS representatives and alternates is two (2) years. If a regular representative is unable to serve out his/her term, an alternate replaces him/her for the remainder of the term, and a new alternate is named. The replacement representative shall be chosen in order of number of votes received.

# APPENDIX 7 PROCEDURES FOR REVIEW OF COLLEGE AND SCHOOL PLANS OF ORGANIZATION

- 1. In accordance with Article 11 of the Plan, each College, School, Department and other Academic Program, and the Library, shall have a Plan of Organization.
  - a. The Plan of Organization of each College, School, and the Library shall be reviewed by the University Senate according to the procedures detailed in section 2 of this appendix. All revisions to such Plans of Organization must be approved by the University Senate and the President prior to taking effect.
  - b. The Plan of Organization of a Department or other Academic Program shall be reviewed and revised by the Faculty Advisory Committee of the College to which it belongs. In the review and revision of such Plans, the University Senate may act in an advisory capacity if asked to do so by the College.
- 2. Each College, School, and the Library shall review and revise its Plan of Organization in accordance with Article 11.3 of the Plan and shall submit it to the University Senate for review.
  - Revised Plans of Organization shall be reviewed by the Senate Elections, Representation, and Governance (ERG) Committee for compliance with the University's Plan of Organization, University policy, and best practices of shared governance.
  - b. The Senate Faculty Affairs Committee shall review the Appointment, Promotion, and Tenure or Permanent Status section of each Plan and any related documentation for compliance with the University's APT Policy. The Senate Faculty Affairs Committee shall also review the Appointment, Evaluation, and Promotion Policy and any related documentation for compliance with University policies on professional track faculty and the University's Guidelines for Appointment, Evaluation, and Promotion of Professional Track Faculty.
  - c. The ERG and Faculty Affairs Committees shall communicate any concerns or requested revisions to the respective College, School, or Library.
  - d. Once all necessary revisions have been made, the ERG and Faculty Affairs Committees shall certify that they find the Plan to be in compliance and the revised Plan of Organization shall be submitted to the College Assembly or equivalent for approval.
  - e. Upon approval of the College Assembly or equivalent, the ERG Committee shall submit the revised Plan and its accompanying report to the Senate Executive Committee for review and placement on the Senate Agenda.
  - f. The revised Plan of Organization shall require final approval by the University Senate and the President.
- 3. During the initial implementation of a recently approved Plan of Organization, a College, School, or the Library may submit additional minimal or technical amendments to the Senate within one year of final approval by the University President. These revisions will undergo an expedited review process by the Senate ERG Committee, and by the Faculty Affairs Committee if appropriate. The committee(s) shall review only those amendments submitted by the College, School, or the Library, and shall not conduct a full review of the Plan. Upon approval by the ERG Committee (and the Faculty Affairs Committee, if necessary), the amendments shall be submitted to the College Assembly, the Senate Executive Committee, the Senate, and the President according to the procedures outlined above in section 2 d-f.

4. Until a revised Plan of Organization is approved by the University Senate and President, the version of the Plan of Organization of each College, School, and the Library that was most recently approved by the University Senate and President remains in effect, and provides the rules under which the College, School, or the Library must review and approve future revisions to its Plan. The University Plan of Organization supersedes any provisions in the Plan of any College, School, the Library, Department, or Academic Program that are in conflict with the purpose, applicability, or intent of the University Plan.

#### Dates of Approval, Updates and Amendments to the Senate Bylaws

Approved, Campus Senate, October 9, 1986 Approved, Board of Regents, February 6, 1987

Updated, July 11, 1988

Amended, February 13, 1986 Amended, December 7, 1986

Amended, May 7, 1990

Amended, September 13, 1990 Amended, November 15, 1990

Amended, October 14, 1993

Amended, December 6, 1993

Amended, March 31, 1994

Amended, April 18, 1994

Amended, May 5, 1994

Amended, November 10, 1994

Amended, August 28, 1996

Amended, May 15, 1997

Amended, March 5, 1998

Amended, April 2, 1998

Amended, April 6, 2000

Amended, February 12, 2001

Amended, September 19, 2002

Amended, February 3, 2003

Amended, October 16, 2003

Amended, April 19, 2004

Amended, April 4, 2005

Amended, May 15, 2007

Amended, May 8, 2008

Amended, October 16, 2008

Amended, February 9, 2009

Amended, May 4, 2009

Amended, November 12, 2009

Amended, March 3, 2010

Amended, February 9, 2011

Amended, May 4, 2011

Amended, March 8, 2012

Amended, April 19, 2012

Amended, May 2, 2013

Amended September 18, 2013

Amended, April 15, 2015

Approved after 2015 Plan of Org Review, May 4, 2015

Amended, November 20, 2015

Amended, December 14, 2015

Amended, February 18, 2016

Amended, March 18, 2016

Amended March 24, 2017

Amended November 8, 2017

Amended May 3, 2019

Amended February 7, 2020

Amended March 30, 2020

Amended November 12, 2020

Amended December 10, 2020

Amended April 9, 2021

Amended September 10, 2021

Amended September 29, 2023

Senate Programs, Curricula, & Courses Committee

# Revise the Policy and Guidelines for Combined Bachelor's/Master's Programs

## 2024-2025 Committee Members

Wendy Stickle (Chair)
Adam Bain (Undergraduate Student)
Theresa Bickham (Faculty)
Robert Brame (Ex-Officio – Provost's Rep)
Matthew Cain (Ex-Officio - President's Rep
George Mel (Staff)
Goger Joanna (Undergraduate Student)
Juana Hurtado (Faculty)
Ritesh Karki (Faculty)
Lisa Kiely (Ex-Officio – Ombuds Officer)

Daniel Mack (Faulty)
Pawan Pradhan(Faculty)
William Reed (Faculty)
Radford Skudrna
Faculty)
Tracy Sweet (Faculty)
Paul Turner (Faculty)
Leeann Arbelo Wong
(Undergraduate Student)
Ji Seung Yang (Faculty)

Date of Submission February 13, 2025

#### **BACKGROUND**

Jason Farman (Faculty)

In October 2024, the Graduate School submitted a proposal to Revise the Policy and Guidelines for Combined Bachelor's/Master's Programs (<a href="Policy Number: III-2.20(A)">Policy Number: III-2.20(A)</a>). The Graduate School provided rationale that Combined programs are often aligned within specific areas and the curricula across the two degrees typically share content. In recognition, programs are often designed with "overlapping" credits, such that courses that provide similar instruction within the two separate degree programs are effectively removed from the graduate program rather than having students duplicate very similar experiences.

Historically, 25-35% of graduate degree credits have been approved for such overlap, thus reducing the requirements of the graduate degree. Additionally, most master's programs were ~30 credits, but now the campus has several master's degree programs with 45 or more credits. The Graduate School was hoping to move away from a specific credit number to a percentage of overlapping credits, thus maintaining the spirit of the policy while recognizing the more varied structures of modern master's degrees (i.e., many are no longer structured as 10, 3-credit courses).

On November 13, 2024, the Senate Executive Committee (SEC) charged the Programs, Curricula, & Courses (PCC) Committee (Appendix 1) with reviewing proposal to Revise the Policy and Guidelines for Combined Bachelor's/Master's Programs.

# **COMMITTEE WORK**

The Faculty Affairs Committee (FAC) began its consideration of the charge in February 2025. The Committee reviewed the "Proposal to revise the Policy and Guidelines for Combined Bachelor's/Master's" (Appendix 2) which provided context on the historical practice for Combined Bachelor's/Master's programs, as well as motivation for the current change. The Proposal also clarified that the present policy only changes a degree program's requirements at the graduate level; the bachelor's degree requirements of 120 credits are unchanged, as undergraduate courses are replaced by graduate courses in the combined program curriculum. The master's program requirements are

affected, as some fraction of required graduate credits are taken while the student is an undergraduate.

The Committee were provided the relevant materials, and considered the policy proposal, the current university policy, and the USM policy, as required by the charge.

Consultation with the Graduate School representative, Stephen Roth, Dean of the Graduate School took place on the February 7, 2025 PCC Meeting. This provided an opportunity for Dean Roth to explain in greater depth the context behind the proposed policy change, and allow the committee to ask questions.

During the presentation to the PCC Committee, Dean Roth provided rationale for changing a specific credit limit (of 9 credits) to a proportional credit. The proposed policy indicates a 35% maximum overlap, defined as up to 35% of the graduate course credit requirement. Dean Roth also explained that minor changes had been made to the policy, but only in the Graduate Catalog, not reflected in the greater University Policy (III-2.20(A)). The proposal brings those changes into alignment. The Registrar and Senate Office were consulted, and those comments were included in the materials to the committee.

Discussions during this presentation revolved around restrictions, with members wondering if specific courses (ie; 600- and 700-level courses) would be restricted. Dean Roth provided clarity that small restrictions on types of courses (ie: internships) do exist in the policy, and that there was an attempt not to include too many specific restrictions, because others involved in the approval process (PCC Committee, Directors, or Faculty themselves) would be in a better place to articulate and expand the rationale for individual proposals.

Other discussion included if accreditation was involved in determining this policy as University-level, as opposed to allowing each program to decide. Dean Roth provided insight that the University System of Maryland (USM) does have a requirement (III-2.20- Policy on Combined Bachelor's/Master's Programs) that each school of a University-level policy, and that allowing programs individual flexibility, with some general guidance has been beneficial.

The policy in full was reviewed by the PCC Committee, with minimal feedback, and no concerns. The Committee. After due consideration, the Programs, Curricula, & Courses Committee approved the proposed revisions to the University of Maryland, College Park Policy and Guidelines for Combined Bachelor's/Masters Programs.

The Office of General Counsel reviewed the proposed policy and approved, with no edits.

#### RECOMMENDATIONS

The PCC Committee recommends that the University of Maryland, College Park Policy and Guidelines for Combined Bachelor's/Masters Programs (III-2.20(A)) should be revised as shown immediately following this report.

#### **APPENDICES**

Appendix 1 - Charge from the Senate Executive Committee

Appendix 2- Proposal submitted to the University Senate Office



# **CHARGE**

Charged: November 13, 2024 | Deadline: April 4, 2025

# Revise the Policy and Guidelines for Combined Bachelor's/Master's Programs (Senate Document #24-25-20)

# Programs, Courses & Curriculum Committee | Chair: Wendy Stickle

The Senate Executive Committee (SEC) and Senate Chair Sly request that the Programs, Courses & Curriculum (PCC) Committee review the *Proposal to Revise the Policy and Guidelines for Combined Bachelor's/Master's Programs*.

Specifically, the PCC Committee should:

- 1. Review the proposal entitled *Proposal to Revise the Policy and Guidelines for Combined Bachelor's/Master's Programs*.
- 2. Consult:
  - a. The University of Maryland Graduate Council
  - b. The original proposer, for rationale and background on the proposed changes
  - c. Office of General Counsel
- 3. Consider the proposed revisions to the policy as outlined in the submitted proposal.
- 4. If appropriate, recommend whether the policy should be revised and if so, provide suggested revisions.

We ask that you submit a report to the University Senate Office no later than **April 4**, **2025**. If you have questions or need assistance, please contact the University Senate Office, <a href="mailto:senate-admin@umd.edu">senate-admin@umd.edu</a>.

# UNIVERSITY SENATE

# **PROPOSAL**

Submitted on: 14 October 2024

Proposal to revise the Policy and Guidelines for Combined Bachelor's/Master's Programs

NAME/TITLE	Stephen Roth, Associate Provost and Dean		
EMAIL	sroth1@umd.edu	PHONE	301-405-5793
UNIT	Graduate School	CONSTITUENCY	Faculty

# DESCRIPTION OF PROPOSAL

The Policy and Guidelines for Combined Bachelor's/Master's Programs (Policy Number: III-2.20(A)) has not been amended since 2001 despite many changes to master's program structures across the campus. The current policy provides for up to 9-credit overlaps for combined programs that are built upon a traditional 30-credit master's degree containing 3-credit courses. Modern master's programs are highly varied in structure, including overall credits and course-specific credits. The considerable expansion of the number of combined bachelor/master's degree programs over the past 20 years has revealed other policy revisions that are needed regarding student and program eligibility, among other issues. The current proposal provides a suite of revisions to the policy to bring it up to date and provide better guidance for programs on developing combined bachelor's/master's programs.

# <u>Additional background information</u> that may be helpful for reviewers:

The university offers various combined and dual degree programs to support the educational and career goals of students seeking a master's degree. These combination programs are valuable to students as they both shorten the time needed to complete the graduate degree and save money. Combined programs are often aligned within specific areas and the curricula across the two degrees typically share content. In recognition, programs are often designed with "overlapping" credits, such that courses that provide similar instruction within the two separate degree programs are effectively removed from the graduate program rather than having students duplicate very similar experiences. Historically, 25-35% of graduate degree credits have been approved for such overlap, thus reducing the requirements of the graduate degree.

We distinguish between *Structured* and *Individual* combined degree programs as follows: *Structured* (*PCC-approved*) programs have been pre-approved by the *Programs*, *Curricula*, & *Courses* (*PCC*) *Committee* through the standard approval process, whereas *Individual combined degree programs* are custom-designed for a particular student and are not pre-approved through the PCC Committee.

Of note, for bachelor's-master's programs, the bachelor's degree still requires completion of 120 or more credits. As such, in a combined program, the undergraduate will take some number of graduate-level credits in the bachelor's program, with those credits replacing typical undergraduate credits taken in the final year. Those graduate credits then count for <u>both</u> the undergraduate and graduate degrees. The bachelor's degree still retains 120 credits, and the graduate degree still retains its full curriculum, but the overlapped credits were taken while the student was an undergraduate.

The current policy for bachelor's-master's programs allows for 9 overlapping credits, with some added flexibility for master's programs over 30 credits or other situations. This overlap is available for both

Individual and Structured combination programs. Historically, most master's programs were ~30 credits, but now the campus has several master's degree programs with 45 or more credits. The spirit of the existing policy is that a fraction of credits should be approved for overlap between two related degree programs. However, the 9-credit limit was established based on earlier master's program structures, as the USM policy was first approved in 1990, followed by the UMD policy in 1991.

# DESCRIPTION OF CHANGE YOU WOULD LIKE TO SEE

In the present combined bachelor's-master's policy proposal, we are seeking to move away from a specific credit number to a percentage of overlapping credits, thus maintaining the spirit of the policy while recognizing the more varied structures of modern master's degrees (i.e., many are no longer structured as 10, 3-credit courses). The proposed policy indicates a 35% maximum overlap, defined as up to 35% of the graduate course credit requirement.

We propose that any previously approved programs that exceed the 35% overlap threshold would remain approved as is. Examining all of the currently approved combined bachelor's-master's programs, we find only two programs with greater than 35% of graduate credits overlapped: a combined bachelor's-master of public policy (MPP), with 18 overlapping credits approved on a 48-credit MPP program (37.5%), and a long-standing bachelor's-master of education (MED) in Special Education with 12 overlapping credits for a 30-credit master's (note that the bachelor's degree in this case requires 123 total credits, resulting in 141 total credits for both degrees). The Special Education degree was first approved in the early 1980s, prior to this policy being enacted, and has maintained its accreditation and successful student placements since that time. As such, we are recommending that the combined bachelor's-master's policy be revised to move to a 35% credit overlap maximum, but a higher fraction could be considered with a strong rationale.

See Appendix A for the full text of proposed policy revisions.

## SUGGESTION FOR HOW YOUR PROPOSAL WOULD BE PUT INTO PRACTICE

The proposed policy revision would be put into practice immediately upon approval, providing updated guidance to programs and the PCC approval process both before and during any combined degree program proposals.

# ADDITIONAL INFORMATION

The present policy, III-2.20A, is the only policy that straddles both the Senate and Graduate Council, with the Graduate Council charged with overseeing all graduate-only academic policies. Because the present policy includes bachelor's degrees, the Senate has provided overall policy ownership.

For the present policy, however, the only changes to a degree program's requirements are at the graduate level; the bachelor's degree requirements of 120 credits are unchanged, as undergraduate courses are replaced by graduate courses in the combined program curriculum. The master's program requirements are affected, as some fraction of required graduate credits are taken while the student is an undergraduate.

Because the policy impacts graduate academic policy, we seek Senate Executive Committee approval to assign this policy review and approval process to the Graduate Council rather than the Senate. The Graduate Council is well prepared to address the implications of the policy on master's degree programs and consider the consequences for combined programs.

#### APPENDIX A

Policy Number: III-2.20(A)

University of Maryland, College Park Policy and Guidelines for Combined Bachelor's/

<del>\dagaalaster's</del> Programs

(Approved By The President August 01, 1991, Amended May 10, 2

In a combined bachelor's/master's program, some graduate level throughout the policy:en for undergraduate credit may also be applied towards the graduate master's degree program at the University of Maryland. A bachelo be developed for an individual student, or it may be a structured programs on the 15-week semester term are eligible for the com student can pursue one combined bachelor's/master's program combined bachelor's/master's program may not defer the start but may request a Leave of Absence in their first year in the mas who do not begin their master's program the term after completing will have the double-counted credits converted to being counted degree.

Stephen M. Roth 2024-10-07 17:06:00

Consistent with apostrophe usage s/master's program may

Stephen M. Roth 2024-09-30 16:07:00

Provides information about the eligibility of students and degree programs covered by this policy. These eligibility limits have been in place for some time.

# Individual Student Bachelor's/Master's Program:

A 🚰 individual bachelor's/master's program may be developed 🖞 stephen M. Roth undergraduate student in consultation with his/her their academic is available only to students whose academic performance is exc developed according to the individual career interests and goals be an integrated learning experience rather than merely the comp of graduate and undergraduate credits. The proposed program requires the approval of the directors of both the undergraduate and the graduate programs involved and of the Dean for Undergraduate Studies and the Dean of the Graduate School Normally no more than

2024-10-04 20:26:00

Providing clarifying specificity.

nine credits 35% of graduate courses course credits applied to the bachelor's degree may be counted also for graduate credit in an individual student program. Courses to be double-counted must be at the 600 graduate level or above and must be passed with at

least a "B-" grade. ⊜urses at the 400-level may be included in the if the component master's degree allows 400-level courses to c Individual study courses, internships, or courses given as credit by eligible. The credits to be double-counted will be designated as a alow 400 level credits to fulfill program program after the student receives the bachelor's degree and mat requirements he Graduate School. This designation will be canceled if the student withdraws from the graduate program before completing the master's degree.

Stephen M. Roth 2024-10-04 20:26:00

Clarifying that some master's programs

# Structured Bachelor's/Master's Program:

A structured bachelor's/master's program is an articulated a clearly defined curriculum

combining an existing **wiversity** of Maryland (UMD) College Park and an existing master's program at the University of Maryland UN by the same or by different departments. Such a program is to be

Stephen M. Roth 2024-10-04 20:25:00

This policy is not relevant for partnership agreements with programs outside of UMCP.

whose academic performance is exceptional and should be an integrated learning experience rather than merely the completion of a certain number of graduate and undergraduate credits. A proposal for such a program should be submitted by the college(s) housing the academic programs concerned and requires the approval of the

Graduate Council Programs, Curricula, & Courses (PCC) Committ Stephen M. Roth the University Senate PCC Committee, the Senate, and the Prov

Necessary features of a structured bachelor's/master's program in

Updating the current PCC approval process.

- There must be Specific requirements for admission to the combined program that speak to the exceptional performance of the students to be admitted. At a minimum, students accepted for the program must be clearly admissible to the graduate program portion.
- The program should be designed so as not to unduly delay the students' receipt of their bachelor's degrees. Taking graduate credits should not unduly limit the breadth of the student's experience through premature specialization.
- Parequirements of the bachelor's program and of the mat Stephen M. Roth <del>completed to receive the two degrees.</del> Where appropriate, while an undergraduate may substitute for courses require This is not technically possible with a major program.
- the students may be offered deferred admission to the gra of the Junior year program, subject to completion of the s timely fashion and with a specified level of achievement. F gGraduate sSchool will require completion of all requirem overly prescriptive. degree.
- The credits to be double-counted will be designated as applicable to the graduate program after the student receives the bachelor's degree and matriculates in the Graduate School. This designation will be canceled if the student withdraws from the graduate program before completing the master's degree.

2024-10-04 20:23:00

combined program and the sentence results

Stephen M. Roth 2024-10-04 20:24:00

Redundant to the following sentence, or

A structured bachelor's/master's program may normally include up to nine 35% of credits of graduate level courses-credits (600 level and above) that are counted both for the bachelor's program and the master's program. Courses at the 400-level may be included in the double-counting credits if the component master's degree allows 400-level courses to count toward the degree. Here than nine double-counted credits Stephen M. Roth 2024-10-04 20:22:00 the following conditions are satisfied.

- The additional graduate credits applied to the undergraduate limit the breadth of the student's experience through pren condition may be satisfied, for example, if the graduate c courses required in the undergraduate program that would have been taken in any case, but at a less advanced level.
- The master's program requires substantially more than thirty credits. This will be deemed to be satisfied if the combined program, with double-counting, still requires 150 or more credit hours to complete.

These conditions are no longer relevant when moving to a proportion of overlapping course credits. to for

#### APPENDIX A

Policy Number: III-2.20(A)

University of Maryland, College Park Policy and Guidelines for Combined Bachelor's/ MastersMaster's Programs

(Approved By The President August 01, 1991, Amended May 10, 2001)

In a combined bachelor's/master's program, some graduate level courses initially taken for undergraduate credit may also be applied towards the graduate credit requirements for a master's degree program at the University of Maryland. A bachelor's/master's program may be developed for an individual student, or it may be a structured program. Only graduate programs on the 15-week semester term are eligible for the combined degree. Each student can pursue one combined bachelor's/master's program. Students admitted to a combined bachelor's/master's program may not defer the start of their master's program, but may request a Leave of Absence in their first year in the master's program. Students who do not begin their master's program the term after completing their bachelor's degree will have the double-counted credits converted to being counted only for the bachelor's degree. Students and programs must identify the courses selected for double counting prior to course registration. Students completing two bachelor's degrees simultaneously must complete both bachelor's degrees before matriculating into the master's program; if the bachelor's degree outside of the combined program is not completed prior to matriculation, it will need to be pursued following completion of the master's degree and will require reenrollment at the undergraduate level.

#### Individual Student Bachelor's/Master's Program:

A An individual bachelor's/master's program may be developed by an individual a current undergraduate student in consultation with his/her their academic advisor. Such a program is available only to students whose academic performance is exceptional. It is to be developed according to the individual career interests and goals of the student and should be an integrated learning experience rather than merely the completion of a certain number of graduate and undergraduate credits. The program should be designed so as not to unduly delay the students' receipt of their bachelor's degrees. Taking graduate credits should not unduly limit the breadth of the student's experience through premature specialization. The proposed program requires the approval of the directors of both the undergraduate and the graduate programs involved and of the Dean for Undergraduate Studies and the Dean of the Graduate School, An individual bachelor's/master's program may normally include up to 35% of credits of graduate level courses that are counted both for the bachelor's program and the master's program. Normally no more than nine credits of graduate courses applied to the bachelor's degree may be counted also for graduate credit in an individual student program. Courses to be double-counted must be at the 600 graduate level or above and must be passed with at least a "B-" grade. Courses at the 400-level may be included in the double-counting credits if the component master's degree allows 400-level courses to count toward the degree. Individual study courses, internships, or courses given as credit by examination are not eligible. The credits to be

**Commented [SMR1]:** Consistent with apostrophe usage throughout the policy.

Commented [SMR2]: Provides information about the eligibility of students and degree programs covered by this policy. These eligibility limits have been in place for some time.

Commented [SMR3]: Providing clarifying specificity.

**Commented [SMR4]:** Mimicking the phrasing from the Structured Bachelor's/Master's for consistency.

**Commented [SMR5]:** Mimicking the phrasing from the Structured Bachelor's/Master's for consistency.

Commented [SMR6]: Clarifying that some master's programs allow 400 level credits to fulfill program requirements.

double-counted will be designated as applicable to the graduate program after the student receives the bachelor's degree and matriculates in the Graduate School. This designation will be canceled if the student withdraws from the graduate program before completing the master's degree.

#### Structured Bachelor's/Master's Program:

A structured bachelor's/master's program is an articulated a clearly defined curriculum combining an existing University of Maryland (UMD) College Park undergraduate program and an existing master's program at the University of Maryland UMD College Park, oaered by the same or by diaerent departments. Such a program is to be designed for students whose academic performance is exceptional and should be an integrated learning experience rather than merely the completion of a certain number of graduate and undergraduate credits. A proposal for such a program should be submitted by the college(s) housing the academic programs concerned and requires the approval of the Graduate Council Programs, Curricula, & Courses (PCC) Committee , the Graduate Dean, the University Senate PCC Committee, the Senate, and the PrevestPresident.

Necessary features of a structured bachelor's/master's program include the following:

- There must be Specific requirements for admission to the combined program that speak to the exceptional performance of the students to be admitted. At a minimum, students accepted for the program must be clearly admissible to the graduate program portion.
- The program should be designed so as not to unduly delay the students' receipt of their bachelor's degrees. Taking graduate credits should not unduly limit the breadth of the student's experience through premature specialization.
- All requirements of the bachelor's program and of the master's program must be completed to receive the two degrees. Where appropriate, graduate courses taken while an undergraduate may substitute for courses required in the undergraduate major program.
- The students may be oaered deferred admission to the graduate school at the endof the Junior year program, subject to completion of the senior year program in a
  timely fashion and with a specified level of achievement. Formal admission to the
  gGraduate sSchool will require completion of all requirements for the bachelor's
  degree.
- The credits to be double-counted will be designated as applicable to the graduate
  program after the student receives the bachelor's degree and matriculates in the
  Graduate School. This designation will be canceled if the student withdraws from
  the graduate program before completing the master's degree.

A structured bachelor's/master's program may normally include up to nine 35% of credits of graduate level courses credits (600 level and above) that are counted both for the bachelor's program and the master's program. Courses at the 400-level may be included in the double-counting credits if the component master's degree allows 400-level courses to

**Commented [SMR7]:** This policy is not relevant for partnership agreements with programs outside of UMCP.

**Commented [SMR8]:** Updating the current PCC approval process.

Commented [SMR9]: This is not technically possible with a combined program and the sentence results in confusion.

**Commented [SMR10]:** Redundant to the following sentence, or overly prescriptive.

count toward the degree. More than nine double-counted credits may be allowed if both of the following conditions are satisfied.

- The additional graduate credits applied to the undergraduate program do not unduly-limit the breadth of the student's experience through premature specialization. This condition may be satisfied, for example, if the graduate credits substitute for courses required in the undergraduate program that would have been taken in any case, but at a less advanced level.
- The master's program requires substantially more than thirty credits. This condition
  will be deemed to be satisfied if the combined program, with double-counting, still
  requires 150 or more credit hours to complete.

Commented [SMR11]: These conditions are no longer relevant when moving to a proportion of overlapping course credits.

# it>Regents>Bylaws>Section1ll> 111220

68.0 III-2.20-POLICY ON COMBINED BACHELOR'S/MASTER'S PROGRAMS

(Approved by the Board of Regents, January 11, 1990)

Each institution may establish guidelines for the development and implementation of combined bachelor's/master's programs. These plans may be of two types: (1) plans which permit the development by an individual student of a program specific to that student's career interests and goals, and (2) structured plans within or among departmental or program units or institutions which provide integrated programs attractive to larger groups of students. The program should be available only to students whose academic performance is exceptional and who are admissable to the graduate program. Institutional policy shall set the maximum number of credits applicable to both degrees, normally no more than nine. No more than one master's degree may be earned through a combined bachelor"s/master's degree program.

Replacement for: BOR I-12.00